UC Irvine UC Irvine Electronic Theses and Dissertations

Title Neurological and Behavioral Predictors of Aphasia Recovery

Permalink https://escholarship.org/uc/item/2x7176r4

Author Duncan, E. Susan

Publication Date 2016

Peer reviewed|Thesis/dissertation

UNIVERSITY OF CALIFORNIA, IRVINE

Neurological and Behavioral Predictors of Aphasia Recovery

DISSERTATION

submitted in partial satisfaction of the requirements for the degree of

DOCTOR OF PHILOSOPHY

in Psychology

by

E. Susan Duncan

Dissertation Committee: Professor Steven L. Small, Chair Professor Gregory S. Hickok Professor Ramesh Srinivasan

Chapter 2 © 2015 Elsevier Portion of Chapter 3 © 2016 SAGE Publications Chapter 4 © 2016 SAGE Publications All other materials © 2016 E. Susan Duncan

DEDICATION

Do mo mháthair. Sláinte, táinte agus áthas! Tá grá agam leat.

Airson m' athair. Wha's like us? Tha gaol agam ort.

To the questions:

So what is this mind of ours: what are these atoms with consciousness? Last week's potatoes!

They now can remember what was going on in my mind a year ago – a mind which has long ago been replaced. To note that the thing I call my individuality is only a pattern or dance, that is what it means when one discovers how long it takes for the atoms of the brain to be replaced by other atoms. The atoms come into my brain, dance a dance, and then go out – there are always new atoms, but always doing the same dance, remembering what the dance was yesterday.

Richard P. Feynman What Do You Care What Other People Think?

and the answers:

New knowledge is the most valuable commodity on earth. The more truth we have to work with, the richer we become.

> Kurt Vonnegut, Jr. Breakfast of Champions

Table of Contents

LIST OF FIGURES	vi
LIST OF TABLES	viii
ACKNOWLEDGMENTS	ix
CURRICULUM VITAE	xi
ABSTRACT OF THE DISSERTATION	xvi
Chapter 1: Introduction	
1.1 Aphasia	
1.2 History of Aphasia	2
1.2.1 Ancient Egypt	2
1.2.2 Ancient Greece	
1.2.3 Europe: Middle Ages and Early Modern Period	
1.2.4 Europe: Nineteenth Century	4
1.2.5 Twentieth Century	
1.3 Present Day	
1.4 Rationale for this Thesis	
Chapter 2: Imitation-Based Aphasia Therapy	
2.1 Repetition and Imitation in Aphasia	
2.2 Neurobiological Approaches to Language and Aphasia	
2.3 Mirror Neuron System	
2.3.1 Macaque	
2.3.2 Human	
2.4 Mirror Neuron System and Language	
2.4.1 Perception and Production of Articulated Speech	
2.4.2 Comprehension of Action Language	
2.5 Dual Streams for Speech	
2.5.1 The Ventral Stream	
2.5.2 The Dorsal Stream and Parietal Cortical Connectivity	
2.6 Aphasia Therapy: Speech Imitation as Therapeutic Tool	
2.7 Mirror Neuron System and Rehabilitation	
2.8 Aphasia Therapy: Speech Imitation as Therapeutic Theory	
2.8.1 IMITATE	
2.8.2 Speech Entrainment	
2.9 Aphasia Therapy: Nonspeech Motor Observation and Imitation	
2.9.1 Visual Action Therapy	
2.9.2 Action Observation Treatment	
2.10 Conclusion	
Chapter 3: Participants and Study Design	
3.1 Participants	
3.2 Experimental Summary	
3.3 Therapy Description	
3.3.1 Therapy Features	
3.3.2 Selection of Lexical and Phrasal Stimuli	
3.4 Behavioral Measures	
3.4.1 Western Aphasia Battery	

3.4.2 Repetition Test	47
3.4.3 Narrative Production	47
3.5 Neuroimaging	48
3.5.1 Acquisition	48
3.5.2 Virtual Brain Transplant and FreeSurfer	48
3.5.3 Brain Parcellation.	52
3.5.4 Preprocessing	53
Chanter 4. Derformance Variability of a Dredictor of Degraphic to Anhagia Treatmen	+ 51
Chapter 4: Performance variability as a Predictor of Response to Aphasia Treatmen	l 54
4.1 1 Declaration	
4.1.1 Background	
4.1.2 Intra-individual Variability in Cognitive and Motor Function	
4.1.3 Intra-Individual Variability in Aphasia	
4.1.4 Motivation for the Present Study	
4.2 Materials & Methods	50
4.2.1 Participants	
4.2.2 Experimental Summary	
4.2.3 Behavioral Measures	
4.3 Results	
4.3.1 Changes in Behavioral Performance	
4.3.2 Intra-individual Variability as Predictor of Improvement	
4.4 Discussion	66
Chapter 5: Effects of Imitation-Based Aphasia Therapy on Narrative Content	71
5.1 Introduction	
5.2 Materials & Methods	
5.2.1 Participants	73
5 2 2 Neuroimaging	73
5.2.3 Behavioral Measures	73
5.3 Results	
5.3.1 Neuroimaging Findings	
5 3 2 Behavioral Measures	76
5.4 Discussion	
Chapter 6: Changes in Dynamic Resting State Network Connectivity Following Apha	sia
Therapy	81
6.1 Introduction	81
6.2. Materials & Methods	83
6.2.1 Participants	83
6.2.2 Behavioral Measures	84
6.2.3 Neuroimaging	84
6.2.4 Dynamic Functional Network Connectivity	85
6.2.5 Individual Correlations in Functional Connectivity States and Behavioral Measures	87
6.3. Results	88
6.3.1 Resting State Networks (RSNs)	88
6.3.2 Dynamic Functional Network Connectivity	89
6.3.3 Individual Correlations in Functional Connectivity States and Behavioral Measures	90
6.4 Discussion	92
Chanter 7. Natwork Properties Underlying Changes in Nervetive Disduction	04
Thanked votion	90
7.1 IIII OUUCIOII	
1.2. Iviateriais & Iviethous	

7.2.1 Participants	
7.2.2 Behavioral Measures	
7.2.3 Neuroimaging Measures	
7.2.4 Correlation of Modularity and Behavior	
7.3 Results	
7.4 Discussion	
Chapter 8: Conclusion	108
References	
Appendix A: Artifactual ICA Components	123
Appendix B: Overlap Of Resting State Networks with Healthy Controls	128
Appendix C: Non-Significant Dynamic Functional Connectivity States	

LIST OF FIGURES

Figure 1.1	The Edwin Smith Papyrus	2
Figure 1.2	Phrenological Chart	5
Figure 1.3	Brain of Louis Victor Leborgne, Broca's Original Patient	6
Figure 1.4	Carl Wernicke's Language Network	6
Figure 1.5	Lichtheim's "House" Model	7
Figure 1.6	Localization of Aphasia Symptoms According to Charcot	8
Figure 2.1	Visual and Motor Responses of a Mirror Neuron	15
Figure 2.2	Motor Evoked Potentials (MEPs) During Action Observation	19
Figure 2.3	Somatotopic FMRI Activation During Action Observation	21
Figure 2.4	Diagram of Inverse-Forward Model Pairs	23
Figure 2.5	Regions of Overlap for Speech Perception and Production	25
Figure 2.6	Dorsal and Ventral Streams for Language	28
Figure 2.7	Common Network for Speech Observation and Imitation	33
Figure 2.8	Post-therapy Production of Correct, Novel Words	36
Figure 3.1	Lesion Overlap for All Subjects	41
Figure 3.2	Visual Depiction of Experimental Design	42
Figure 3.3	Six Talkers Presenting Therapy Stimuli	43
Figure 3.4	Example of Therapy Interface	45
Figure 3.5	Virtual Brain Transplant (VBT) Workflow	50-51

Figure 3.6	Group Template	51
Figure 3.7	Lateral View of the Parcellated Left Hemisphere	52
Figure 3.8	Medial View of the Parcellated Left Hemisphere	52
Figure 3.9	Lateral View of the Parcellated Right Hemisphere	53
Figure 3.10	Medial View of the Parcellated Right Hemisphere	53
Figure 4.1	Pre-treatment Intra-Individual Variability, Mean, and Improvement	64
Figure 4.2	Changes in Intra-Individual Variability and Mean on the Repetition Test	65
Figure 5.1	Number of Sessions Completed and Percent Change in CIU Production	77
Figure 6.1	Resting State Networks (RSNs) Displayed on a Group Template	88
Figure 6.2	Similarity Graph for ICASSO Estimates	89
Figure 6.3	Correlation Matrix Representing State 10	90
Figure 6.4	Changes from Baseline in State 10 Dwell Time and CIU Production	91
Figure 6.5	Post-therapy Changes in State 10 Dwell Time and CIU Production	92
Figure 7.1	Visual Depiction of Two Graphs	97
Figure 7.2	Correlations Between Changes in CIU production and RSN Modularity	104

LIST OF TABLES

		Page
Table 1.1	Typical Aphasia Taxonomy	9
Table 3.1	Individual Data for Each of the 19 Subjects	40
Table 4.1	Full and Partial Correlation Coefficients for Pre-treatment Intra-	61
	Individual Variability and Improvement in Mean Accuracy for the	
	Repetition Test	
Table 4.2	Performance Measures for All Subjects	63
Table 5.1	Select Demographic Information and Pre-/Post-therapy Performance on	74
	Cinderella Narrative Task	
Table 5.2	Potential Predictors of Maintenance Used in the Regression Model	75

ACKNOWLEDGMENTS

First and foremost, I thank Steve Small, for taking me under his proverbial wing. I think it took me about two years to understand the distinctions you make and why they really do matter. I've appreciated our discussions of politics and world affairs as much as those about language and neuroscience. Thank you for being a great editor as well as a staunch defender of dietary freedom.

I also thank my committee, Greg Hickok and Ramesh Srinivasan for their roles in my M.S. concentration exam and Ph.D. defense. Further appreciation goes to the other committee members who participated in my advancement, Hernando Ombao and Claudia Kawas, and Departments of Cognitive Sciences and Neurology.

Emily Grossman has been far more than the Graduate Director of the Cognitive Neuroscience concentration. She has been an incredible source of wisdom and a phenomenally supportive ally throughout my program completion, particularly during my intensive job search. Thank you, Emily, for keeping on your "hit list", on task, and on time to this milestone.

Loraine Obler and JungMoon Hyun at CUNY Graduate Center were instrumental in my transition from the clinical world to research, as was the support of Jeff Fine and Pat McCaul. I would also like to thank all the folks at the Gertz clinic at Queens College and Queens Hospital Center for the preparation and direction that brought me here, as well as Wendy Morse at PS 165 for long ago introducing me to the world of speech-language pathology by fixing my speech.

I would like to thank Elsevier and Sage for granting the necessary permissions to include published work in this dissertation. Chapter 2 originally appeared in *Neurobiology of Language*. A portion of Chapter 3 and Chapter 4 originally appeared in *Neurorehabilitation and Neural Repair*. Appreciation is also extended to the other publishers who permitted reproduction of figures included in introductory chapters.

This dissertation was supported by grant funding from the National Institute of Deafness and other Communication Disorders (NIDCD) of the National Institutes of Health (NIH) under Grants R01-DC007488, R01-DC003378, and R33-DC008638 (Steven L. Small, PI); the James S. McDonnell Foundation under a grant to the Brain Network Recovery Group (A.R. McIntosh, PI); and Mr. William Rosing, Esq. This indispensable support is gratefully acknowledged.

This thesis would not have been possible without a dedicated team. Leora Cherney and her staff at the Rehabilitation Institute of Chicago coordinated and conducted the speech and language evaluations. Additional research staff at The University of Chicago included Blythe Buchholz and Robert Fowler, who helped coordinate the project, and Dan Rodney, who authored the IMITATE software. Tanya Schmah and Priya Santhanam have also been great collaborators on projects completed and in progress. Every neuroimaging lab should have an Ana Solodkin, but there's only one of her. She demands neurophysiological fidelity at all times. And she's funny doing it.

The Data Science Initiative (DSI) at UC Irvine deserves mention for introducing me to several of the software tools that I used in these analyses. I will genuinely miss their fantastic workshops.

The Center for Engaged Instruction offered a great community for my final year at UCI. Thanks to De Gallow, Brenda Fisher, Robert Nyenhuis, Edelina Burciaga, Danny Mann and all my fellow Pedagogical Fellows.

I would especially like to thank everyone at the Solodkin/Small Brain Circuits Laboratory for their support and camaraderie over the years. I may not miss all our lab meetings, but I'll miss all of you! I would particularly like to thank Mike Andric for introducing me to python and AFNI and to spending time with me when I knew literally nothing about neuroimaging. Jeff Riley, for introducing me to DTI analysis with FSL. Sarah Tune, for training me in the use and analysis of EEG. Inez Falcon, for being my fellow grad student in the lab lo these many years. Dave Keator, for just being a great guy to have around when you want to kick around a question.

Duke Shereen merits his own paragraph. Duke has been the best-natured and most helpful colleague I could have hoped to meet. Thanks for being a friend, for helping me keep perspective, and for listening to me rant PRN. I'd say I'll miss you, but you don't get off that easy....

Thanks to Sarah "Elizabeth" Kenny for being the funniest computer programmer and the nerdiest stand-up comedian to ever coexist in one single awesome package. I hear the comedy scene in Baton Rouge is really picking up and they're installing tons of bike lanes, too....

Shoshana Negev Abolafia gets all my love now and forever. You complete me. I'm glad to have you and Athanasios Galitsis in my life, if too far away.... Whatever.

Thanks to everyone who makes NYC still feel like home when I visit. And thanks to Louisiana State University for making me excited for the next phase.

Mama and Papa. There are no words to describe what completely fantastic human beings you are. I couldn't have asked for better, and I could never aspire to be more.

Finally, thank you to the patients who participated in this study and their families, and to everyone whose life has been touched by aphasia.

CURRICULUM VITAE

E. Susan Duncan

June 2016

EDUCATION

Ph.D.	Cognitive Neuroscience	2016
	University of California, Irvine	
	Dissertation: Neurological and behavioral predictors of aphasia recovery	
	Chair: Steven L. Small, Ph.D., M.D.	
M.S.	Cognitive Neuroscience University of California, Irvine	2015
M.A.	Speech-Language Pathology Queens College, City University of New York	2002
B.A.	Linguistics & Communication Disorders Queens College, City University of New York	2000
B.A.	Psychology Bard College at Simon's Rock	1996

TEACHING EXPERIENCE

Instructor	Stanbridge College	
	Applied Neuroscience and Occupations (graduate)	Fall 2015
		Spring 2015
		Fall 2014
		Spring 2014
Instructor	School of Social Sciences, UC Irvine	
	Introduction to Linguistics	Summer 2015
		Summer 2014
Teaching Assistant	School of Social Sciences, UC Irvine	
C	Cognitive Neuroscience	Winter 2016
	C C	Winter 2015
	Research Methods in Psychology	Fall 2015
	Psychology in Film	Spring 2015
	Psychology in Film	Spring 2015

	Introduction to Psychology	Fall 2014 Winter 2013
	Brain Disorders Introduction to Linguistics	Winter 2012 Spring 2014 Winter 2014 Fall 2013
	Language and Brain Medical Anthropology	Spring 2013 Fall 2012 Fall 2011
Pedagogical Fellow	Center for Engaged Instruction, UC Irvine Teaching Assistant Professional Development Program	Summer 2015

RESEARCH EXPERIENCE

Graduate Student Researcher	Solodkin/Small Brain Circuits Laboratory University of California, Irvine PI: Steven L Small, Ph.D., M.D.	2011 - 2016
Volunteer Researcher	Neurolinguistics Laboratory City University of New York PI: Loraine Obler, Ph.D.	2010 - 2011

PUBLICATIONS

Journal Articles (Peer reviewed)

Duncan, ES & Small, SL. (under review). Increased modularity of resting state networks supports improved narrative production in aphasia therapy.

Duncan, ES & Small, SL. (under review). Imitation-based therapy increases narrative content.

Duncan, ES & Small, SL. (under review). Changes in dynamic resting state network connectivity following aphasia therapy.

Santhanam, P, **Duncan, ES &** Small, SL. (in revision). Treatment-induced neural reorganization in chronic aphasia is dependent on time since stroke.

- **Duncan, ES**, Tune, S & Small, SL. (in press). The neurobiology of language: Relevance to linguistics. Yearbook of the Poznań Linguistic Meeting.
- **Duncan, ES**, Schmah, T, & Small, SL. (2016). Performance variability as a predictor of response to aphasia treatment. *Neurorehabilitation and Neural Repair*.
- Cahana-Amitay, D, Spiro, A, Higby, E, Ojo EA, Sayers, JT, Oveis, AC, **Duncan, ES,** et al. (2015). Who gets the last word? Cognitive contributions to word recognition in sentential context in aging. *Aging, Neuropsychology, and Cognition*, 1-27.

Invited Chapters

Duncan, ES & Small, SL. (2015). Imitation-based aphasia therapy. Chapter in *Neurobiology of Language*, Hickok, G & Small, SL, Eds. Elsevier Health Sciences.

AD HOC MANUSCRIPT REVIEWS

Annals of Neurology Brain & Language Cerebral Cortex Cortex Human Brain Mapping Journal of Neurology, Neurosurgery & Psychiatry Journal of Neuroscience Methods Neuroimage: Clinical Neuron Neurorehabilitation & Neural Repair Restorative Neurology & Neuroscience

CONFERENCE PRESENTATIONS

- **Duncan, ES** & Small, SL. (2014). Disrupted functional connectivity of left pars opercularis during viewing of animated vs. human speech. Presented at Society for Neurobiology of Language.
- Yang, J, Andric, MJ, Duncan, ES, Holt, A, Hasson, U, Cooper, EA, & Small, SL. (2013). Topdown modulation of brain networks during discourse comprehension. Presented at Society for Neurobiology of Language.
- Schmah, TS, Duncan, ES, Yourganov, G, Zemel, RS, Small, SL & Strother, SC. (2013). Predicting language recovery after stroke from variability of performance and complexity of functional connectivity. Presented at Canadian Stroke Congress.
- Schmah, TS, **Duncan, ES,** Yourganov, G, Zemel, RS, Small, SL & Strother, SC. (2013). Predicting language recovery after stroke using variability of performance and complexity of functional connectivity. Presented at Organization for Human Brain Mapping.
- Schmah, TS, **Duncan, ES,** Yourganov, G, Zemel, RS, Small, SL & Strother, SC. (2013). Predicting language recovery after stroke using variability of performance and complexity of functional connectivity. Presented at Rotman Research Institute Conference.
- **Duncan, ES**, Schmah, T, & Small, SL. (2012). IMITATE: an aphasia treatment motivated by motor cortical connectivity. Presented at Academy of Aphasia.
- **Duncan, ES**, Hyun, JM, & Obler, LK. (2011). Speech perception in noise in an older population: The relationship with working memory and response inhibition. Presented at CUNY Graduate Center Student Research Day.

CLINICAL EXPERIENCE

Per Diem Speech Pathologist	2012 - 2016
Orange County Global Medical Center	
Santa Ana, CA	
Senior Speech Pathologist	2008 - 2011
Mount Sinai Affiliate at Queens Hospital Center	
Jamaica, NY	
Speech Pathologist	2003 - 2008

Mount Sinai Affiliate at Queens Hospital Center	
Jamaica, NY	
Speech Pathologist, Clinical Fellowship Year	2002 - 2003
Professional Service Centers for the Handicapped Article 16 Clinic Astoria, NY	

CLINICAL CREDENTIALS

Licensed Professional Speech-Language Pathologist	2012 – December 2017
State of California	
SP #19578	
Licensed Professional Speech-Language Pathologist	2003 – December 2012
State of New York	
NYS #014259 (Inactive)	
Certificate of Clinical Competence	2003 – December 2019
American Speech Language Hearing Association	
ASHA #12092607	
Basic Life Support Provider	2012 – May 2018
American Heart Association	-

AWARDS & HONORS

Kathy Alberti Prize	2016
School of Social Sciences, UC Irvine	
Saffran Student Scholar Award	2015
Eleanor M. Saffran Center for Cognitive Neuroscience, Temple University	
UC Irvine Pedagogical Fellowship	2015
Center for Engaged Instruction, UC, Irvine	
CENCE Graduate Merit Fellowship	2014
Center for Cognitive Neuroscience and Engineering, UC Irvine	
Award for Continuing Education	2014
American Speech Language Hearing Association	
Saffran Student Scholar Award	2013
Eleanor M. Saffran Center for Cognitive Neuroscience, Temple University	
John I. Yellott Scholar Award	2012
Department of Cognitive Sciences, UC Irvine	
Caroline Mossip Scholarship	2012
Mensa Education and Research Foundation	
Associate Dean's Fellowship	2012
School of Social Sciences, UC Irvine	
Social Science Teaching Assistant Award	2011
School of Social Sciences, UC Irvine	
Award for Continuing Education	2011
American Speech Language Hearing Association	
Award for Continuing Education	2009
American Speech Language Hearing Association	

M.A., with honors	2002
Queens College, City University of New York	
B.A., Summa Cum Laude, Departmental Honors	2000
Queens College, City University of New York	
Phi Beta Kappa Honor Society	2000
Queens College, City University of New York	

SERVICE

Scientific consultation	2015 - 2016
Disability Services Center, UC Irvine	
Teaching consultation/mentorship	2015 - 2016
Center for Engaged Instruction, UC Irvine	
Teaching Assistant Professional Development Program	September 2015
School of Social Sciences, UC Irvine	
Ad hoc clinical course reviewer	2012 - 2015
speechpathology.com	
Cleft palate surgical mission	November 2009
Palestine, Middle East	

AFFILIATIONS

American Speech Language Hearing Association, Member ASHA Division 2, Neurophysiology and Neurogenic Speech Disorders, Member Society for Neurobiology of Language, Member

ABSTRACT OF THE DISSERTATION

Neurological and Behavioral Predictors of Aphasia Recovery

By

E. Susan Duncan

Doctor of Philosophy in Cognitive Neuroscience University of California, Irvine, 2016 Professor Steven L. Small, Chair

This thesis reports behavioral and neurological results of a novel form of aphasia therapy – based on imitation of audiovisual speech – that was motivated by neurophysiological findings in human and non-human primates. The six-week intensive therapy program was completed by nineteen participants with post-stroke aphasia.

Participants demonstrated significant improvement on the practiced repetition task and generalization to other tasks. Measures included subtests of the Western Aphasia Battery-Revised (Aphasia Quotient, Cortical Quotient, Repetition, Naming and Word Finding) and specific characteristics of narrative production (number and percent correct information units).

This research investigated the role of performance variability within individuals during repetition to predict improvement with practice. We found that individuals demonstrating greater variability at baseline made greater improvement following treatment. Additionally, changes in variability over the course of treatment were negatively correlated with changes in performance. The more participants improve, the more consistent (less variable) their performance.

On the narrative task, significant improvement was positively correlated with the number of

xvi

therapy sessions that were completed, a result not explained by other traits that are often associated with aphasia recovery, such as lesion size and time since onset. The imitation therapy generalizes to other tasks, and more practice confers greater benefit.

With functional neuroimaging during the resting state, we discovered two inter-related functional predictors of improvement in narrative production. Using a sliding window approach, we investigated the dynamic nature of resting state networks (RSNs) as they change over the course of therapy. An increase in the amount of time spent in one of the states – a state characterized by minimal correlation among the identified RSNs – predicted improvement on the narrative task. We interpreted this finding as evidence for adaptive segregation among the RSNs.

Using a graph theoretical approach, we found a second functional predictor. By grouping the RSNs into communities, we examined network changes representative of segregation in the brain, and found increased RSN modularity to be positively correlated with behavioral improvement.

These findings of previously unexplored behavioral and neurological changes and predictors associated with post-therapy recovery are consistent with an emerging approach to personalized (precision) aphasia treatment.

Chapter 1: Introduction

1.1 Aphasia

Aphasia is a multimodal disorder resulting from injury to parts of the brain – particularly in the left hemisphere – that contribute to our human capacity for language. Such damage is most commonly caused by stroke affecting the distribution of the left middle cerebral artery yet can have other etiologies such as tumor or trauma, and in rare cases, right hemisphere injury (e.g., (Lee et al., 2016)). Although the presence of aphasia is not indicative of a generalized intellectual impairment, it is associated with deficits in other cognitive domains, such as working memory (Kasselimis et al., 2013) and problem solving (Baldo, Paulraj, Curran, & Dronkers, 2015).

Approximately one third of individuals with acute ischemic stroke (Engelter et al., 2006) and one fifth of those in chronic stages (Wittenauer & Smith, 2012) are afflicted. One million Americans currently have aphasia (NIDCD, 2010), and this number is increasing as life expectancy extends and new medical interventions offer greater probability of survival following neurological injury (Code, 2010). People with aphasia are less likely to return to work than other stroke survivors (Tanaka, Toyonaga, & Hashimoto, 2014), contributing significantly to the 33 billion dollars of direct and indirect expenses that stroke costs the American public annually (Mozaffarian et al., 2016).

There is much variability in aphasia recovery and limited ability to predict duration or severity from acute to chronic stages. Severity during the first three days following stroke has been reported to predict performance three months later (Lazar et al., 2010), although individuals with initially severe deficits are more heterogeneous in their recovery (Lazar, Speizer, Festa, Krakauer, & Marshall, 2008). Improvement is greatest in the first three months (Laska,

Hellblom, Murray, Kahan, & Von Arbin, 2001) and plateaus approximately one year following onset (Basso, 1992), although patients can demonstrate measurable gains with intervention even a decade post onset with speech-language therapy (Duncan, Schmah, & Small, 2016; Szaflarski et al., 2008). Aphasia is, today, still incurable, and therapy typically provides only modest benefit (Brady, Kelly, Godwin, & Enderby, 2012). Yet the field of aphasia intervention remains relatively young; it has only been for the past century that aphasia has been seen as a treatable condition with established pathophysiology, despite several millennia of documented cases.

1.2 History of Aphasia

1.2.1 Ancient Egypt

The earliest record of the written word "brain", and the earliest descriptions of aphasia, can be found in the Edwin Smith surgical papyrus (see Figure 1.1), dating from 1700 BCE (Minagar, Ragheb, & Kelley, 2003). This ancient medical text describes a patient who is frustrated by the inability to speak following traumatic head injury. This disorder is described as untreatable.



Figure 1.1 The Edwin Smith Papyrus. Document containing the oldest known references to aphasia. Image in public domain.

1.2.2 Ancient Greece

"Speechlessness" was first connected to brain injury in the Hippocratic Corpus of the late fifth/early fourth century BCE (Prins & Bastiaanse, 2006), although the contemporary Aristotle believed the brain to simply be a cooling unit for the passions of the heart, which in turn was thought to be the source of the nerves (Finger, 2000). Around 200 CE, the Greek physician Galen better developed this understanding through vivisection. Galen found that severing the laryngeal nerves and compressing the exposed brains of live animals caused their cries and squeals to cease, leading him to believe that the soul, housed in the ventricles, traveled through hollow nerves to move the body (Prins & Bastiaanse, 2006).

1.2.3 Europe: Middle Ages and Early Modern Period

While there are many descriptions of aphasic symptoms between the fifth and nineteenth centuries, there are a few of particular note in demonstrating the emergence of a more accurate biological understanding of the disorder.

The first of these is the Renaissance description of Italian physician and professor Antonio Guainerio. He described two patients with aphasia. In keeping with the tradition of Galen, he attributed their deficits to a phlegmatic blockage of the posterior ventricle, thereby disrupting memory (Benton, 1964). Biological inaccuracy notwithstanding, Guainerio's fifteenth century diagnosis may be the earliest known localization of a language deficit within the brain.

German physician Johann Schenck von Grafenberg was the first figure to record a clear distinction between motor speech, as lingual paralysis, and language, conceptualized as a function of memory (Luzzatti & Whitaker, 1996). Despite this, treatment of the tongue for the purposes of curing aphasia was widely practiced until the middle of the nineteenth century (Prins

& Bastiaanse, 2006). While by no means the only example, Schenck's sixteenth century work is also notable for a shift from a ventricular to a brain tissue basis for language (Eling & Whitaker, 2009).

In 1770, another German physician, Johann Gesner, wrote a landmark work entitled Die Sprachamnesie (language amnesia), which provided extensive behavioral descriptions of aphasia as well as attempts to elucidate pathogenic contributions (Prins & Bastiaanse, 2006). He described and connected impairments of speech with impairments of writing, addressing the existence of a shared language system underlying both skills (Luzzatti, 2002). Gesner's work may be viewed as a landmark in aphasiology, and one that paved the way for the better-known, and more biologically grounded, work of the following century.

1.2.4 Europe: Nineteenth Century

1.2.4.1 Phrenology

The true origins of the neurobiology of aphasia may be found in the work of nineteenth century European physicians. One of the earliest of these was Viennese neuroanatomist Franz Joseph Gall around the turn of the century. Gall interpreted bumps and depressions on the surface of the skull as indications of the development of the underlying neural tissue, and thus as evidence of the presence of personality traits attributable to the underlying brain region (Gall, 1825). While Gall's phrenological methods have been rightly discredited as pseudoscience, he deserves credit for introducing the idea that various regions of the brain provide unique contributions to different forms of information processing. Phrenology served as a precursor to the cerebral localization of mental processes, including language, then thought to be situated behind the eye (see Figure 1.2).



Figure 1.2 Phrenological Chart. Published in Fowler & Fowler (1859); image in public domain.

1.2.4.2 Lesion Studies

It should be acknowledged that French neurologist Marc Dax made the discovery of left hemisphere dominance for language in the early nineteenth century, although this was not published until 1863 (Cubelli & Montagna, 1994). Thus, the next significant development was Pierre-Paul Broca's presentation of a case study of a patient with nonfluent aphasia before the Anthropological Society of Paris in 1861, in which he reported a lesion primarily occupying the posterior portion of the third convolution of the left frontal lobe (Broca, 1861; see Figure 1.3), citing the error of the phrenologists and declaring this area (now named Broca's area) to be the seat of spoken language.



Figure 1.3 Brain of Louis Victor Leborgne, Broca's Original Patient. Reproduced from Dronkers, Plaisant, Iba-Zizen, & Cabanis (2007) with permission from Oxford University Press.

The German anatomist and neuropathologist Theodor Meynert expanded on Broca's findings through autopsies documenting that aphasia could be associated with damage to other left hemisphere brain regions in the vicinity of the Sylvian fissure, including a region now named for his student, Carl Wernicke. Wernicke, in turn, expanded on this by conceptualizing the perisylvian cortex and insula as a "speech center" and developing a model of regions critical to language with connecting association pathways (Wernicke, 1874), pioneering the first of the connectionist theories of language (see Figure 1.4).



Figure 1.4 Carl Wernicke's Language Network. Primary auditory cortex is represented by a, Broca's area by b. These regions are shown as connected to each other and to cranial nerve nuclei in the brainstem. One of several language networks drawn in Wernicke (1874). Image in public domain.



Figure 1.5 Lichtheim's "House" Model. Language system conceptualized through the following regions/functions: B= concept center (broadly distributed); M= motor image center (Broca's area); m= motor output (speech motor cortices); A= auditory image center (Wernicke's area); a= acoustic-sensory input (primary auditory cortices). Lesions at numbered sites are conceptualized as: 1= Broca's aphasia; 2= Wernicke's aphasia; 3= conduction aphasia; 4= transcortical motor aphasia; 5= apraxia/ dysarthria; 6= transcortical sensory aphasia; 7= pure word deafness. Figure from Lichtheim (1885). Image in public domain.

Further evolution of this idea was demonstrated by Ludwig Lichtheim's "house" model (1885), which elaborated the role of cortical connections and identified "disconnection" syndromes (See Figure 1.5) and Joseph Jules and Augusta Déjerine's identification of the role of white matter connections in his writing on pure alexia without agraphia (Déjerine & Déjerine-Klumpke, 1895). Pioneering French neurologist Jean Martin Charcot developed his own anatomical scheme for localizing various aphasia symptoms (Bernard, 1889; see Figure 1.6). Sigmund Freud objected to these conceptualizations of language as a property arising from isolated centers and their interspersed connections, positing that aphasia does not result in the absolute deficits such models predict and that the entire perisylvian cortex provides crucial, if heterogeneous, contributions to language (Freud, 1891), although this work was largely ignored.



Figure 1.6 Localization of Aphasia Symptoms According to Charcot. Labeled sulci: 1= Sylvian fissure; 2= central sulcus; 3= intraparietal sulcus; 4= superior temporal sulcus; 5= parietoccipital sulcus. Labeled gyri: F1/2/3 = superior/middle/inferior frontal gyrus; Fa= precentral gyrus; T1/2/3= superior/middle/inferior temporal gyrus; Ps/i= superior/inferior parietal lobule; Pa= postcentral gyrus; Pc= angular gyrus. Shaded areas corresponding to deficits: F2=agraphia; F3= Broca's aphasia; T1= pure word deafness/Wernicke's aphasia; Pi= alexia. Figure from Bernard (1889). Image in public domain.

1.2.5 Twentieth Century

French neurologist Pierre Marie began the twentieth century with the declaration that "the third left frontal convolution plays no special role in language", citing cases of damage to this region without associated aphasia as well as cases of Broca's aphasia in which this region was spared (Tesak & Code, 2008). Still, in the mid-twentieth century, the Wernicke-Lichtheim model was resurrected by Norman Geschwind (Geschwind, 1970), and it remains perhaps the predominant model in aphasia theory today. The behavioral classifications (see Table 1.1) associated with the model can be identified using clinical tools such as the Boston Diagnostic Aphasia Examination (BDAE; Goodglass & Kaplan, 1983) and the Western Aphasia Battery (WAB; Kertesz, 1982). While the correlation is imperfect, the various classifications are often associated with different lesion patterns (Willmes & Poeck, 1993).

	Fluency	Content	Comprehension	Repetition	Naming
Broca's	Poor	Good	Fair-Good	Poor-Fair	Fair-Good
Wernicke's	Good	Poor-Fair	Poor-Fair	Poor-Fair	Poor-Fair
Conduction	Good	Good	Good	Poor-Fair	Fair
Global	Poor	Poor	Poor	Poor	Poor
Trans. Motor	Poor	Good	Fair-Good	Good	Fair-Good
Trans. Sensory	Good	Poor-Fair	Poor-Fair	Good	Poor-Fair
Mixed Trans.	Poor	Poor	Poor	Fair-Good	Poor
Anomic	Good	Good	Good	Good	Fair

Table 1.1 Typical Aphasia Taxonomy. Classifications based on Goodglass & Kaplan (1983), Hillis (2007), and Kertesz (1982). Trans= Transcortical.

The importance of the lesion studies of the nineteenth century and the neuropsychological classifications of the twentieth century to the study of aphasia cannot be overstated. Yet, the types of questions we ask will always inform the types of answers we find. Lesion studies will invariably present us with localizationalist answers, as lesions are inherently localized. Similarly, neuropsychological classifications impose categories on behavioral continua.

In the latter part of the twentieth century, functional neuroimaging revolutionized our thinking about language and aphasia. Positron emission tomography (PET), functional magnetic resonance imaging (fMRI), and magnetoencephalography (MEG) demonstrated activation during language tasks that was both bilateral and present in cortical and subcortical regions not traditionally associated with language, underscoring the complexity of even the most basic tasks (Hillis, 2007). Further, intraoperative cortical stimulation mapping (Ojemann, Ojemann, Lettich, & Berger, 1989) and noninvasive transcranial magnetic stimulation (Epstein, 1998) allowed the introduction of "virtual lesions" and the investigation of anatomical hypotheses not previously testable in humans. These relatively new methods have significantly informed our understanding of aphasia and the neural organization of language.

1.3 Present Day

Ongoing advances in neuroimaging techniques and statistical methods are changing the kinds of questions we can answer, and perhaps more importantly, ask, about aphasia. Early neuroimaging studies demonstrated a tendency to pinpoint circumscribed regions of activation correlated with language and other higher-order functions by using high activation thresholds, such that a single region might remain (e.g., Small et al., 1996). Many early studies that did not force such localized activation through statistical means nevertheless described their results in focal terms (e.g., Binder et al., 1994). Currently, these perspectives are evolving.

Beyond localizationalism lies an appreciation of the brain as the product of overlapping and interconnected circuits. As our analytical methods advance in tandem with our theoretical conceptualization, our investigations become more complex, reflective of the characteristics of the subject matter. Thus, the study of the brain basis of aphasia can now transition from strictly region of interest (ROI) analyses to the realm of connectivity. We can assess structural connectivity by examining white matter pathways through use of diffusion tensor imaging (DTI) on an individual basis and through atlases derived from multi-subject DTI. In functional connectivity, the brain is typically understood through correlations between the time courses of anatomically separate regions, similarities of which are believed to reflect functional integration. The resulting connectivity measures commonly reflect average statistical dependencies collapsed across time, while more advanced analyses yield dynamic measures that express how functional connectivity varies as a function of time. Additionally, graph theoretical analysis of neural networks, whether functional or structural, allows insights into organizational principles that govern the flow of information by modeling pairwise relationships among brain regions (Bullmore & Sporns, 2009). Such quantitative methods, with their focus on interconnections

rather than isolated regions, currently represent our strongest tools for investigations into the neurobiology of language and aphasia.

1.4 Rationale for this Thesis

With consideration of the changing nature of our understanding of aphasia and our methods for investigating the brain, this dissertation identifies neurological and behavioral predictors and correlates of an individual's ability to benefit from aphasia therapy. The finding that all aphasia therapies offer essentially equivalent benefit (Brady et al., 2012) may be due, in part, to the lack of individualized criteria for selecting a particular type of therapy or the failure to track progress on a biological, as well as behavioral, basis. Thus, treatment studies that apply the same therapy to a sample of stroke patients and offer a single effect size as summary may be attenuating findings by averaging results across a diverse group. Consistent with contemporary notions of precision (personalized) medicine, it is likely that we will make significant strides in the ability to benefit those with aphasia when we have explanatory power for the considerable heterogeneity in individual recovery and response to treatment.

It is increasingly recognized that, as a biological disorder, attention must be paid to the role of biology in aphasia treatment (Small & Llano, 2009), with a potential for the most significant gains when we are able to regrow brain tissue. At that point, the purpose of therapy will be to reestablish adaptive circuits incorporating new tissue into the existing system. Until this time, however, we continue to expand our mechanistic understanding of aphasia recovery for the purpose of enhancing current benefit, as well as laying the groundwork for future intervention.

The imitation-based aphasia treatment work described in this thesis is based on a biological rationale described in Chapter 2. The therapy itself, and the methods with which it has been

investigated, are described in Chapter 3. Chapter 4 describes a behavioral analysis focused on the role of intra-individual variability in improvement on a practiced task. Chapter 5 discusses the generalization of benefit associated with the therapy to a narrative production task. Chapter 6 focuses on a data-driven investigation into the dynamic functional connectivity of resting state networks as it relates to behavioral gains post-therapy, and Chapter 7 describes a hypothesis-driven validation of the findings described in Chapter 6 through the use of graph theoretic methods. Chapter 8 concludes the dissertation with a summary of findings and a brief exploration of future directions motivated by the present work.

Chapter 2: Imitation-Based Aphasia Therapy

2.1 Repetition and Imitation in Aphasia

In 1683, the German physician Peter Rommel was the first to write about repetition deficits in a patient with nonfluent aphasia (Benton & Joynt, 1963). Imitation has since been a key diagnostic and treatment tool for such acquired language disorders. All popular standardized instruments for the assessment of aphasia, such as the Western Aphasia Battery-Revised (WAB; Kertesz, 2006), Boston Diagnostic Aphasia Examination (Goodglass & Kaplan, 1983), and Aachen Aphasia Test (Huber, Poeck, & Willmes, 1984) include repetition ability in their classification scheme.

Imitation was fundamental to the nascent field of aphasia therapy at the turn of the 20th century, and it remains so today (Duffy, 1995). This chapter describes the neurobiological rationales for, and current implementations of, imitation in aphasia therapy. "Imitation" and "repetition" are used interchangeably, and modes of stimulus presentation are clarified as needed. Additionally, it should be noted that acquired apraxia of speech, a motor planning deficit frequently accompanying nonfluent aphasia (Duffy, 1995), is not specifically addressed in this chapter.

2.2 Neurobiological Approaches to Language and Aphasia

The earliest approaches were based on behavioral and educational principles, and this philosophy dominates aphasia therapy today (Small, 2004). Although aphasia is a neurological impairment resulting from brain damage, typically stroke (Ellis, Dismuke, & Edwards, 2010), treatment programs are rarely biologically motivated. Since the end of the 20th century, studies

in aphasia have been relying less on applied psychology and linguistic models, instead seeking to link observed deficits to impairments in the underlying neural systems (Blumstein, 1997). Rehabilitation of the behavioral deficits of aphasia must target the plasticity and repair of affected biological systems. Two main biological models characterizing these systems are considered here, the human mirror system and the dual-stream hypothesis for speech.

2.3 Mirror Neuron System

2.3.1 Macaque

Mirror neurons were discovered serendipitously during single-cell recordings of hand motor representations in the macaque. Rizzolatti and colleagues found neurons firing in premotor cortex (area F5) in a monkey during observation of the experimenter (di Pellegrino, Fadiga, Fogassi, Gallese, & Rizzolatti, 1992), although the monkey itself was motionless (see Figure 2.1). Further investigation found that individual neurons were active during observation and execution for specific hand or mouth movements (Ferrari, Gallese, Rizzolatti, & Fogassi, 2003; Gallese, Fadiga, Fogassi, & Rizzolatti, 1996). Additional mirror neurons possessing such visuomotor properties were subsequently identified in the inferior parietal region of the macaque (Fogassi, Gallese, Fadiga, & Rizzolatti, 1998), primarily in subcomponents PF and PFG (Rozzi, Ferrari, Bonini, Rizzolatti, & Fogassi, 2008), which have strong anatomical projections to the ventral premotor cortex (F5). These findings led to the suggestion of a functional "mirror" network (Rozzi et al., 2006).



Figure 2.1 Visual and Motor Responses of a Mirror Neuron. Behavioral situations are depicted in upper panels, raster plots representing neural discharge in middle panels, and histograms of relative response in lower panels. A, neuron discharging for monkey's observation and execution of grasping of food. B, neuron does not discharge when observing grasping food using tool. C, neuron discharges for monkey grasping food in darkness. Reproduced from Gallese et al., (1996) with permission from Elsevier.

The existence of mirror neurons immediately prompted hypotheses about their role in action recognition (Rizzolatti, Fadiga, Gallese, & Fogassi, 1996). Further support for this has been provided by the discovery that some mirror neurons in macaque F5 have auditory as well as visual and motor properties (Kohler et al., 2002), firing in response to observation and execution of actions, as well as for sounds associated with those actions. This multimodal integration at the level of a single cell may form the basis for action understanding and motor learning (Jeannerod, 1994).

2.3.2 Human

Ethical considerations prohibit systematic human studies investigating individual mirror neurons. However, support for the existence of a human parieto-frontal mirror neuron system is converging from behavioral, neurophysiological, and brain imaging studies (Small, Buccino, & Solodkin, 2012). The "direct matching hypothesis" postulates that imitation is subserved by simple neural mechanisms mapping observed actions onto internal motor representations of the same action by neurons with mirror properties, populations of which are more strongly activated for actions elicited by preceding observations of the same action (Iacoboni et al., 1999).

2.3.2.1 Behavioral

Behavioral studies demonstrate motor facilitation when action execution immediately follows observation, supporting the existence of a mirror system in humans. Finger movements are faster if the stimulus cue is a modeled finger movement compared to an unrelated symbol (Brass, Bekkering, Wohlschlager, & Prinz, 2000). Response speed further increases as the modeled movement more closely resembles the target, even when the stimulus image is flipped

upside-down (Brass, Bekkering, & Prinz, 2001). Grasping response speed increases when subjects are shown a picture of a hand with optimal orientation for their own final hand position (Craighero, Bello, Fadiga, & Rizzolatti, 2002). On language tasks, response times for plausibility judgments are faster when the action response required is similar to the action described in the stimulus sentence (Glenberg & Kaschak, 2002). These behavioral findings suggest that action perception influences the functioning of one's own motor system.

2.3.2.2 Neurophysiology

2.3.2.2.1 Electroencephalography

Studies using electroencephalography (EEG) demonstrate a central mu rhythm in the alpha frequency range (8–13 Hz) that is present when a subject is at rest. This rhythm, detected in electrodes overlying sensorimotor cortices, is suppressed during action observation as well as motor activity, as was first described in 1954 (Cohen-Séat, Gastaut, Faure, & Heuyer, 1954). These findings have since been replicated for observation, imitation, and execution of actions with EEG (Altschuler et al., 2000; Cochin, Barthelemy, Roux, & Martineau, 1999). Responses measured via implanted subdural electrodes also show a reduction in absolute power in the alpha band over primary motor cortex and Broca's area during both observation and execution of finger movements (Tremblay et al., 2004).

Suppression of this mu rhythm is stronger for object grasping than for movements that are not goal-oriented in adults (Muthukumaraswamy & Johnson, 2004) and in children (Lepage & Theoret, 2007). A precursor to the mu rhythm, that is similarly reduced during both observed and executed grasping movements, is also found in infants, albeit in a lower frequency range (6–9 Hz; Marshall, Young, & Meltzoff, 2011). It has been proposed that such early-developing mirror
neurons may underlie childhood imitation, language acquisition, and the development of other social and cognitive functions (Williams, Whiten, Suddendorf, & Perrett, 2001).

2.3.2.2.2 Magnetoencephalography

Magnetoencephalography (MEG) permits neurophysiological monitoring of a band of activity (15–25 Hz) that is present in the precentral motor cortex during rest. This activity is suppressed during movement and can also be suppressed by stimulating the median nerve of the upper extremity, which allows the study of its rebound in varying contexts. This rebound is extinguished when stimulation is followed by object manipulation and is also significantly reduced when stimulation is followed by passive observation of the same task (Hari et al., 1998). The rebound is not affected by viewing other moving stimuli, linking action observation to the motor system.

2.3.2.2.3 Transcranial Magnetic Stimulation

Transcranial magnetic stimulation (TMS) manipulates cortical responses by either inducing or inhibiting action potentials in populations of neurons. Excitatory TMS applied over the contralateral cortical region that optimally induces contractions of hand or arm muscles produces measurable motor-evoked potentials (MEPs; Kamen, 2004), which provide an objective measure of motor excitability. TMS-induced MEPs increase when observing grasping actions and arm movements that use the same muscles that are being measured (Fadiga, Fogassi, Pavesi, & Rizzolatti, 1995), as demonstrated in Figure 2.2. These results have been replicated using observation of handwriting and arm movements compared to rest (Strafella & Paus, 2000).



Figure 2.2 Motor Evoked Potentials (MEPs) During Action Observation. MEPs induced by transcranial magnetic stimulation (TMS) were recorded from four hand muscles under four different conditions. From left to right, the four muscles are: EDC= extensor digitorum communis; FDS= flexor digitorum superficialis; FDI= first dorsal interosseous; OP= opponens pollicis. Within each muscle, from left to right, the four conditions are: grasping observation, object observation, arm movement observation, and detection of dimming of a visual stimuli. Reproduced from Fadiga et al. (1995) courtesy of the American Physiological Society.

2.3.2.2.4 Single-cell Recording

These neurophysiological findings suggest the influence of mirror properties on the human motor system extends to primary motor areas, in addition to the postulated premotor homologues of the frontal regions where macaque mirror neurons have been identified. Greater extension still has been proposed. In single-cell recordings from subjects with medically intractable epilepsy, a significant number of neurons in the supplementary motor area (SMA) and the medial temporal lobe (MTL) respond to the observation and execution of a single action (Mukamel, Ekstrom, Kaplan, Iacoboni, & Fried, 2010). These regions, with clinical rather than theoretical determination of electrode placement, have not previously shown mirror properties in animal experimentation. Additionally, some neurons change their baseline firing rates but in a manner paradoxical to macaque studies; they respond with increased excitation for execution and suppressed firing rate for observation. The authors propose that these findings may provide evidence of multiple mirroring systems in the brain, with reduced activity of some neurons during observation playing a role in suppressing socially inappropriate imitation.

2.3.2.3 Brain Imaging

Brain imaging studies permit greater spatial localization of mirror properties in humans. Early evidence for these properties comes from a positron emission tomography (PET) study contrasting object observation with action observation (Rizzolatti, Fadiga, Matelli, et al., 1996), which demonstrates activation in response to observation of grasping movements in Broca's area and left hemisphere temporal regions (middle temporal gyrus, superior temporal sulcus). Mirror properties have since been demonstrated via functional magnetic resonance imaging (fMRI) in two regions of the human brain active during both passive observation and imitation of finger movements, the left frontal operculum of Broca's area and the right anterior parietal region (Iacoboni et al., 1999).

Broca's area, the putative frontal oral-motor and speech area in localizationist language models, plays a role in hand motor representation (Binkofski et al., 1999) and is typically identified as the human homologue of macaque F5 in which mirror neurons have been recorded (Rizzolatti, Fadiga, Gallese, et al., 1996). However, the consistent finding of activation in response to observation of hand and arm actions (Decety et al., 1997; Grafton, Arbib, Fadiga, & Rizzolatti, 1996) in Broca's region, in combination with its long history in the neuroscience of language, has raised the question of whether this increased neural response is perhaps an epiphenomenal artifact of internal speech during these tasks.

To resolve this issue, fMRI tasks have explored brain activation patterns when actions are performed by various body parts. Investigation of response to observation of actions performed

by the hand, foot, or mouth reveals a somatotopic organization of premotor cortex similar to that found in the primary sensory and motor cortices, with ventral mouth movements and dorsal foot movements (Buccino et al., 2001). Similar organization is found in the posterior parietal lobe for object-related actions (see Figure 2.3). These findings ground single neuron measures from macaque in a broader network of motor circuitry underlying both action observation and execution in humans.



Figure 2.3 Somatotopic FMRI Activation During Action Observation. Above, observation of non-object-related actions. Below, observation of object-related actions. Color scheme is as follows: red= activation foci during observation of mouth movements; green= activation foci during observation of hand movements; blue= activation foci during observation of foot movements. Somatotopically organized activation is present in premotor and parietal cortices. Figure reproduced from Buccino et al. (2001) with permission from John Wiley and Sons.

Macaque studies find that mirror neurons fire for observation of grasping only in the presence of a graspable object, even if it is not visible (Umilta et al., 2001). It has therefore been

suggested that this system is not simply encoding movements, but goal-oriented motor acts (Gallese et al., 1996). In support of this, left frontal and temporal regions are activated for meaningful, but not meaningless, actions in human PET scans (Decety et al., 1997). Similarly, actions embedded in contexts show increased fMRI activation in the posterior inferior frontal gyrus (IFG) and ventral premotor cortex (PMv) compared to viewing either the action or the context alone (Iacoboni et al., 2005). This is especially pertinent to the discussion of language, a goal-directed behavior in which we use our motor systems to transmit meaningful messages, with mirror neurons bridging the gap between "doing" and "communicating" (Rizzolatti & Arbib, 1998).

2.4 Mirror Neuron System and Language

Language, as a uniquely human property, lacks an ideal animal model, and biological theories of language cannot be directly tested. However, indirect evidence supports a role for the mirror neuron system in human language ability, both phylogenetically through evolutionary selection processes (Rizzolatti & Arbib, 1998) and ontogenetically in facilitating child language acquisition (Kuhl & Rivera-Gaxiola, 2008). It is suggested that mirror deficits may underlie developmental disorders of language and social interaction, notably autism (Williams et al., 2001), although such issues are controversial and beyond the scope of this text.

2.4.1 Perception and Production of Articulated Speech

The observation–execution or direct matching hypothesis suggests that perception of an action, including speech, depends on previous experience with producing that actions or sound (Iacoboni et al., 1999). One model of this is the "inverse-forward model pairs" (IFMPs; Skipper,

Nusbaum, & Small, 2006). These are mechanistic components of the mirror system, in which speech sounds, whether heard or observed, are transformed into corresponding articulatory gestures (inverse) as well as predictions of motor behaviors to be observed next (forward) with resultant sensory consequences affecting perception (see Figure 2.4). These IFMPs operate in the multisensory contexts in which we experience language, consisting of acoustic signals and also the visual cues of oral, facial, manual, and body gestures, particularly when auditory information is distorted or ambiguous. The inverse modeling in this theory predicts that speech perception should produce measurable responses of the speech motor system similar to how it would be engaged during production of that observed speech.



Figure 2.4 Diagram of Inverse-Forward Model Pairs. Observed talker in center. Inverse models (solid lines) are derived from seeing and hearing the talker's facial (light gray) and hand (dark gray) gestures; these models specify the motor goals of the observed movements, facilitated by mirror properties, and can map the talker's behaviors to a plan in the observer's own motor system. Forward models (dashed lines) predict the sensory consequences of implementing those motor plans. Combining these two models into an inverse-forward model pair disambiguates sensory input and improves speech perception. STp= posterior superior temporal areas; SMG= supramarginal gyrus; SI/SII= primary /secondary somatosensory cortices; PM/M1= premotor/primary motor cortices; POp= pars opercularis. Reproduced from Skipper et al. (2006) with permission from Cambridge University Press.

2.4.1.1 Neurophysiology

Listening to the lingual trill /r/ results in significantly increased amplitude in tongue muscle MEPs in neurologically intact participants compared to observation of the nonlingual labiodental phoneme /f /, and also when MEPs are measured from a thumb muscle rather than the tongue (Fadiga, Craighero, Buccino, & Rizzolatti, 2002). This response is more pronounced for real words compared to pseudowords, suggesting that past experience with perception and production influence the motor system. Similarly, increased MEPs in oral muscles, but not finger muscles, are found during either listening to connected speech or viewing silent video of speech-related lip movements (Watkins, Strafella, & Paus, 2003), highlighting multisensory contributions to speech perception. This contrast is not found for nonspeech control conditions, including nonverbal sounds and observation of eye movements. Consistent with widely accepted lateralization theories of speech and language, this follows stimulation of the left, but not the right, hemisphere.

2.4.1.2 Brain Imaging

2.4.1.2.1 Motor Regions Engaged During Speech Perception

Brain imaging studies provide indirect evidence of a relationship between speech observation and execution. Speech motor regions are engaged in response to audiovisual (Skipper, van Wassenhove, Nusbaum, & Small, 2007; Watkins et al., 2003), visual (Nishitani & Hari, 2002), and auditory (Fadiga et al., 2002; Tettamanti et al., 2005) speech perception. Figure 2.5 shows regions active during both syllable production and passive observation of audiovisual, visual, or auditory speech. Bilateral brain activation is present in premotor regions and Broca's area during silent lip-reading (Buccino et al., 2004), indicating that frontal motor cortices are activated in response to multimodal aspects of language.



Figure 2.5 Regions of Overlap for Speech Perception and Production. Logical conjunction analyses from fMRI acquired during production and perception of the same syllables. Orange indicates regions of activation overlap between production and perception (thresholded at p<0.05). Blue indicates regions active during passive perception but not during production (thresholded at p<0.05). Stimuli for speech perception were audiovisual (A), visual only (B), or auditory only (C). Reproduced from Skipper et al. (2007) with permission from Oxford University Press.

Structural equation modeling (SEM) applied to neuroimaging permits the determination of covariances, or weights, among brain regions with known anatomical connectivity (McIntosh & Gonzalez-Lima, 1994). SEM of these effective paths during observation and imitation of audiovisual syllables during fMRI shows common functional connections shared for both these tasks, differing in connection strength but sharing the same essential structure (Mashal, Solodkin, Chen, Dick, & Small, 2012). Like imitation, speech observation engages dorsal and ventral premotor cortices and primary motor cortex.

2.4.1.2.2 Temporal and Auditory Regions Engaged During Speech Production

Regions in posterior auditory cortex are active during speech production, including covert speech, as well as speech perception (Okada & Hickok, 2006; Papathanassiou et al., 2000). Further, posterior lesions of the left temporal cortex, as in Wernicke's aphasia, are associated with errors in verbal expression as well as comprehension deficits (Damasio & Geschwind, 1984). Temporal regions have classically been excluded from the putative mirror neuron system, because there has been no finding of motor activation in the temporal lobe in macaque studies (Keysers & Perrett, 2004). Still, the existence of individual mirror neurons remains poorly defined in humans due to limitations of appropriately noninvasive methods. It is possible that their cortical distribution is more extensive than that in our primate cousins (Mukamel et al., 2010).

2.4.2 Comprehension of Action Language

The human mirror system operates in tandem with low-level sensorimotor aspects of speech, discussed above, and also higher-level language comprehension (Barsalou, 2008). Listening to sentences describing motor activity activates a broad left-lateralized network of frontal, temporal, and parietal regions, as do action observation and execution, which does not occur for sentences that do not encode action (Tettamanti et al., 2005). Listening to or reading action-related language evokes somatotopic motor cortex activation consistent with the described effector performing the action (Aziz-Zadeh, Wilson, Rizzolatti, & Iacoboni, 2006; Hauk, Johnsrude, & Pulvermuller, 2004).

Theories proposing that cognition is grounded in – or superimposed on – basic sensory and motor processes also apply to language. Priming effects are found for objects sharing

affordances, such as a piano and a typewriter, even when the task does not address the object's use (Myung, Blumstein, & Sedivy, 2006). Subjects receiving verbal or visual cues to assume certain hand shapes are faster to respond to the plausibility of action–object pairings congruous with the simulated grasp (Klatzky, Pellegrino, McCloskey, & Doherty, 1989). This difference disappears when a verbal response replaces the motor response, suggesting that the interplay between language processing and the motor system confers the advantage, rather than the semantic relationship.

2.5 Dual Streams for Speech

The dual-stream theory of vision has long been dominant (Mishkin, Ungerleider, & Macko, 1983), with a ventral "what" pathway for object identification and a dorsal "where" or "how" pathway for visuomotor integration (Goodale, 1993). More recently, two stream models have been identified in the study of audition (Rauschecker & Tian, 2000), speech perception (Hickok & Poeppel, 2004), speech production (Hickok & Poeppel, 2007), and sentence comprehension (Friederici, 2009). There is considerable debate among varying theories regarding specific functions, regions, connections, and the role of feedback. Broadly, however, the dorsal stream progresses from temporoparietal regions to frontal premotor areas, whereas the ventral stream progresses through temporal lobes to prefrontal cortex.

The present discussion only superficially describes the putative roles of the two streams to elucidate their role in imitation (see Figure 2.6). The interested reader is referred to the original sources, including those cited here. The streams are typically discussed separately, but it should be understood that this division is an artificial one made for the sake of simplicity rather than accuracy. In actuality, the streams must be integrated for successful functioning, operating

through "cooperative computation" (Fagg & Arbib, 1998). The strongest neurobiological models underpinning language processing in the brain presently comprise the dual-stream model and the mirror neuron system.



Figure 2.6 Dorsal and Ventral Streams for Language. Functional magnetic resonance imaging (fMRI) results for A, repetition (pseudowords > words); B, comprehension (sentences > pseudosentences); and C, both contrasts (colors as in A,B). FOP= deep frontal operculum; F3op/orb/tri= pars opercularis/orbitalis/triangularis of the inferior frontal gyrus; FUS= fusiform gyrus; T1a/p= anterior posterior superior temporal gyrus; T2a/p= anterior posterior middle temporal gyrus. Figure reproduced from Saur et al. (2008) with permission from The National Academy of Sciences.

2.5.1 The Ventral Stream

The ventral pathway is conceived in terms of semantics, extracting meaning from the communicative signal (Saur et al., 2008). It is considered to be bilateral in some models (Hickok & Poeppel, 2007), whereas others identify ventral auditory language pathways only in the left hemisphere (Parker et al., 2005). In the temporal lobe, it includes anterior portions of the superior temporal gyrus and sulcus, the middle and inferior temporal gyri, and the temporal poles

(Skipper et al., 2006). In the macaque, these areas connect to frontal regions including orbitofrontal cortex and pars orbitalis via the uncinate fasciculus, and pars opercularis and pars triangularis via the extreme capsule (Petrides & Pandya, 2009).

Imitation of familiar and meaningful actions is positively correlated with ventral activity in the inferior temporal cortex (Decety et al., 1997). This is consistent with the object identification, or "what," role of the visual ventral stream, which may share connectivity analogous to that of the auditory ventral stream (Seltzer & Pandya, 1978). In contrast, imitation of novel and meaningless actions do not activate these ventral cortices, instead resulting in activation in dorsal parietal and occipital regions (Rumiati et al., 2005).

Words and sentences, which can be conceived of as meaningful "gestures" or speech "objects," also represent the domain of the ventral stream. Consistent with this, temporal lobe atrophy (predominantly anterior and ventral) is associated with semantic dementia, a variant of frontotemporal dementia characterized by progressive deficits in confrontation naming and single word comprehension, as well as loss of the underlying concepts associated with the language (Mummery et al., 2000), while speech repetition remains intact (Gorno-Tempini et al., 2011). This would suggest at most a minimal role for the ventral stream in imitation, although the stimuli used might differentially engage regions and enhance connectivity in ventral areas depending on semantic meaning and social relevance (Kilner, Marchant, & Frith, 2006), such as personal significance, ecological validity, and familiarity.

2.5.2 The Dorsal Stream and Parietal Cortical Connectivity

In contrast with the semantic role of the ventral stream, and analogous to the "where" or "how" role of the visual dorsal stream, the dorsal stream for speech is proposed as a

sensorimotor network mapping sounds onto motor plans to support production (Hickok & Poeppel, 2007). Repetition, especially of meaningless pseudowords, is the prototypical task of dorsal stream function (Saur et al., 2008), as demonstrated in Figure 2.6.

The dorsal stream projects from primary auditory regions to posterior superior temporal and inferior parietal regions, and then to more posterior regions of the frontal lobe compared with ventral stream projections, including pars opercularis of the IFG and premotor and motor cortices (Skipper et al., 2006). At a gross anatomical level, the dorsal stream supporting speech and language functions shares temporal and parietal regions associated with the dorsal stream for vision. These regions show strong fMRI activation during action observation, especially in the left hemisphere (Decety & Grezes, 1999). Some of these shared dorsal stream regions, specifically pars opercularis and inferior parietal cortex, are considered the human homologues of macaque F5 and PF/PFG in which mirror neurons have been identified (Rizzolatti, Fogassi, & Gallese, 2001; Rozzi et al., 2006). Some dual-stream models consider the dorsal stream for speech to be strongly left-dominant (Hickok & Poeppel, 2007).

In conduction aphasia, the ability to repeat is disproportionately impaired. This is classically attributed to damage of the left arcuate fasciculus (Geschwind, 1965), traditionally thought to serve as the primary dorsal pathway (Anderson et al., 1999), although this role is debated due to recent anatomical work implicating the superior longitudinal fasciculus (Schmahmann & Pandya, 2009). Further, voxelwise lesion symptom mapping with perfusion and diffusion-weighted imaging shows damage involving the left supramarginal gyrus and underlying white matter is most strongly related to repetition deficits in aphasia (Fridriksson et al., 2010). Other investigators have implicated damage to temporoparietal regions in conduction aphasia (Buchsbaum et al., 2011), supporting the critical role of the dorsal stream in repetition and the

potential presence of mirror neurons within this functional network. This also suggests that the arcuate fasciculus may not serve the crucial role once suggested for the interconnection of Broca's and Wernicke's areas.

2.6 Aphasia Therapy: Speech Imitation as Therapeutic Tool

Imitation has a long history in therapy for communication disorders, including aphasia. Reducing language formulation demands and using visual input to complement other sensory modalities typically enhances a patients' ability to produce accurate speech output (Duffy, 1995). Early approaches to aphasia therapy were developed prior to the fundamental work of the twentieth century in learning theory and the unfortunate consequences of World War II, which produced many young veterans with head injuries, and often relied solely on repetition and drilling (Basso, 2003). Later researchers continued to use repetition in aphasia rehabilitation, but within a better-defined theoretical framework. For example, in the Helm Elicited Language Program for Syntax Stimulation (Helm-Estabrooks, 1981) designed to treat agrammatism, increasingly more complex syntactic forms are introduced by imitation at level A before the same forms are elicited in context at level B.

Imitation elicits the most accurate picture naming in patients with severe Broca's aphasia (Love & Webb, 1977), and repetition is often the simplest level of a cueing hierarchy (Linebaugh, Shisler, & Lehner 2005). Although imitation is sometimes promoted as a technique to be used only when no other prompts cue correct responses, its ability to facilitate speech output makes it inherently error reducing. Thus, it is a useful tool and desirable starting point in errorless learning designs, in which every response, regardless of accuracy, is viewed as self-reinforcing, and the therapy environment is structured to produce the greatest possible successes

(Sigurðardóttir & Sighvatsson, 2006). However, the benefit of errorless learning remains debated in aphasia rehabilitation (Fillingham, Sage, & Lambon Ralph, 2005).

Many aphasia therapies used in research do not cite imitation as a rationale for their use or theorized effectiveness, yet they still rely heavily on imitation or choral reading (online imitation) in their implementation. Semantic Feature Analysis (SFA) seeks to improve word retrieval by targeting conceptual connections of individually trained words, using modeling and repetition of the target word and its semantic associations when these are not produced independently (Boyle & Coelho, 1995). Melodic Intonation Therapy (MIT), recommended for patients with nonfluent aphasia and poor repetition, uses melody and rhythm to increase speech output, relying on choral productions of intoned targets before progressing to imitation and more naturalistic contexts (Helm-Estabrooks, Morgan, & Nicholas, 1989). Conversational script training introduces scripts to be learned via online and delayed imitation (Youmans, Holland, Muñoz, & Bourgeois, 2005). Choral reading and imitation are also paired with written stimuli in some therapy programs, such as Oral Reading Treatment (Orjada & Beeson, 2005) and Oral Reading for Language in Aphasia (Cherney, 2004).

2.7 Mirror Neuron System and Rehabilitation

Given evidence for motor system activation during action observation (Buccino et al., 2001), and given identification of neural circuits active during both observation and execution of oral movements (Ferrari et al., 2003), there is a sound biological basis for speech imitation as an aphasia rehabilitation technique. Connections between inferior parietal and ventral premotor regions are active during observation and imitation of syllables, as seen in Figure 2.7, and may represent a human mirror neuron network for speech (Mashal et al., 2012). Although the most

straightforward implication of engaging this system may be for the direct motor act of speech production, the role of this network in speech perception (Mottonen & Watkins, 2012) and comprehension of action language (Tettamanti et al., 2005) could result in a broader impact on more general aspects of language rehabilitation (Small et al., 2012).



Figure 2.7 Common Network for Speech Observation and Imitation. Weighted connections obtained from SEM of fMRI during observation (right) and imitation (left) of audiovisual syllables. Connections are shown for the left (top) and right (bottom) hemispheres. Both models share connections between pST, aST, IP, vPM, dPM, and M1S1. Abbreviations are as follows: IP, inferior parietal lobule; M1S1, primary motor/somatosensory cortex; pST, posterior superior temporal gyrus and sulcus; aST, anterior superior temporal gyrus and sulcus; vPM, ventral premotor cortex; dPM, dorsal premotor cortex; M1/S1, primary motor/somatosensory cortex. Figure reproduced from Mashal et al. (2012) courtesy of Frontiers.

A similar approach undertaken in hand motor rehabilitation following stroke comprises viewing videos of daily actions followed by therapist-assisted performance of observed actions with the impaired upper extremity (Ertelt et al., 2007). Patients demonstrate significant improvement following therapy compared to baseline performance or controls, with maintenance of at least 8 weeks after intervention. Increased fMRI activation during object manipulation was found in contralateral supramarginal gyrus and bilateral ventral premotor cortices, consistent with human correlates of the macaque mirror neuron system (Small et al., 2012).

2.8 Aphasia Therapy: Speech Imitation as Therapeutic Theory

2.8.1 IMITATE

The following chapters describe the design and results of a therapy study using IMITATE (Intensive Mouth Imitation and Talking for Aphasia Therapeutic Effect), a novel computer-based aphasia therapy program designed to improve communication skills in aphasia by repetition of audiovisual words and phrases, motivated by neurophysiological findings of mirror properties in human and nonhuman primates (Lee, Fowler, Rodney, Cherney, & Small, 2010). The stimuli are presented by video featuring a view of the speaker's head and shoulders. The therapy is intense and uses ecologically valid stimuli presented by a variety of human talkers, and difficulty increases are graded overall yet are variable within a level. Additional information about the therapy and the study design can be found in Chapter 3.

The control therapy, REPEAT, uses similar principles but audio-only stimuli with a still image of the talker. This therapy also varies the stimulus presentation, such that subjects hear a single presentation by a single talker before each cued repetition, in contrast with the IMITATE group, which hears six consecutive talkers present each stimulus before repeating the target word or phrase several times. Each group hears the same overall number of stimuli and the same number of presentations.

Nineteen subjects completed a 6-week course of therapy (9 h weekly). As there were no significant differences between groups, results for both forms of imitation-based therapy were pooled (Duncan et al., 2016). The lack of significant differences between the two groups may be attributable to the overlap between cortical regions supporting both production and audio-only perception (Skipper et al., 2007), as well as the existence of neurons with auditory-motor as well as visual-motor properties (Kohler et al., 2002). Behavioral and neuroimaging findings are discussed in Chapters 4 through 7.

In a sleep study, high-density EEG recordings were taken for subjects with aphasia on two consecutive nights, before and after participating in a single, highly intensive 3.5-h session of IMITATE (Sarasso et al., 2014). Findings indicate a significant increase in slow wave activity (SWA), associated with synaptic plasticity (Huber, Ghilardi, Massimini, & Tononi, 2004), in regions active during observation–execution of speech in healthy controls (Mashal et al., 2012) in the right (intact) hemisphere. A positive correlation was found between increased SWA over the left ventral premotor cortex and improvement on the Repetition subtest of the WAB. This finding is of interest due to premotor cortex involvement in imitation and the inclusion of this region in the lesion extent of most of the participants.

2.8.2 Speech Entrainment

Citing previously mentioned findings of activation in left frontal speech-motor areas when visual observation accompanies auditory speech, Fridriksson et al. (2009) hypothesized that better performance would be elicited when a computer-based naming treatment for patients with

nonfluent aphasia included audiovisual compared to auditory-only stimuli. Statistically significant gains are made for audiovisual treatment only (see Figure 2.8), including trained and untrained items.



Figure 2.8 Post-therapy Production of Correct, Novel Words. Bar graph showing percent of novel words produced correctly in each of 3 experimental conditions. SE-AV= speech entrainment-audiovisual; SE-AV= speech entrainment-audio only; SS= spontaneous speech. Asterisks indicate p < .01; NS= not significant. Figure reproduced from Fridriksson et al. (2012) with permission from Oxford University Press.

Fridriksson et al. (2012) coined the term "speech entrainment" to describe the ability of some subjects with nonfluent aphasia to produce more fluent speech with an audiovisual model compared to spontaneous speech. Subjects perform online imitation of scripts, which are heard while viewing the speaker on an iPod screen. Only the speaker's mouth is visible to emphasize visual perception of the speech act. This therapy results in production of twice as many words during entrainment for patients with Broca's aphasia. Significant increases in word variety are maintained for at least one week following treatment termination for production of practiced scripts during both entrainment and spontaneous speech. Entrainment of untrained scripts remains significantly improved compared to baseline for at least six weeks.

Using fMRI to explore the neural mechanisms underlying these behavioral findings, Fridriksson et al. (2012) find greater activation in left BA 37 and bilateral anterior insula/BA 47 for the speech entrainment condition compared to spontaneous speech. The authors propose that imitation of speech may facilitate word retrieval (BA 37) and visceral speech support (anterior insula/BA 47) for rapid, online lexical processing and/or airflow modification or for lexical prediction and anticipation of respiratory demands. Broca's area may serve as an internal temporal gating device, which, although injured, can be compensated for by the external temporal gating offered by real-time imitation of an observed speaker, entraining the requisite regions to again function as part of a coordinated network.

2.9 Aphasia Therapy: Nonspeech Motor Observation and Imitation

Speech is a motor activity and gestures are a rich aspect of human communication, whether for independent information transmission or to supplement spoken language (Goldin-Meadow, 1999). Verbal communication among humans may have evolved on top of existing gestural communication systems relying on the observation–execution matching system (Rizzolatti & Arbib, 1998). Gesture has thus been targeted as a means of treatment for aphasia.

2.9.1 Visual Action Therapy

Visual Action Therapy is a nonverbal therapeutic intervention for global aphasia, using real and drawn objects in a hierarchy (Helm-Estabrooks, Fitzpatrick, & Barresi, 1982). The patient imitates manipulation of real objects or pantomimed gestures associated with the use of these objects, with the ultimate goal that the patient is able to produce a pantomimed action as a representation of an unseen object. The rationale is that gestures, requiring only unilateral gross motor control compared to the more precise bilateral motor control required for speech, may be used symbolically (Helm-Estabrooks et al., 1982). Although Visual Action Therapy is not a

³⁷

contemporary subject of research, similar gesture-based therapies continue to be investigated in aphasia rehabilitation.

2.9.2 Action Observation Treatment

Patients with nonfluent aphasia improve in verb retrieval abilities after training with gesture labels paired with either imitation or observation of the target gesture, but not when they observe the gesture while producing a meaningless movement (Marangolo et al., 2010). There is no significant difference between therapy using observation alone and therapy using meaningful imitation, with improvement maintained for 2 months. These findings support the existence of a bilateral distribution of frontoparietal connections that can be engaged by action observation or execution when there is damage to regions associated with the human mirror neuron network in the left hemisphere (Aziz-Zadeh, Koski, Zaidel, Mazziotta, & Iacoboni, 2006). While either action observation or execution engages this network, addition of a meaningless gesture interferes with this process and eliminates the beneficial result of observation and the resultant therapeutic gains in verb retrieval.

However, observation therapy only improves verb production for actions within the human motor repertoire, such as dancing compared to printing (Marangolo, Cipollari, Fiori, Razzano, & Caltagirone, 2012). FMRI findings also demonstrate differences between activation patterns resulting from observation of actions within the realm of human behavior, even when performed by nonhumans (e.g., a dog biting), compared to those that are not (Buccino et al., 2004). These findings further reinforce the role of a mirror neuron network implicated in action observation and execution that subserves language production.

2.10 Conclusion

Although aphasia is a biological disorder resulting from neurological damage, aphasia rehabilitation has traditionally neglected biological approaches to treatment in favor of behavioral and educational models. However, increasing understanding of the neurobiology underlying language is shifting the discourse toward biological mechanisms.

The two main biologically based models for language that have empirical support at this time are the human mirror neuron system and the dual-stream hypothesis. These models support imitation as a powerful tool to rehabilitate the speech and language deficits of aphasia. Action observation engages mirror properties of the same neural networks that are activated during execution, which is as true for speech and oral motor actions as for the grasping behaviors for which they were initially discovered. Higher-level language skills are also grounded in motor systems. Both observation and imitation of speech engage a similar network including components of the dorsal and ventral pathways for language.

Imitation has long been, and continues to be, used in many aphasia interventions. More recently, several researchers have developed neurophysiologically motivated aphasia therapy programs targeting online or delayed imitation as a strategy to improve speech output and language function. Some aphasia therapies have also used nonspeech imitation of actions to enhance gestural communication and production of action labels.

Chapter 3: Participants and Study Design

3.1 Participants

Nineteen native English speakers with aphasia following single, left hemisphere ischemic stroke, confirmed by neurological examination and MRI, were recruited (age range= 31-72; mean= 53.5; SD= 11.7; 4 female (21%)). All had sustained a single stroke 5 to 130 months prior to enrollment (mean= 41.6; SD= 42.9). Select demographic and neurological information can be found in Table 3.1. Lesion overlap can be seen in Figure 3.1.

Subject	Sex	Age	Months	Treatment	# of Sessions	Lesion	Lesion	Aphasia
_		_	Post-Onset	Group	Completed	Size (% LH)	Location	Classification
1	F	72	17	Ι	88	9.64	FIPT	Broca's
2	Μ	60	5	R	84	9.72	FI	Broca's
4	Μ	63	7	Ι	54	7.52	FIPT	Broca's
5	Μ	56	16	R	90	3.31	FPT	Broca's
6	Μ	65	8	Ι	90	6.36	ТР	Conduction
9	F	46	28	Ι	108	17.86	FPT	Broca's
10	F	31	11	Ι	106	10.62	FIPT	Anomic
11	Μ	58	13	R	101	19.78	FIPT	Trans. Motor
12	F	55	22	R	108	0.95	BG	Anomic
13	М	36	78	R	108	12.35	FIPT	Broca's
14	Μ	37	51	R	105	10.06	TIPO	Broca's
15	Μ	70	120	Ι	99	26.34	FIPT	Anomic
16	Μ	58	29	Ι	108	3.25	FI	Anomic
17	Μ	57	130	Ι	107	13.52	FIPT	Anomic
18	Μ	55	81	R	79	11.54	FIPTO	Wernicke's
19	Μ	42	124	R	53	5.21	FI	Anomic
20	Μ	60	7	Ι	103	11.42	TPO	Trans. Sensory
21	М	43	15	Ι	108	12.44	FIT	Broca's
22	М	49	29	Ι	108	11.24	FIPT	Broca's

Table 3.1 Individual Data for Each of the 19 Subjects. The fourth column gives months post stroke onset at time of enrollment. The fifth column gives treatment group (IMITATE or REPEAT). Lesion size is given as the percentage of left hemisphere (LH) voxels included in the lesion mask. Lesion location is abbreviated as follows: F- Frontal, P- Parietal, T-Temporal, I-Insular, O-Occipital, BG- Basal Ganglia. Aphasia classifications are as determined by the Western Aphasia Battery-Revised (WAB) and Transcortical is abbreviated (Trans). Table reproduced from (Duncan et al., 2016) online supplement courtesy of Sage Publications.



Figure 3.1 Lesion Overlap for All Subjects.

3.2 Experimental Summary

The intensive imitation-based therapy (IMITATE or REPEAT; Lee et al., 2010) was administered 6 days per week for three 30 minute sessions each day. Both therapies required participants to listen to words and phrases presented by six different speakers and to repeat them multiple times. Half of the participants also saw a video of the speaker during the presentation. Because there was no statistical difference in any measure between those subjects who saw the speaker and those who did not, all data have been aggregated for the studies described in Chapters 4 through 7.

Over the six-week therapeutic period (Weeks 1-6 of the overall study), participants undertook the specialized speech therapy on a preprogrammed, dedicated laptop. Participants underwent behavioral assessments (WAB, repetition test, narrative production; described in Section 3.4 below) that were administered twice before and twice after therapy, with all evaluations six weeks apart (Weeks -6, 0, 6, and 12). These measures were administered twice pre-therapy to establish a stable baseline, and twice post-therapy to establish immediate changes and maintenance. Figure 3.2 depicts the timing of these assessments relative to therapy.



Figure 3.2 Visual Depiction of Experimental Design. Two pre-therapy assessments (Weeks -6,0) were separated by a 6-week interval during which no therapy was provided. Between Weeks 0 and 6, subjects participated in six weeks of therapy. No therapy was provided in the interval between the two post-therapy sessions (Weeks 6,12).

3.3 Therapy Description

3.3.1 Therapy Features

The important therapeutic features in IMITATE include visual observation, oral repetition, speaker variability, ecologically valid stimuli, high intensity, graded incremental learning and variability in gradation (Lee et al., 2010; Small & Llano, 2009). Visual observation refers to the use of audiovisual stimuli in which the speaker's moving face, lips and mouth are visible, in contrast to the control REPEAT therapy, in which the patients viewed a static image of each speaker while simultaneously hearing their voice utter the target word or phrase. Oral repetition was performed during both IMITATE and REPEAT therapy sessions. Speaker variability was implemented by the presentation of each target word or phrase by each of six different talkers, seen in Figure 3.3. This was used for both therapies and was theorized to aid in generalization to a wider variety of talkers. Ecological stimuli refers to the use of real words and phrases that might be used by an English speaker in the course of daily activities (rather than non-speech oral movements, isolated syllables or nonsense words), as well as to the display of a visible talker such as one might engage with in normal communicative interactions. The use of real words and

phrases was common to both groups, but visual observation (video) was used only for patients receiving IMITATE. Use of ecologically valid stimuli was physiologically motivated by the shared substrates found for observation and execution of action, including oral communicative actions, on a cellular level in studies of macaque cortex (Ferrari et al., 2003; Rizzolatti, Fadiga, Gallese, et al., 1996) and in a motor cortical network model supported by fMRI findings in neurologically intact human subjects (Mashal et al., 2012). Such stimuli may enhance the efficacy of neural connectivity by matching observed actions, such as speech, to internal motor representations existing within the viewer's repertoire of action performance (Kohler et al., 2002).



Figure 3.3 Six Talkers Presenting Therapy Stimuli.

High intensity was implemented by requiring patients from both groups to participate in 90 minutes (3 30-minute sessions) daily, 6 days a week, for 6 weeks. This level of intensity, far greater than the level that can typically be provided by a trained therapist, was chosen based on

the positive correlation between intensity and therapeutic outcome (Bhogal, Teasell, & Speechley, 2003). This massed practice used in both the experimental and control therapies is consistent with other biologically motivated behavioral therapies for aphasia rehabilitation, notably Constraint Induced Aphasia Therapy (CIAT; Pulvermüller et al., 2001). The IMITATE subjects were also exposed to the entire stimulus block of a target before producing a block repetitions, while the REPEAT group had a shorter period for repetition following each of the 6 presentations. The massed stimulation of the IMITATE therapy has been theorized to prime synaptic connections and facilitate the generation or re-instatement of neural pathways (Ferguson, 1999).

Graded incremental learning was addressed by advancing patients in both groups through levels featuring successively more difficult stimuli, such as longer words, more complex phonology, more varied word classes, and longer word sequences (phrases and sentences). Each patient started the therapy at a level that was judged to be appropriate to pre-treatment repetition capability, and advanced through a level each week, unless the clinician performing the weekly repetition test (described in Section 3.4.2) judged that repetition of a level was necessary. Finally, variability in gradation was implemented in both therapies by occasionally providing simpler stimuli at higher levels as well as more difficult stimuli at lower levels according to a probabilistic algorithm. In addition to being consistent with the variability of daily communicative demands, this approach sought to combine two conflicting bodies of evidence on learning strategies that suggest that either a simple (Elman, 1993) or a complex (Kiran & Thompson, 2003) origin (i.e., initial level of difficulty) yields maximum benefit. An example of the therapy interface can be seen in Figure 3.4.



Figure 3.4 Example of Therapy Interface.

3.3.2 Selection of Lexical and Phrasal Stimuli

Therapy items for IMITATE were selected by an algorithm that was designed to consider parameters including number of letters, phonemes, and syllables, part of speech, written frequency, familiarity, frontal and total visibility, and phonemic complexity. These values were derived from the MRC Psycholinguistic Database (Coltheart, 1981), the Kucera and Francis corpus (Kucera & Francis, 1967), the Hoosier mental lexicon (Nusbaum, Pisoni, & Davis, 1984), and various measures of viseme content (Bement, Wallber, DeFilippo, Bochner, & Garrison, 1988; Owens & Blazek, 1985). Measures of phonemic complexity were calculated by coding stimuli for presence of consonant blends in the initial position. Visibility of consonants and vowels was assessed on a 4-point scale that assigned high values to high-visibility productions, like consonants /p/, /b/, /m/, /f/, and /v/. Combining all criteria yielded a final pool of 2568 words, which was augmented by the addition of 68 words that were added due to high functional utility (e.g., "blue", "March", "chair", "Monday"). Words contained between 1 and 4 syllables (mean = 1.42) and between 1 and 12 phonemes (mean = 4.09).

The stimulus set also included 405 phrases that were chosen due to common use and high functional utility for people with aphasia (e.g., "sit down", "watch out", "nice to see you", "please pass the salt"). These were selected from a large variety of English language textbooks, travel guides, and intuition. Phrases were assigned a value on the basis of the number of words and syllables, as well as verb and preposition frequency. Phrases contained two to nine syllables (mean = 4.03) and two to five words (mean = 3.36).

Twelve treatment levels were developed, which increased gradually in complexity. Individual patients were assigned to a treatment level based on their level of functioning. The stimuli analyzed for the study described in Chapter 4 were selected on the basis of the level to which the individual was expected to advance over the course of therapy. This higher level was selected to diminish ceiling effects.

3.4 Behavioral Measures

3.4.1 Western Aphasia Battery

The Western Aphasia Battery-Revised (WAB; Kertesz, 2006) was used as the primary outcome measure, as it was anticipated that benefits of our imitation-based therapy would generalize to other domains of language (Duncan & Small, 2015; Lee et al., 2010; Small & Llano, 2009). The WAB was administered at each of the four main behavioral assessment sessions by a speech-language pathologist (SLP) blind to treatment group. We analyzed the

WAB Aphasia Quotient (WAB-AQ), Cortical Quotient (WAB-CQ), and the four subcomponents of the WAB-AQ (Spontaneous Speech, Auditory Verbal Comprehension, Repetition, and Naming and Word Finding). Paired t-tests were used to compare differences between pre-therapy (Week -6,0) WAB scores and between post-therapy (Week 6,12) WAB scores.

3.4.2 Repetition Test

Tests of repetition accuracy were administered during all four pre- and post-treatment behavioral assessments. These were administered by an SLP using words and phrases randomly selected from the pool of IMITATE therapy stimuli. Repetition tests and other behavioral assessments were performed by different SLPs, blinded to the other's findings.

Repetition test stimuli, as included in Chapter 4, consisted of words and phrases of high difficulty, based on the level to which the subject was expected to advance. Each block of words contained 10 words. Each block of phrases contained 10 phrases with a varying number of words (2-6) depending on level. For both blocks, each word was scored on a 5-point scale (0 signifying no vocalization; 5 indicating accurate, prompt repetition). Scoring was performed once offline by a single SLP for all subjects, and therefore reliability rates are not reported. Performance on these measures was combined in a single repetition score for each time point (mean score for Words and Phrases). Paired t-tests were used to compare differences between pre-therapy (Week -6,0) repetition scores and between post-therapy (Week 6,12) repetition scores.

3.4.3 Narrative Production

At each of the four behavioral assessments, narratives were elicited by having participants tell the story of the fairy tale Cinderella (Saffran, Berndt, & Schwartz, 1989). These narratives

were recorded, transcribed, and subsequently analyzed for number of correct information units (CIUs) produced (Nicholas & Brookshire, 1993). Words were scored as CIUs if they were intelligible, novel during the narrative task, and relevant to the story. Paired t-tests were used to compare number (and percent) of CIUs produced pre- vs. post-therapy, between the two baseline sessions, and between the two post-therapy sessions.

3.5 Neuroimaging

3.5.1 Acquisition

Magnetic resonance imaging (MRI) was acquired at three time points before therapy (Weeks -6, -3, 0) and three time points after its conclusion (Weeks 6, 9, 12). Images were acquired using a 3T Siemens Trio MRI scanner (Siemens Medical Solutions USA Inc., Malvern, PA) at Northwestern University. Anatomical images were acquired with a T1-MPRAGE sequence with TR=2300 ms, TE=3.36 ms, TI=900 ms, flip angle=9°, and 1 mm isotropic voxel size. Resting state fMRI (rsfMRI) images were acquired with an EPI sequence with TR=1500 ms, TE=20 ms, FA=71°, FOV=220x220 mm², matrix size=64x64, 29 axial slices with 4 mm thickness (1 mm gap), and inplane voxel size of 3.75 x 3.75 mm. During 5 minutes of scanning, 200 volumes were acquired. Participants were instructed to get into a comfortable position prior to the rsfMRI scan in order to minimize motion, and to stay awake and keep their eyes open during the scan.

3.5.2 Virtual Brain Transplant and FreeSurfer

One of the difficulties of working with neuroimaging in aphasia is that the presence of large cortical lesions can cause catastrophic failure when attempting to use many neuroimaging tools designed for the analysis of brain imaging data. In order to improve registration and to facilitate

the use of standard software packages, such as FreeSurfer (Fischl, 2012), we performed a Virtual Brain Transplant (VBT; Solodkin et al., 2010) on the T1-weighted anatomical scans of individual participants. In VBT, imaging data from the intact hemisphere is morphed to fit into the lesioned space in the left hemisphere, in which tissue is damaged or absent. The workflow for this technique is depicted in Figure 3.5.

In VBT, a lesion mask is first manually drawn on the anatomical scan. The original brain scan is then extracted from the skull and divided into two separate hemispheres. This allows the nonlesioned hemisphere to be flipped into the mirrored space of the lesioned hemisphere, followed by nonlinear warping using symmetric diffeomorphic image registration via ANTS (Avants, Epstein, Grossman, & Gee, 2008) with cost function masking of the lesioned region. Once the nonlesioned hemisphere is aligned to the shape of the lesioned hemisphere, the lesion mask is used to extract an image of intact tissue from the flipped nonlesioned hemisphere. This image is enlarged by a few millimeters in order to better capture the expanded sulci of the lesioned hemisphere. It is then used to replace the region overlaid by the lesion mask in the left hemisphere of the whole brain image, and this transplanted region is morphed with the surrounding area to blur the border between images.

Performing VBT permitted the use of FreeSurfer (Fischl, 2012) for the segmentation of brain scans. With FreeSurfer, we were able to reconstruct individual cortical surfaces for each of our participants. We used individual volumes, surfaces, and curvatures to construct a common template (see Figure 3.6) for use in group analysis, in order to facilitate more accurate registration compared to a standard atlas comprising neurologically intact brains.



Figure 3.5 Virtual Brain Transplant (VBT) Workflow. (A) shows an axial view of the original brain with large left hemisphere lesion. This brain is split into nonlesioned right (B) and lesioned left (C) hemispheres. In (D), the nonlesioned right hemisphere (B) has been flipped around the x axis and aligned to the lesioned left hemisphere (C) using a nonlinear warp. Note the size of the horns of the lateral ventricle in (D), compared to (B) and (C). (E) shows a mask that has been hand drawn on the original image (A) to demarcate the limits of the lesion. In (F), the lesion mask from (E) has been used to extract an image from the flipped right hemisphere (D) for use in the virtual transplant, and the original lesion appears filled by intact tissue.



Figure 3.6 Group Template. Created from the brains of all participants using FreeSurfer (Fischl, 2012). Montage shows axial slices from inferior (top left) to superior (top right).

3.5.3 Brain Parcellation

Following VBT and FreeSurfer, each participant's brain was parcellated into 463 regions using the Connectome Mapping Toolkit (Hagmann et al., 2008), including 448 right hemisphere cortical regions, 14 subcortical nuclei, and the brainstem. Parcellated surfaces can be seen in Figures 3.7 to 3.10.



Figure 3.7 Lateral View of the Parcellated Left Hemisphere. Left image shows pial surface, right image shows an inflated surface permitting viewing of surfaces within the sulci.



Figure 3.8 Medial View of the Parcellated Left Hemisphere. Left image shows pial surface, right image shows an inflated surface permitting viewing of surfaces within the sulci.



Figure 3.9 Lateral View of the Parcellated Right Hemisphere. Left image shows pial surface, right image shows an inflated surface permitting viewing of surfaces within the sulci.



Figure 3.10 Medial View of the Parcellated Right Hemisphere. Left image shows pial surface, right image shows an inflated surface permitting viewing of surfaces within the sulci.

3.5.4 Preprocessing

Further details about the preprocessing and analysis of structural and functional MR images can be found in the Materials and Methods sections of Chapters 6 and 7.
Chapter 4: Performance Variability as a Predictor of Response to Aphasia Treatment

4.1 Introduction

4.1.1 Background

Therapeutic research in aphasia typically characterizes baseline and improved language skills in terms of mean scores on a specific task or assessment battery. Whereas this approach succeeds at capturing variability across individuals, it fails to capture such variability within individuals. Performance fluctuations within a single individual (intra-individual variability) are typically perceived as an inconvenient impediment to reaching the desired general conclusions about a new therapy. But treating intra-individual variability as a nuisance parameter or measurement error (e.g., of the same magnitude and significance as inter-individual variability; Van Geert & Van Dijk, 2002), may be giving up important and highly relevant information about the therapy. In fact, short-term performance inconsistency on a particular task may represent a characteristic feature of – and a metric to gauge – an individual's functional status. Particularly in the context of large variability, mean performance may oversimplify the true nature of behavior and inadequately capture the range of ability (Nesselroade, 2002), obscuring insight into potential therapeutic benefit and outcome assessment on an individual basis.

4.1.2 Intra-individual Variability in Cognitive and Motor Function

Existing limited research on intra-individual variability in cognitive and perceptual-motor function in healthy aging (Garrett, Macdonald, & Craik, 2012; Li, Lindenberger, & Sikström, 2001; Lövdén, Li, Shing, & Lindenberger, 2007; Nesselroade & Salthouse, 2004) and dementia (Duchek et al., 2009; Gamaldo, An, Allaire, Kitner–Triolo, & Zonderman, 2012; MacDonald, Hultsch, & Dixon, 2003) suggests a relation between increased intra-individual variability and decreased performance. Yet other, seemingly contradictory, findings suggest that greater intraindividual variability has positive implications for acquiring skills with practice or training. For example, increased intra-individual variability in a cognitive or motor skill during learning precedes (and presages) mastery of that skill during development (Courage, Edison, & Howe, 2004), and in cognitive training of healthy older adults, the pre-treatment degree of intraindividual variability predicts higher response accuracy and performance improvement (Allaire & Marsiske, 2005).

These data suggest that performance variability may suggest susceptibility to change and/or the potential for learning. Fluctuations representing adaptive variability (Li et al., 2004) may be conceived of not as vulnerability, but as potential. Further, distinguishing between adaptive and maladaptive variability may be key to understanding the significance of these measures in predicting future outcomes.

4.1.3 Intra-individual Variability in Aphasia

In the realm of stroke recovery, extensive investigation has addressed differences between individuals, yet little work studying performance variability within individuals (although important work has explored the role of attention in intra-individual variability in aphasia (e.g., Erickson, Goldinger, & LaPointe, 1996; Tseng, McNeil, & Milenkovic, 1993)). In the recovery of language functions after stroke, intra-individual performance variability has not been investigated, either as a correlate of present functioning or as a predictor of post-treatment ability (but see Small, Holland, Hart Jr, Forbes, & Gordon (1995) for a theoretical study). The implications are far-reaching. From a research standpoint, our knowledge of language recovery

in aphasia is limited to mean scores and effect sizes using pooled standard deviations, thus neglecting individual parameters of variability. Such data may represent fundamentally incomplete metrics, substituting a crude numerical proxy for the more nuanced complexity of performance, and thus profoundly affecting our understanding of recovery. From a clinical standpoint, such omission could have graver consequences, since the most desirable measure of rehabilitation success is a patient's consistent performance in the real world, not maximal performance in the clinic or the possibility of good performance under ideal conditions.

4.1.4 Motivation for the Present Study

We hypothesize that intra-individual variability on a language task is predictive of the ability of an individual with aphasia to improve mean performance on that task through training. We investigate this hypothesis in a clinical trial of an intensive, imitation-based aphasia therapy motivated by neurophysiological evidence (Lee et al., 2010; Small, 2009) that uses a computer interface to prompt repetition of words and phrases to engage a frontal-parietal motor cortical network involved in both observation and execution of speech (Hari et al., 1998). In this paper, we report on an experiment testing the hypothesis that pre-treatment intra-individual variability predicts therapeutic outcome.

4.2 Materials & Methods

4.2.1 Participants

Participants are nineteen subjects as described in Section 3.1. Select demographic and neurological information is listed in Table 3.1, and lesion overlap is depicted in Figure 3.1.

4.2.2 Experimental Summary

The details of the study are described in Section 3.2 and visually depicted in Figure 3.2. Features of the IMITATE therapy and stimuli selection are described in Section 3.3.

4.2.3 Behavioral Measures

4.2.3.1 Assessment Tools

Two measures are described in this study, the WAB and the repetition test (see Section 3.4).

4.2.3.1.1 Western Aphasia Battery-Revised

Additional information about WAB administration can be found in Section 3.4.1. Six WAB measures were used: WAB Aphasia Quotient (WAB-AQ), Cortical Quotient (WAB-CQ), and the four subcomponents of the WAB-AQ (Spontaneous Speech, Auditory Verbal Comprehension, Repetition, and Naming and Word Finding). There was no significant difference in any of these measures between the two pre-treatment sessions or between the two post-treatment sessions (p > 0.05 on two-tailed paired t-tests).

4.2.3.1.2 Repetition Test

Additional information about the repetition test can be found in Section 3.4.2. One subject (2) was excluded from this analysis due to missing data, leaving 18 subjects. There were no significant differences between Week -6 and Week 0 repetition scores, or between Week 6 and Week 12 repetition scores (p > 0.05 on two-tailed unpaired t-test).

4.2.3.2 Changes in Behavioral Performance

Seven measures of language performance were studied: WAB-AQ, WAB-CQ, the four

subcomponents of the WAB-AQ, and the score from the repetition test (defined as the average percent correct for blocks of words and phrases). For each measure, the pre-treatment score was taken to be the mean of Week -6 and Week 0 scores, and the post-treatment value is the Week 6 score (post-therapy repetition). We did not use scores from Week 12 (fourth behavioral assessment) as two subjects missed this assessment. Therefore, our definition of improvement, for all measures, is the Week 6 score minus the mean for Weeks -6 and 0.

Pre-treatment scores were compared with post-treatment scores using two-tailed paired ttests. All significance tests use $\alpha = 0.05$. Due to the nested nature of the WAB measures (i.e., four subcomponents of the WAB-AQ are used, which also contribute to WAB-CQ), Bonferroni correction with n = 5 was applied for the repetition assessment and the four subcomponents of the WAB-AQ (Spontaneous Speech, Auditory Verbal Comprehension, Repetition, and Naming and Word Finding).

4.2.3.3 Intra-individual Variability as Predictor of Improvement

The repetition test based on stimuli from the pool of IMITATE items (see Sections 3.3 and 3.4 for further detail) was used to test directly the hypothesis that intra-individual variability in a language task is predictive of the ability of an individual with aphasia to improve performance on that task through training. This single measure was selected for two reasons: (1) there were two days on which the repetition test was administered at least twice (Week 0 and Week 6), allowing a robust assessment of individual variability before and after treatment, and (2) these stimuli were developed to be grossly equivalent in complexity, in contrast with the hierarchical ordering of increasing complexity on the WAB subtests. We chose not to pool data from Weeks -6 and 0 when computing intra-individual variability to avoid confounding variability on

different time scales; our intra-individual variability scores measure performance variability within a given day only. We computed a repetition intra-individual variability measure pooling variances of words and phrases blocks. Details about the calculation of this measure can be found in Section 4.2.3.3.1.

Our specific question was the extent to which pre-treatment repetition intra-individual variability predicted improvement in repetition mean, which we determined by computing a Pearson correlation coefficient. We used Week 0 and Week 6 repetition mean scores as the preand post-treatment values, ignoring Week -6 scores for consistency with intra-individual variability calculations. There was no significant difference between pre-treatment repetition mean calculations regardless of whether Week -6 repetition test scores were included (two-tailed paired t-test, p > 0.05).

We used stepwise linear regression to identify those pre-treatment variables that best predicted improvement. In addition to pre-treatment repetition intra-individual variability, these variables included participant age, months post stroke onset (MPO), number of sessions completed (NSC), aphasia type (fluent vs. nonfluent), and pre-treatment repetition mean. NSC was tracked by automated video recording of patient participation during each session via the built-in laptop camera, and then verified by review of these recordings. Stepwise regression was performed with the MATLAB stepwise function, using the default settings: a new predictor is selected if its regression coefficient would be significantly nonzero at the 0.05 level, and an existing predictor is removed if its coefficient is not significantly nonzero at the 0.10 level.

4.2.3.3.1 Intra-individual Variability Measure

For each time point (Week 0 or Week 6) and each subject, we computed the variance of the

per-word scores in the pooled Word blocks, and the same for the pooled Phrase blocks. Our intra-individual variability measure for repetition score is the square root of the mean of the variances in the two blocks (words and phrases), which is comparable to a pooled standard deviation but gives equal weight to the words and phrases blocks. This is analogous to our definition of the repetition mean (see Section 4.2.3.2), which gives equal weight to words and phrases, despite the phrases block containing a larger, and more variable, number of words than the words block.

4.2.3.3.2 Exclusion of Subjects from Intra-individual Variability Analysis

Beginning with all 18 subjects for which intra-individual variability data was available (i.e., all except Subject 2), we calculated full and partial Pearson correlation coefficients for the relationship between pre-treatment intra-individual variability and improvement. The full correlation coefficient was r= 0.79. The partial correlation coefficient, controlling for age, months post onset and number of sessions completed as potential confounding variables, was r= 0.78. The partial correlation coefficient, with these confounding variables plus an additional one of pre-treatment repetition mean performance, was r= 0.76.

We then excluded Subject 4 as an outlier, since its pre-treatment mean performance lies more than 3 standard deviations below the group mean (see Figure 4.1B). Excluding this subject only, we recalculated the correlation coefficients between intra-individual variability and improvement in post-therapy repetition mean, with results shown in Table 4.1. Note that the full correlation coefficient of r= 0.75 (excluding subject 4) is very similar to the value r= 0.79 obtained using all subjects. The same is true of the partial correlation coefficient, with age, months post onset, and number of sessions completed as confounding variables: r= 0.76 (excluding subject 4) versus r=

0.78 using all subjects. However, with the additional confounding variable of pre-treatment mean, the partial correlation coefficient is now r= 0.48 (excluding subject 4) versus r= 0.76 using all subjects, and is not significant at the p= 0.05 level. This means that, when subject 4 is excluded, pre-treatment mean and the other confounding variables listed predict improvement well enough that including pre-treatment intra-individual variability as an additional predictor does not significantly improve the prediction.

Repetition measures:	r	р
Intra-individual variability vs. improvement	0.75	0.0005
(controlling for age, MPO, aphasia type, NSC)	0.73	0.0050
(controlling for age, MPO, NSC, aphasia type, and mean)	0.48	0.1153
Mean vs. improvement	-0.64	0.0055

Table 4.1 Full and Partial Correlation Coefficients for Pre-treatment Intra-Individual Variability and Improvement in Mean Accuracy for the Repetition Test. Partial correlations control for age, months post onset (MPO), aphasia type (fluent vs. nonfluent), and number of sessions completed (NSC) and pre-treatment mean. The full correlation between pre-treatment mean and improvement is also included. Subject 4 has been excluded as an outlier from all calculations. Table reproduced from Duncan et al. (2016) online supplement courtesy of Sage Publications.

Noting the effect of controlling for pre-treatment mean in the context of the overall high performance scores (group mean 79.4%, SD 18.8%), we considered that ceiling effects might have artificially constrained both intra-individual variability and improvement scores. The asymptotic appearance of Figure 4.1B, showing pre-treatment repetition mean vs. intra-individual variability, further reinforces this possibility. This scatter plot shows several near-ceiling means, and also a strong negative association (excluding the outlier, subject 4) between mean and intra-individual variability. Therefore, our intra-individual variability measure may be artificially low for subjects with high mean performance, due purely to a ceiling effect,

especially since our intra-individual variability measure is based on variance, the estimation of which gives high weight to extreme values. In order to compensate partially for this, we recalculated the above correlation coefficients using a restricted set of subjects. We calculated the overall mean of the per-subject standard errors of the repetition mean, which was 2.2% pre-treatment and 1.9% post-treatment, and excluded those subjects with a repetition mean, either pre- or post- treatment, within one mean standard error of the ceiling (100%). (Note that the standard error of the repetition mean score (SEM) equals one half of the square root of the sum of squared SEMs for words and phrases separately.) The subjects excluded are listed in the main text, and marked with crosses in Figures 4.1 and 4.2. Table 4.2 contains results analogous to Table 4.1, but with the listed subjects excluded.

4.3 Results

4.3.1 Changes in Behavioral Performance

Statistically significant improvements were demonstrated in five of the seven language measures assessed, with correction for multiple comparisons. Results of two tailed t-tests are summarized in Table 4.2 with uncorrected p values. We used Bonferroni correction (n = 5; see Section 4.2.3.2) for the six WAB measures vs. repetition test to determine significance. Significant improvement was measured for the repetition test, WAB-AQ, WAB-CQ, and two of the four WAB-AQ subcomponents (Repetition, Naming and Word Finding). The two remaining subcomponents of the WAB-AQ (Spontaneous Speech, Auditory Verbal Comprehension) did not demonstrate significant change.

MEASURE	Mean Pre	Mean Post	Mean Improvement	p value
	(SD)	(SD)	(SD)	(uncorrected)
WAB-AQ	67.72	70.34	2.61	.0068*
	(20.00)	(18.33)	(3.73)	
WAB-CQ	71.27	73.89	2.62	.0005*
	(16.50)	(15.44)	(2.72)	
WAB-SS	12.42	12.81	0.47	.1460
	(4.49)	(3.89)	1.36	
WAB-AVC	164.40	166.42	2.02	.5186
	(26.94)	(25.78)	(7.67)	
WAB-Rep	67.16	72.00	4.84	.0044*
	(26.64)	(24.85)	(6.49)	
WAB-NWF	64.62	67.00	2.39	.0037*
	(29.21)	(29.19)	(3.13)	
Repetition Test	79.40	86.19	6.58	.0002*
	(19.03)	(19.11)	(5.90)	

Table 4.2 Performance Measures for All Subjects. The asterisk (*) marks significant values after Bonferroni correction for 5 comparisons (* p < 0.05/5). SD: standard deviation; WAB: Western Aphasia Battery-Revised; AQ: Aphasia Quotient; CQ: Cortical Quotient; SS: Spontaneous Speech; AVC: Auditory Verbal Comprehension; Rep: Repetition; NWF: Naming and Word Finding. Table reproduced from Duncan et al. (2016) courtesy of Sage Publications.

4.3.2 Intra-individual Variability as Predictor of Improvement

In this section, only the repetition test results are used. In contrast to Table 4.2, pre-treatment results are from Week 0 only, for reasons explained in Section 4.2.3.3. Pre-treatment repetition mean ranged from 20.5% to 99.5% (overall mean 79.4%, SD 18.8%). Improvement in repetition mean from pre-treatment to post-treatment (Week 0 to Week 6) ranged from -3.8% to 16.5% (median 5.3, mean 6.7, SD 5.7), representing a mean improvement of 0.34 points on the 5-point scale used to rate the repetition performance.

Figure 4.1A shows pre-treatment intra-individual variability versus improvement in repetition mean performance, and Figure 4.1B shows pre-treatment repetition mean vs. pre-treatment intra-individual variability.



Figure 4.1 Pre-treatment Intra-Individual Variability, Mean, and Improvement on the Repetition Test. Data shown for repetition test: A, improvement plotted against intra-individual variability; B, intra-individual variability plotted against pre-treatment mean. Participant 2 was excluded because of missing data, leaving 18 participants. Those marked with black crosses are excluded from further analysis (see Section 4.2.3.3.2). Figure reproduced from Duncan et al. (2016) courtesy of Sage Publications.

We removed several subjects from further analysis due to outlier status (4) and possible ceiling effects (i.e., subjects near threshold pre- or post-therapy: 10,11,12,16), as detailed in Section 4.2.3.3.2. For the remaining subjects, there is a positive correlation (r= 0.68, p= 0.01 uncorrected) between pre-treatment intra-individual variability and improvement – i.e., higher pre-treatment intra-individual variability is associated with greater improvement. We then considered all of the pre-treatment variables listed earlier (age, MPO, NSC, aphasia type, pre-treatment intra-individual variability and pre-treatment mean) as possible predictors of improvement in post-therapy repetition accuracy. In a stepwise regression, the optimal regression model found intra-individual variability to be the only predictor of improvement (p = .01, as noted above). With all subjects included, the relationship remains highly significant (p = .0001),

with no additional predictors selected. Repeating this stepwise regression without variability included in the model resulted in the selection of no variables, whether for the entire group or with near-threshold subjects excluded.



Figure 4.2 Changes in Intra-Individual Variability and Mean on the Repetition Test. Participants shown are as in Figure 4.1. Figure reproduced from Duncan et al. (2016) courtesy of Sage Publications.

Finally, we examined intra-individual variability in repetition accuracy immediately posttreatment (Week 6). This decreased significantly over the course of treatment (two-tailed paired t-test, p < 0.05), regardless of whether we consider all subjects or exclude near-threshold subjects (as detailed in Section 4.2.3.3.2). Post-treatment intra-individual variability in repetition accuracy is positively correlated with change in repetition mean when we consider all 18 subjects (Pearson's r = 0.49, p = 0.04 in a 2-tailed t-test). However, it is no longer significant when we exclude subjects who were near-threshold either before or after therapy (r = 0.25, p = 0.41). The change in intra-individual variability in repetition accuracy over the course of treatment has a significant negative correlation with improvement, whether considering all subjects (r = -0.48, p =0.053) or excluding near-threshold subjects (r =-0.57, p =0.04), as shown in Figure 4.2. This effect remains if we control for all of the confounding variables (age, MPO, NSC, aphasia type, pre-treatment mean, pre-treatment intra-individual variability; r = -0.69, p = 0.018). Put another way, reduction in intra-individual variability is positively correlated with improvement.

4.4 Discussion

The present study reviews outcomes of a clinical trial of imitation-based therapy for chronic aphasia, and explores a new hypothesis about the role of intra-individual variability in predicting benefit from language therapy following stroke.

This study showed positive effects of the IMITATE system of repetition based, computer assisted speech-language therapy for patients with chronic aphasia. In particular, participants undergoing the therapy had statistically significant gains on composite language and cognitive measures on a standard test for aphasia (WAB), as well as on two repetition accuracy measures. Significant gains were made over a relatively short treatment period (6 weeks) in subjects who, in some cases, were more than a decade removed from their stroke. Future investigation will refine the IMITATE therapeutic protocol in view of the results from the current study, related research (Fridriksson et al., 2012), and other theoretical considerations.

Our analysis suggests that subjects demonstrating higher levels of performance variability prior to therapy are likely to experience greater improvement over the course of treatment. Specifically, subjects demonstrating greater intra-individual variability during repetition before therapy demonstrated greater improvement in repetition than those with lower intra-individual variability. Perhaps most interestingly, intra-individual variability declined over the course of treatment, and there was a significant correlation between performance improvement and intra-individual variability reduction.

The finding that intra-individual variability is a positive predictor of language improvement appears to conflict with existing literature on the relation between intra-individual variability and task performance in cognitive and perceptual-motor domains. In healthy aging and dementia, intra-individual variability has generally been negatively correlated with both short-term performance (Hultsch, MacDonald, Hunter, Levy-Bencheton, & Strauss, 2000) and long-term variables (Lövdén et al., 2007), including time until death (MacDonald, Hultsch, & Dixon, 2008). On the other hand – and perhaps most relevant here – evidence also suggests that increased variability in a particular cognitive or motor domain may be associated with greater potential for change following training specific to that domain (Li et al., 2004).

Correlation of task-specific variability with performance improvement has been attributed to influences of learning and strategy use in development (Siegler, 2007). During skill acquisition, changes occur in execution of strategies, even in the absence of changes in strategy selection (Siegler & Lemaire, 1997). These subtle changes may result in adaptive variability while learning specific tasks. Thus as an individual achieves maximal potential on a task, variability decreases. In expert motor control, when an individual is performing a highly practiced skill at or near peak level, performance variability is reduced, and this consistency is reflected in precise activation of neural networks during motor planning (Milton, Solodkin, Hlustik, & Small, 2007). Our finding that intra-individual variability decreased over the course of therapy provides further support for this proposition, especially given the significant correlation between improvement on the repetition test and intra-individual variability reduction. Within the limitations of their language impairment, our participants became more expert at the practiced task, thus demonstrating more consistent and more accurate performance. Although it is impossible to determine from the present study, it would be of great interest to explore whether such variability

might continue to play a predictive role in the outcome of further therapy or with the introduction of new or more difficult tasks.

That increased variability has been found at dynamic periods of cognitive decline and development suggests, not surprisingly, that these transitions do not occur uniformly. It seems probable that such variability indicates a lack of system stability that is influenced by opposing tendencies. On one hand, in a progression towards overall decline, increased variability results as the valleys of performance drop more deeply; on the other, in the case of development, or recovery, the heightening of peaks is responsible for the observed fluctuation. In support of this, increasing latency for an individual's slowest reaction times is related to increasing variability for older adults (Williams, Hultsch, Strauss, Hunter, & Tannock, 2005). Nevertheless, cognitive enhancement can occur with training and stimulation programs, with functional gains reported in daily life despite increasing age (Willis et al., 2006). As in development and recovery, when older subjects realize increased potential, greater intra-individual variability is correlated with improved learning (Li et al., 2004).

In the present study, it is possible that individuals demonstrating less variability are at or near an asymptote of their abilities, given their neurological status, the extent of lesion damage, and the degree to which they have already experienced recovery. While it is generally accepted that individuals with aphasia encounter a plateau within the first year following stroke (Pedersen, Jorgensen, Nakayama, Raaschou, & Olsen, 1995), intra-individual variability may serve as a more sensitive, individualized measure of potential than time post stroke, as well as an immediate and cost efficient means of prediction.

The implications for language rehabilitation are of great significance, as predictors of response to aphasia treatment are presently limited. Specifically, it may be productive for

clinicians to target skills in which patients demonstrate high performance variability prior to treatment, rather than areas in which limited variability suggests a reduced capacity for gains with therapy. It may also be productive to periodically re-assess patient performance on a variety of tasks, in order to determine whether cycling through treatment goals, selected on the basis of variability as a proxy for potential, may be beneficial. However, such possibilities should be interpreted with caution, as the present study represents a new avenue of inquiry, and little is yet known about how intra-individual variability changes over the course of recovery. Although the present analysis considered time post onset, subjects participating in this study were all at chronic aphasia stages. Therefore, there is no suggestion that findings would be identical or even similar in acute phases of recovery. Additionally, several measures that may impact variability in task performance were not included in our assessment, such as attention, mood, and fatigue. Future studies would benefit from operationalization and inclusion of these variables.

Further limitations of our study include the potential for practice effects, given the relatively short time over which these tests were administered. However, we believe that the lack of significant differences between the two pre-therapy time points and the two post-therapy time points suggests that this is not a major confound for the present study. While our inclusion of fluent vs. nonfluent aphasia classifications did not indicate significant differences in benefit between these groups, there was not adequate power in our sample to address the differential effects of repetition therapy that may exist for different aphasia types. It is also worth noting that our imitation-based therapy was heavily dependent on motor processes, as was our repetition outcome measure. Therefore, it is not possible to definitively state that intra-individual variability would predict improvement on purely cognitively based tasks.

While extrapolation from the present study to clinical guidelines would be premature, if the relationship between behavioral intra-individual variability and post-treatment performance withstands further exploration, it may suggest that those demonstrating higher levels of baseline variability are good candidates for intervention. Intra-individual variability, in this conception, could represent a measure of plastic potential, the extent to which an individual's present neurological status is conducive to the kind of recovery or reorganization necessary to manifest improvement with practice and stimulation. However, individuals performing consistently at the same level may require different types of intervention if they are to realize enhanced function, and these patients may be better candidates for referral to clinical trials, pharmacology or more invasive forms of treatment.

Chapter 5: Effects of Imitation-Based Aphasia Therapy on Narrative Content

5.1 Introduction

Generalization has long been recognized as one of the most significant challenges to the treatment of disorders of speech and language (e.g., Ingham, 1980; Palyo, Cooke, Schuler, & Apolloni, 1979; Wambaugh, Kalinyak-Fliszar, West, & Doyle, 1998). Typically, a patient trained on one task, such as orally producing the word "hammer" when shown a hammer, demonstrates improvement for that same item and task with practice. Generalization refers to the ability for an individual trained in one behavior to demonstrate improvement on a different, untrained behavior, such as being able to verbally label a screwdriver or to write the word "hammer".

With aphasia, as with other language deficits, generalization is both a primary concern and an often unrealized goal. This is true across various measures, including naming (Nickels, 2002), discourse comprehension (Kiran, Des Roches, Villard, & Tripodis, 2015), and the production of scripts (Cherney, Kaye, Lee, & van Vuuren, 2015). For the roughly one million Americans diagnosed with aphasia (NIDCD, 2010), there is no cure, although speech therapy provides some benefit (Brady et al., 2012). Despite this prevalence, similar to that of Parkinson's disease, aphasia is little known outside the spheres of neurologically trained medical professionals, aphasia patients, and their families.

Unlike Parkinson's disease, aphasia is not an organic disease per se. Rather, the deficits that we observe in aphasia are the result of a biological process such as stroke, traumatic brain injury, or brain tumor that results in injury to neural circuits supporting language behaviors. Thus, aphasia may be more accurately conceived of as symptomatic, rather than causal. As aphasia results from biological damage, the greatest benefits in its treatment may be achieved through

interventions that are biologically informed (Small, 2000).

Imitation has historically been incorporated into many different treatment protocols and used with many types of aphasia (Duncan & Small, 2015). More recent discoveries in neurophysiology have given rise to a biological motivation for the use of imitation in language rehabilitation, as mirror neurons in macaque (di Pellegrino et al., 1992) and analogous mirror properties in humans (Buccino et al., 2001) have been found to underlie both observation and execution of mouth actions, including speech (Skipper et al., 2007). This suggests that we may be able to harness the stimulatory effects elicited by observing speech to support a consequent increase in the production of speech.

IMITATE therapy was motivated by this concept, that engagement of the shared anatomical network underlying the observation and execution of speech (Mashal et al., 2012) would enhance the ability of individuals with aphasia to produce verbal output (J. Lee et al., 2010). In the therapy, individuals with aphasia repeat words and phrases presented by human speakers on a laptop. We have previously shown (Chapter 4; Duncan et al., 2016) therapeutic benefit on a standard test of aphasia, with particular effects on repetition, but additional effects on naming and word finding tasks from the Western Aphasia Battery (Kertesz, 2006). These findings suggest that improvement following imitation-based treatment extends beyond trained items and is not restricted to the practiced task.

The current investigation hypothesizes that the imitation-based therapy will generalize to narrative production. In particular, we examine the changes of our patients on the "Cinderella" task (Saffran et al., 1989), i.e., their ability to produce this well-known fairytale. We also seek to investigate the relationship of select demographic, behavioral, and neurological variables with this benefit.

5.2 Materials & Methods

5.2.1 Participants

Participants are nineteen subjects as described in Section 3.1. Select demographic and neurological information is listed in Table 3.1, and lesion overlap is depicted in Figure 3.1.

5.2.2 Neuroimaging

Structural magnetic resonance imaging (MRI) was acquired prior to therapy to assess lesion size and location. Details about acquisition of T1-weighted anatomical scans can be found in Section 3.5.1. Lesions were drawn by hand under the supervision of a trained neuroanatomist in order to assess lesion location and calculate lesion volume.

5.2.3 Behavioral Measures

Behavioral evaluations were performed at two separate time points before therapy (baseline assessment) and two time points after therapy. Baseline evaluations occurred 6 weeks (Week -6) and immediately (Week 0) before therapy. Post-therapy evaluations occurred immediately following the 6-week therapy interval (Week 6) and 6 weeks following the conclusion of treatment (Week 12). At each time point, narratives were elicited by having participants tell the story of the fairytale Cinderella (Saffran et al., 1989) and analyzed for correct information units (CIUs) as described in Section 3.4.3. One subject (20) was excluded due to missing data, leaving 18 subjects.

One-tailed paired t-tests ($\alpha = 0.05$) were used to compare number and percent of CIUs produced pre- vs. post-therapy due to our strong *a priori* hypothesis that the therapy would increase productive output. Two-tailed paired t-tests were used to compare number and percent

of CIUs produced between the two baseline sessions, and between the two post-therapy sessions. One subject (10) missed the Week 12 assessment and was therefore excluded from the t-test comparing the two post-therapy sessions, leaving 17 subjects.

Pre-therapy and mean scores for both number and percent of CIUs as well as post-therapy (Week 6) scores for both measures are shown in Table 5.1. This table also includes select demographic variables (age and sex; see Table 5.2 for additional variables).

Subject	Sex	Age	Pre-therapy	Week 6 CIUs	Pre-therapy	Week 6 CIUs
		-	CIUs (#)	(#)	CIUs (%)	(%)
1	F	72	0.0	0	0	0
2	М	60	10.0	0	15.0	0
4	М	63	0.0	0	0	0
5	М	56	30.5	119	30.0	38.3
6	М	65	146.5	244	38.6	43.8
9	F	46	22.5	33	40.7	48.5
10	F	31	92.0	178	43.0	42.5
11	М	58	2.5	10	5.2	20.4
12	F	55	145.0	160	70.4	70.8
13	М	36	41.5	45	49.9	57.7
14	М	37	30.0	32	18.0	30.0
15	М	70	91.5	200	61.4	65.2
16	М	58	146.5	270	56.1	70.0
17	М	57	248.0	190	54.0	67.4
18	М	55	16.5	3	9.0	1.1
19	М	42	254.0	383	63.2	60.0
20	М	60	2.0		3.2	
21	М	43	7.0	14	7.3	10.3
22	М	49	4.5	26	7.9	15.7

Table 5.1 Select Demographic Information and Pre-/Post-therapy Performance on Cinderella Narrative Task. CIUs correct information units; # = number; % = percent.

Two separate stepwise regression analyses were used to identify variables associated with changes in number of CIUs and percent of CIUs following treatment (Week 6 compared to baseline). Regression was performed using the MATLAB stepwise function with the default settings using a criterion of $\alpha = 0.05$ to select new predictors and $\alpha = 0.10$ to exclude existing predictors. Variables included as potential predictors in the regression model included age,

months post stroke onset (MPO), number of therapy sessions completed (NSC), fluency (fluent vs. nonfluent), lesion size, and baseline performance (i.e., mean number or percent of CIUs produced pre-therapy). Values for these variables are shown in Tables 5.1 and 5.2.

Subject	MPO	NSC	Aphasia	Fluency	Lesion Size (cm ³)	Lesion
	1.5	0.0	Classification	27	(1.00	Location
I	17	88	Broca's	N	61.88	FIPT
2	5	84	Broca's	Ν	50.46	FI
4	7	54	Broca's	Ν	45.38	FIPT
5	16	90	Broca's	Ν	49.21	FPT
6	8	90	Conduction	F	42.12	ТР
9	28	108	Broca's	Ν	130.58	FPT
10	11	106	Anomic	F	158.29	FIPT
11	13	101	Trans. Motor	Ν	137.13	FIPT
12	22	108	Anomic	F	14.11	BG
13	78	108	Broca's	Ν	86.93	FIPT
14	51	105	Broca's	N	90.69	TIPO
15	120	99	Anomic	F	219.06	FIPT
16	29	108	Anomic	F	48.66	FI
17	130	107	Anomic	F	82.35	FIPT
18	81	79	Wernicke's	F	162.10	FIPTO
19	124	53	Anomic	F	36.58	FI
20	7	103	Trans. Sensory	F	100.62	TPO
21	15	108	Broca's	N	81.98	FIT
22	29	108	Broca's	N	100.62	FIPT

Table 5.2 Potential Predictors of Maintenance Used in the Regression Model. These include age, months post stroke onset (MPO), number of therapy sessions completed (NSC), aphasia classification and fluency (F=fluent; N=nonfluent), lesion size (in cubic centimeters), and lesion location. Lesion location is abbreviated as follows: F- Frontal, P- Parietal, T-Temporal, I-Insular, O-Occipital, BG- Basal Ganglia. Aphasia classifications are as determined by the Western Aphasia Battery-Revised (Kertesz, 2006) and Transcortical is abbreviated (Trans).

Pearson correlation coefficient was calculated for any variable found to be significantly

predictive of either outcome (i.e., change in number or percent CIUs) with the predicted outcome

measure.

5.3 Results

5.3.1 Neuroimaging Findings

The mean lesion size was 89.4 cm^3 (range= 14.1 - 219.1, SD= 52.4). Lesion sizes for individual participants are included in Table 5.2.

5.3.2 Behavioral Measures

The mean change in number of CIUs produced for the narrative task from pre- to posttherapy (Week 6) was 34.36 (SD = 55.15; range -58 to 129). This increase was significant at α = 0.05 (t(17) = -2.64, p= 0.009) with an effect size of 0.377. There were no significant differences between the two pre-therapy sessions (t(17) = -0.17; p = 0.864) or the two post-therapy sessions (t(16) = -0.50; p = 0.622).

The mean change in percent of CIUs produced for the narrative task from pre- to posttherapy (Week 6) was 3.99 (SD= 7.88; range -15.01 to 15.20). This increase was significant at α = 0.05 (t(17)= -2.14, p= 0.023) with an effect size of 0.215. There were no significant differences between the two pre-therapy sessions (t(17) = 0.375; p = 0.712) or the two post-therapy sessions (t(16) = -1.08; p= 0.294).

Stepwise regression performed to identify variables significantly predictive of the change in number of CIUs produced following therapy selected no variables. For the change in percent of CIUs produced following therapy, the sole variable selected was the number of sessions completed (F= 6.67; p = 0.020; MSE= 6.82).

This variable was found to have a significant positive correlation with post-therapy change in percent of CIUs produce during the narrative task (r= 0.542; p = 0.020). Figure 5.1 shows the relationship between these variables.



Figure 5.1 Number of Sessions Completed and Percent Change in CIU Production. Y axis shows the change in percent of words produced that were CIUs during the Cinderella narrative task immediately following therapy (Week 6) compared to baseline. Figure legend indicates individual subjects by aphasia type. CIU= correct information unit.

5.4 Discussion

The present analysis examines the effects of imitation-based aphasia therapy on generalization of therapeutic effect. We find that patients with aphasia demonstrate significant improvement on a narrative task following a 6-week period of intensive therapy involving repetition of words and phrases. This behavioral outcome is of particular interest as the practiced task is quite dissimilar to the measure on which improvement was demonstrated. Failure to achieve generalization is a significant obstacle in aphasia treatment, and it is most common for benefits of therapy to be restricted to precisely the task and items that are explicitly trained, with no effect on untrained items, even on the identical task (Pring, Hamilton, Harwood, & Macbride, 1993). However, recent studies exploring speech entrainment, or online imitation, have found generalization to untrained scripts as well as to spontaneous speech (Fridriksson et al., 2012). These findings may suggest a unique benefit of imitation-based therapy. Our two outcome measures on the narrative task are the number and percent of CIUs produced following therapy. Overall, our participants demonstrate significant increases for both of these measures. These changes indicate that individuals with aphasia are able to produce narratives that are more informative and efficient following intensive imitation practice, despite the fact that neither the narrative task nor the specific words and phrases used in the task are related to the content of training. Additionally, that there were no significant differences found for these measures six weeks following the termination of treatment suggests that they are able to maintain these benefits once achieved. Maintenance of treatment effects, similar to generalization to untrained behaviors, is an infrequently met treatment goal (Dechene et al., 2011; Fridriksson et al., 2012), making this finding of particular practical interest.

None of the demographic, behavioral, or neurological measures included in our regression model predicts the change in number of CIUs produced following therapy. However, individual compliance with the requested therapy intensity does vary in our sample, and we do find a positive correlation between change in percent of CIUs produced and the number of sessions completed over the course of treatment. This is perhaps unsurprising, as more treatment (Carpenter & Cherney, 2016) and higher intensity (Bhogal et al., 2003) are associated with better outcomes following therapy. However, it may be unexpected that this measure trumps other variables previously found to be associated with aphasia prognostication, such as time since onset (Pickersgill & Lincoln, 1983) and lesion size (Plowman, Hentz, & Ellis, 2012).

High intensity was a key design feature of IMITATE. It is theorized that the sort of massed stimulation provided by our treatment program supports implicit learning through the repeated engagement of neural pathways which are consequently strengthened, or that lead to the formation of new pathways (Ferguson, 1999). While the present study does not explore the brain

changes underlying our findings, generalization following aphasia therapy, such as that demonstrated by our participants, is associated with changes in functional connectivity as measured by functional magnetic resonance imaging (Sandberg, Bohland, & Kiran, 2015). Further suggestion that synaptic changes facilitate generalization can be found in the literature on transcranial direct current stimulation, a form of noninvasive brain stimulation that is found to promote generalization, as well as maintenance, when paired with aphasia therapy (de Aguiar et al., 2015; Meinzer, Darkow, Lindenberg, & Floel, 2016).

There is strong evidence for causality in the relationship between our therapy and changes on the narrative task, as significant changes occur only over the duration of therapy. No significant changes occur during the two equally spaced pre-therapy assessments, or during two similar post-therapy sessions, allowing us to be quite confident that it was indeed the therapy that produced the observed effect. Our findings of a significant positive correlation between the number of sessions completed and the change in percent CIUs produced may or may not be causal. It is possible that participants who did not feel that the therapy was benefiting them were therefore less motivated to complete as many sessions, and thus the poor performance actually caused fewer sessions to be completed. This explanation is judged to be unlikely, however, as individuals with poor compliance completed fewer sessions throughout the entire 6-week duration of treatment, rather than reducing participation over the course of the program as might be expected in the case of waning enthusiasm for the therapy. However, it remains possible that another factor such as attention – known to be impaired in aphasia (Tseng et al., 1993) – may have resulted in both limited participation and lesser benefit of therapy.

The present treatment study uses a biologically motivated approach to aphasia therapy and finds significant generalization beyond the practiced task of imitation. Further, the more therapy

sessions that are completed, the greater is the therapeutic benefit. This is interpreted as support for our hypothesis that through the engagement of populations of neurons active during the observation and execution of speech, we are strengthening those networks' ability to support the production of speech. By using a general approach that targets the processes subserving imitation, we are able to achieve generalized benefits in the domains of speech and language.

Chapter 6: Changes in Dynamic Resting State Network Connectivity Following Aphasia Therapy

6.1 Introduction

Resting state functional magnetic resonance imaging (rsfMRI) permits observation of neural networks produced by correlated low frequency activity across the brain when not engaged in any particular task. These low frequency fluctuations in the hemodynamic response found in rsfMRI are believed to serve as proxy for estimating baseline neuronal activity in the brain, and the relationships that emerge reflect networks that typically are engaged in some shared function (Damoiseaux et al., 2006). Resting state networks (RSNs) have been investigated extensively in healthy controls, and to a lesser extent in neurological disease, including Alzheimer's disease and schizophrenia (Rombouts, Barkhof, Goekoop, Stam, & Scheltens, 2005; Woodward, Rogers, & Heckers, 2011). Examination of RSNs in the study of acute and chronic stroke has been less common, and there is limited understanding of how these networks are associated with functional behavioral deficits and recovery. Characterizing RSNs in chronic stroke, and in particular, individuals with post-stroke deficits such as aphasia, could lead to important biomarkers for therapeutic selection and prognosis, or as an objective measure of rehabilitation.

In aphasia, treatment with any of myriad traditional approaches has some benefit although there is no scientific basis for choosing an appropriate therapy for a given individual, and scant evidence that it even makes a difference (Brady et al., 2012). There is also little insight thus far into the considerable variability in treatment outcomes among patients. Further, the mechanisms of plasticity underlying functional gains experienced with therapy remain debated. Models of plasticity associated with aphasia recovery are typically localizational in nature, focusing on

increasing or decreasing activation in regions thought, respectively, to underlie compensatory or maladaptive reorganization of cortices involved in language tasks (Abel, Weiller, Huber, Willmes, & Specht, 2015; Marcotte et al., 2012). As the field of neuroscience increasingly recognizes the brain as the sum of its connections to a greater extent than its parts, it is appropriate that investigation of aphasia recovery should likewise refocus on the interrelationship of neural networks associated with gains in language ability following therapy. This could lead to novel approaches to solving the problems related to therapy selection, explaining variability, and providing prognosis.

Previous investigation in rsfMRI indicates disrupted functional connectivity following stroke that is significantly associated with degree of language impairment (Zhu et al., 2014) and also suggests that RSNs change with aphasia therapy. Intensive treatment with semantic feature analysis results in increased integration within the default mode network, and this increased integration is positively correlated with better treatment outcome (Marcotte, Perlbarg, Marrelec, Benali, & Ansaldo, 2013). Increased functional connectivity has also been found within language networks, with reorganization following treatment to appear more similar to healthy controls (van Hees et al., 2014).

Appreciation of the interconnected nature of the brain is not without its own controversies. Graph theoretic analyses of the brain suggest that its complex composition simultaneously requires segregation of functions, and therefore of regional activations and their connectivity, as well as integration of these units to form a cohesive whole capable of directing behavior (Sporns, 2013). Thus, segregation promotes specialization within a unit, while integration facilitates the unification necessary to coordinate interaction among these specialized units. Stroke necessarily disrupts both of these processes to some extent, and functional recovery may hinge on their

return to a relative balance (Falcon et al., 2015).

Currently, most investigations of functional connectivity collapse across scan time, using a single statistic to represent the aggregate interactions within a network or between regions of interest. While this work is of significant importance, the dynamic nature of brain function, perhaps most acutely assessed by electroencephalography (EEG) or electrocorticography (ECoG), but also informative at the lower temporal resolution of rsfMRI, permits unique insight into the time-dependent roles of neural networks (Calhoun, Miller, Pearlson, & Adali, 2014). Sliding window approaches to functional connectivity examine time-varying correlations among RSNs. Such approaches can extend our understanding of brain function by providing more nuanced measurement of network dynamics, which are particularly valuable in deciphering the neural underpinnings of human behavior (Hutchison et al., 2013).

The present study interrogates the changing nature of the relationships among RSNs over the course of a novel, computer-based imitation therapy program (IMITATE; Duncan et al., 2016; Lee et al., 2010) as it relates to behavioral gains in the production of meaningful language. We analyze rsfMRI scans acquired before and after treatment to better understand the mechanisms of plasticity underlying recovery of language functions. We hypothesize that aphasia patients will demonstrate changes in the correlations of RSNs associated with behavioral changes following therapy.

6.2. Materials & Methods

6.2.1 Participants

The nineteen patients recruited for participation in the study are described in Section 3.1 and Table 3.1, with lesion overlap depicted in Figure 3.1. From this group, twelve participants were selected for the present analysis on the basis of participation in three baseline rsfMRI scans prior to initiation of therapy (excluding subjects 1,2,5,18,20). Of the fourteen participants who qualified, two additional subjects (14, 17) were excluded due to excessive motion, leaving the twelve subjects included in this analysis. These subjects were overall similar to the larger sample (age range 31-70; mean= 52.08; SD= 11.75; 3 female (25%)). All participants had sustained a single stroke 7 to 124 months prior to enrollment (mean = 40.33; SD 42.55). Additional information about the therapy can be found in Sections 3.3 and 3.4.

6.2.2 Behavioral Measures

The behavioral task and related significant results are reported in Chapters 3 and 5, respectively. Participants were recorded telling the story of the Cinderella fairytale (Saffran et al., 1989) twice before and twice following six weeks of intensive aphasia therapy (see Section 3.2 and Figure 3.2 for study design). The number of correct information units (CIUs; (Nicholas & Brookshire, 1993)) produced was calculated for each time point (see Sections 3.4.3 for methods and 5.3.2 for results). One subject (10) missed the second post-therapy (Week 12) behavioral evaluation.

6.2.3 Neuroimaging

6.2.3.1 Acquisition

Magnetic resonance imaging (MRI) was acquired at three time points before therapy (Weeks -6, -3, 0) and one to three time points after its conclusion (Weeks 6, 9, 12). Study description can be found in Section 3.2 and acquisition details can be found in Section 3.5.1. Both high resolution anatomical and five minutes of functional resting state data were acquired.

6.2.3.2 Preprocessing

The first four rsfMRI volumes were discarded to ensure the steady state of the scanner. Preprocessing of rsfMRI images consisted of slice timing correction, despiking, and registration using AFNI (Cox, 1996) and FSL (Smith et al., 2004). A Gaussian smoothing kernel of 4 mm FWHM was applied to increase signal-to-noise ratio. White matter, ventricle and hand-drawn lesion masks were used to create nuisance time series that were regressed out of the signal along with motion and polynomial (linear and quadratic) trends using AFNI's 3dDeconvolve. For time points containing motion greater than 3 mm, neither the volume containing the movement nor the one following it were included in the regression. After deconvolution using AFNI's 3dDeconvolve, these censored time points were filled using a cubic spline interpolation to facilitate the continuity required for the sliding window analysis described below. Band-pass filtering (0.01 to 0.1 Hz) was then performed in order to identify the low frequency fluctuations of interest.

Anatomical scans were used to create a common template following reconstruction of the cortical surface with FreeSurfer (Fischl, 2012) facilitated by Virtual Brain Transplant (Solodkin et al., 2010). For each time point, an anatomical scan was registered to the rsfMRI for individual preprocessing, and then to this common template, with individual transformation matrices applied to the band-pass filtered rsfMRI data to permit group analysis.

Two subjects (11,19) missed the scan immediately post-therapy (Week 6). Two subjects (4,10) missed the scan at the 6-week maintenance interval (Week 12).

6.2.4 Dynamic Functional Network Connectivity

Spatial independent component analysis (ICA) and dynamic functional network connectivity

(dFNC) were performed using the Group ICA of fMRI Toolbox (GIFT; Calhoun, 2004).

Time series were initially mean-centered, and then whitening and dimension reduction were achieved using subject-specific principal component analysis (PCA) to extract the first twenty eigenvectors (low-order Gaussian features). Group ICA was then performed using the Infomax algorithm to identify twenty independent higher-order non-Gaussian features of the reduced data. ICASSO (Himberg, Hyvarinen, & Esposito, 2004) was used to ensure the stability of the extracted features, repeating the decomposition ten times using random initiation and bootstrapping.

Of the twenty components yielded via ICA, eight were identified as true components that did not overlap with regions of known vascular, motion, and susceptibility artifacts and that were consistent with RSNs previously identified in the literature (Damoiseaux et al., 2006; Lee et al., 2012). These components were used to compute dynamic functional network connectivity via a sliding window approach using windows of twenty volumes (30 s) in steps of one volume (1.5 s). Edges were tapered by convolving a rectangular window with a Gaussian function using the standard deviation of three windows (i.e., the window of interest and surrounding windows). Due to the brevity of the window (30 s) and concomitant risk of increased influence of noise, sparsity was induced by applying graphical LASSO (Friedman, Hastie, & Tibshirani, 2008), with lambda value optimized for each scan using cross-validation. Following estimation of covariance among these eight RSNs, values were Fisher-Z transformed to permit valid comparison across individuals.

Normalized covariance relationships between networks within a given window were then grouped into ten states using k-means clustering. The number of states was selected based on the amount of variance accounted for by each added state in order to control for overfitting. The

cutoff (k) was selected as the largest number of clusters that explained more variance (measured by sum of squared L_1 distances) than did k-1 clusters.

6.2.5 Individual Correlations in Functional Connectivity States and Behavioral Measures

Differences in the amount of time (number of 30-second windows) spent in each of the ten functional network connectivity states ("dwell time") were correlated with differences in number of CIUs produced before and after therapy to identify states that are facilitative or obstructive to treatment benefit. Pre-therapy dwell times were calculated as the mean number of windows clustered in a particular state during the three baseline scans (Weeks -6, -3, 0). Post-therapy dwell times were calculated as the mean number of windows clustered in a particular state during the one to three scans acquired after conclusion of therapy (Weeks 6, 9, 12). The pre-therapy CIU measure was defined as average number of CIUs for the two pre-therapy assessments. These correlations were corrected for multiple comparisons (α =0.05/10 states).

If a state was significantly correlated with behavioral change, three further correlations were calculated to examine the relationship between changes in dwell time and number of CIUs produced for each individual post-therapy time point (Week 6 vs. baseline, Week 12 vs. baseline), as well as between the two post-therapy time points (Week 12 vs. Week 6). Pearson correlation coefficients were corrected for multiple comparisons (α =0.05/3 comparisons). These comparisons were also repeated using a partial correlation controlling for number of CIUs produced pre-therapy.

A repeated measures ANOVA was used to compare baseline dwell times to ensure that connectivity states did not significantly differ among pre-therapy scans.

6.3. Results

6.3.1 Resting State Networks (RSNs)

The eight RSNs identified via group ICA are pictured in Figure 6.1 (highest 5% of Z scores for mean components for the group). They are most consistent with a dorsal attention network (DAN), frontoparietal control network (FPC), default mode network (DMN), language network (LAN), left ventral attention network (LVAN), right ventral attention network (RVAN), sensorimotor network (SMN), and visual network (VIS) (Lee et al., 2012). Four of these networks (DAN, DMN, LAN, SMN) appear to be more right lateralized than those found in healthy controls, consistent with left hemisphere lesion. Figures showing the overlap of these components with those from healthy controls can be seen in Appendix B (Figures B.1 – B.8).



Figure 6.1 Resting State Networks (RSNs) Displayed on a Group Template. A: dorsal attention network (DAN); B: default mode network (DMN); C: frontoparietal control network (FPC); D: language network (LAN); E: left ventral attention network (LVAN); F: right ventral attention network (RVAN); G: sensorimotor network (SMN); H: visual network (VIS). As radiological convention is used, left hemisphere is depicted on right side.

The twelve artifactual ICA components that were excluded from further analysis can be seen in Appendix A (Figures A.1 – A.12). Similarity estimates for the ten repetitions of ICA can be seen for all components in Figure 6.2, with identified RSNs labeled.



Figure 6.2 Similarity Graph for ICASSO Estimates. Each of the ten repetitions of ICASSO (Himberg et al., 2004) for each of the twenty components detected with independent component analysis (ICA), visualized as a unitless two-dimensional plot. Each dot represents a component estimate from one repetition of the ICA decomposition. A circle estimates the centroid of each individual component cluster. Tighter clusters indicate more reliable components. Numbers represent components that were identified as artifactual (see Appendix A). Acronyms represent components identified as resting state networks (RSNs) and are abbreviated as follows: DAN= dorsal attention network; DMN= default mode network; FPC= frontoparietal control network; LAN= language network; LVAN= left ventral attention network; RSNs can be seen in Figure 6.1.

6.3.2 Dynamic Functional Network Connectivity

Figure 6.3 shows the correlation matrix for the single dynamic functional network

connectivity state (state 10) found to correlate significantly with changes in behavior (α =

0.05/10; see below).
State 10 may be described as a state of near zero correlation among RSNs (mean = 0.014; range -0.105 to +0.134). An ANOVA comparing baseline dwell times in state 10 indicated no significant differences among the three pre-therapy sessions (p = 0.376).

6.3.3 Individual Correlations in Functional Connectivity States and Behavioral Measures

Dwell time in a single state (state 10; See Figure 6.3) was found to change in tandem with changes in number of CIUs produced ($\alpha = 0.05/10$) when pre- and post-therapy measures were compared. Pearson correlation coefficient between increased number of CIUs produced and increased dwell time in state 10 was significant (r = 0.785; p = 0.004).



Figure 6.3 Correlation Matrix Representing State 10. Increased time spent in state 10 was positively correlated with increase in correct information units (CIUs) on the narrative production task at the maintenance interval compared to baseline (p = 0.002) and immediately post-therapy (p = 0.012). From left to right on the x axis and top to bottom on the y axis, resting state networks (RSNs) are: dorsal attention (DAN), default mode (DMN), frontoparietal control (FPC), language (LAN), left ventral attention (LVAN), right ventral attention (RVAN), sensorimotor (SMN), and visual (VIS). Color bar shows binned correlation values, upper triangle shows actual Pearson correlation coefficients (r).

The nine additional states dynamic functional network connectivity states that were not significantly correlated with behavioral change can be seen in Appendix C (Figure C.1).

When each of the post-therapy CIU scores and state 10 dwell times were compared to baseline, the Week 6 comparison was significant at α = 0.05 (p= 0.033), but did not withstand correction for multiple comparisons. The Week 12 comparison was significant when corrected for multiple comparisons (r = 0.842; p = 0.002), as was the comparison between Week 6 and Week 12 (r = 0.821, p = 0.012). Figure 6.4 shows the relationship between changes in dwell time in state 10 and changes in number of CIUs produced post-therapy compared to pre-therapy and at Weeks 6 and 12 compared to baseline. Comparison using partial correlation to control for pre-therapy CIU production remained significant at α =0.05 for Week 12 performance compared to Week 6 (r= 0.819, p= 0.024; see Figure 6.5), with trends for post- vs. pre-therapy (r= 0.565, p= 0.089) and for Week 12 compared to baseline (r= 0.650, p= 0.058).



Figure 6.4 Changes from Baseline in State 10 Dwell Time and CIU Production. Left panel averages across all post-therapy behavioral and neuroimaging measures (including Week 9, at which time no behavioral measures were recorded). Middle panel indicates differences between Week 6 and baseline. Right panel indicates differences between Week 12 and baseline. CIU= correct information unit.



Figure 6.5 Post-therapy Changes in State 10 Dwell Time and CIU Production. Both x and y axis show contrasts between Week 12 (6 weeks following therapy) and Week 6 (immediately post-therapy). As two subjects (11,19) missed the scan immediately post-therapy and two subjects (4,10) missed the scan at the maintenance interval, this analysis includes only 8 participants. CIU= correct information unit.

6.4 Discussion

The present study finds a shift in dynamic functional resting state connectivity associated with improvement on a narrative production task. Individuals demonstrating behavioral change (i.e., number of novel, correct information units produced during a narrative task) also demonstrated change in the amount of time spent in one particular state of brain functional connectivity. Improvement on the language task was associated with concomitant increase in the amount of time spent in a functional connectivity state best characterized by its minimal correlation between any of the RSNs identified here. This finding suggests that it may be increasing segregation among RSNs that drives recovery, at least in response to the imitation-based behavioral treatment used in this study.

In graph theoretic models of brain connectivity, the RSNs described here might be conceived of as distinct modules. In this conceptualization, a greater proportion of intermodular connections (i.e., among RSNs) would represent increased integration, and a greater number of intramodular connections (i.e., within RSNs) would represent increased segregation. The driving force behind the tendency towards increased segregation, as observed in this investigation, might relate to the connectivity of specific hubs, or regions in the network. Connector hubs link different modules to a greater extent than any one single module, and thus promote integration, whereas provincial hubs, which are highly interconnected within a module and have few or no connections outside of it, increase segregation (Sporns, Honey, & Kotter, 2007).

Following this model, improvement in accurate, qualitative narrative production is driven by an increased ability to isolate those networks necessary to perform the task for individuals with chronic aphasia. This suggests that integration may be increasing within individual networks, as has been found previously (Marcotte et al., 2013; van Hees et al., 2014), yet with the global balance consequently shifting towards segregation of functional units. This is consistent with findings of more focal, efficient and presumably specialized activation that is observed in expert compared to novice motor control (Milton et al., 2007) and is associated with functional gains in motor (Ward, Brown, Thompson, & Frackowiak, 2003) and language (Abel et al., 2015) in stroke recovery.

When the architecture of the brain is disrupted by a focal lesion, many connections are necessarily lost. Processes of homeostatic plasticity attempting to ensure that surviving neurons continue to receive adequate input may cause inappropriate connections to form (Murphy & Corbett, 2009), and proximity will dictate that short-range connections are the most likely. The increased segregation associated with improvement in the present study is believed to represent the loss of short-range connections (Fair et al., 2007) that were not adaptive. The IMITATE therapy, designed to engage a cortical network underlying both observation and execution of

speech (Lee et al., 2010), induces experience-dependent Hebbian plasticity that likely increases synaptic weights within individual RSNs via long-term potentiation, while other (maladaptive) connections become weaker or are lost entirely.

Decreased segregation among brain networks is observed in healthy aging, and greater segregation predicts superior episodic memory across the lifespan (Chan, Park, Savalia, Petersen, & Wig, 2014), suggesting a positive role for specialization of neural networks in cognitive ability. Decreased clustering and local efficiency, both measures of segregation, have also been found for somatosensory and visual motor networks in Parkinson's disease (Tinaz, Lauro, Hallett, & Horovitz, 2016). These findings suggest that segregation of neural networks decreases with normal and pathological degeneration, and that this deterioration underlies the loss of functional skills.

This association between RSN segregation and improved verbal communication has a number of potentially significant clinical implications. For example, if segregation plays an adaptive role, behavioral interventions might seek to engage isolated functional RSNs through specific tasks, aiming to enhance network segregation through experience-dependent plastic changes. Further, it may be possible to enhance such segregation through pharmacotherapy or genetic intervention, to the extent that these distinct brain networks have different characteristics at a cellular level (e.g., dopamine in the default mode network). Although this line of reasoning requires considerable further investigation, we anticipate that the functional distinctions in the organization of these intrinsic RSNs may reflect important cellular and molecular differences that can provide therapeutic targets – that is, that there are neurotransmitters or receptors that can be targeted to induce integration within RSNs and thus promote segregation between them. Noninvasive brain stimulation techniques, such as transcranial magnetic stimulation and

transcranial direct current stimulation, may also be used to target regions included in one network that are anatomically distinct from regions in other networks. The most effective treatment might be a highly specified behavioral program seeking to target individual networks in combination with both physiological stimulation and pharmaceutical intervention, which in aggregate are designed to enhance synaptic plasticity in the individual networks.

Although the clinical possibilities are exciting, the present study does have several limitations. One is the small sample size and heterogeneity of the patient group in measures of age, aphasia type, baseline performance, and lesion size/location. Additionally, while the findings of reduced network coupling associated with behavioral improvement are provocative, the nature of this analysis does not permit exploration of intra-network dynamics. Future investigations will focus on understanding the roles of specific network relationships and the regions that contribute most significantly.

While a task as complex as narrative production certainly engages multiple networks, including those supporting sensorimotor, attention and cognitive control as well as language capacities, it is of great interest that it may be the functional integrity of the individual networks, rather than the strengthening of the connections between them, that underlies improvement on this task. Although the entire network requires integration of its constituent pieces in order to function, segregation of these components into synchronized individual units is paramount. Enhancing the specialization of distinct neural networks is key to achieving optimal functioning of the unified brain in chronic aphasia.

Chapter 7: Network Properties Underlying Changes in Narrative Production

7.1 Introduction

Graph theoretic approaches provide simplified metrics to characterize complex networks. By defining regions as nodes and the (functional or structural) connections between them as edges, the brain can be modeled as a graph and analyzed through the pairwise relationships among its component parts (Rubinov & Sporns, 2010). As measured by resting state functional magnetic resonance imaging (rsfMRI), our brains demonstrate the same types of topological properties found in other complex networks across various systems (Bullmore & Sporns, 2009).

Modularity is one key organizational principle found in social and biological systems (Girvan & Newman, 2002), including the human brain. To characterize the modularity of a network, individual nodes are first assigned to discrete communities by various methods. The modularity value of the network quantifies how many of the edges connected to a given node are also connected to another node within the same community, compared to if those edges had been distributed randomly (Newman & Girvan, 2004). Thus, modularity is essentially an index of how cleanly a network can be subdivided with a given partition, with higher values indicating more distinct subnetworks, or a greater level of segregation.

Figure 7.1 shows two different graphs that share many network characteristics. Each has 34 nodes. They share a similar number of edges connecting those nodes (7.1A = 81, 7.1B = 78). They also share a similar degree distribution, with each node having, on average, just under five connections to other nodes (7.1A = 4.59, 7.1B = 4.76). Each graph has been separated into five communities, indicated by node color, as this was determined to be the optimal number of communities for maximizing modularity for each of these networks individually. However, the

communities in Figure 7.1A overlap and share many edges among them, whereas the communities in Figure 7.1B are more discrete, and their modularity scores reflect this difference (0.29 and 0.41, respectively). This major difference in their organization is not surprising as the graph in 7.1A is a random construction, whereas the graph in 7.1B represents data from a social network (Zachary, 1977).



Figure 7.1 Visual Depiction of Two Graphs. A and B share similar network properties, including number of nodes, edges, communities, and average degree, yet the modularity of B (0.41) is much higher than that of A (0.29). A is a random network; B depicts data from a social organization (Zachary, 1977).

As modularity is a basic characteristic of complex networks under normal circumstances, it might be expected that a disruption to that network would result in a decrease in its modularity. In the brain, deprivation of blood flow results in changes in both cognitive function and functional connectivity and modularity as assessed by rsfMRI. For patients with unilateral carotid stenosis, modularity is negatively correlated with performance on neuropsychological tasks, including the Mini-Mental Status Examination (MMSE) and measures of reading and memory (Chang et al., 2016), and positively correlated with better post-operative cognitive outcomes (Soman et al., 2016). Focal lesions are also found to reduce measures of modularity (Gratton, Nomura, Pérez, & D'Esposito, 2012).

In healthy adults, higher modularity is associated with better working memory performance, a finding thought to reflect efficient organization and transmission of information throughout the brain (Stevens, Tappon, Garg, & Fair, 2012). It may, therefore, be unsurprising that measures of modularity are decreased in patients with Alzheimer disease (AD). This is also true for cognitively intact individuals demonstrating preclinical biomarker pathology suggestive of AD (Brier et al., 2014) when compared to controls without such pathology. However, even in healthy aging, brain changes are associated with alteration in its modular organization. Whole brain modularity declines with aging across the life span (Onoda & Yamaguchi, 2013), including when examining the modularity of intrinsic resting state networks (RSNs) exclusively (Song et al., 2014).

In a recent investigation of changes in dynamic functional network connectivity (dFNC) in patients with aphasia (see Chapter 6), we investigated the amount of time spent by each participant in one of a small number of states determined by clustering the dynamic network correlations. We found that imitation-based aphasia therapy led to improvement in narrative production that is associated with an increasing amount of time spent in one of these states, a state characterized by minimal correlations among resting state networks (Duncan & Small, under review). We interpret these findings as evidence for an association between greater functional segregation and better performance, as previously demonstrated in healthy aging (Chan et al., 2014) and Parkinson's disease (Tinaz, Lauro, Hallett, & Horovitz, 2016). They are also consistent with the finding that diffuse patterns of activation are replaced by more focal, and presumably efficient, organization as behavior improves, as has been found in motor learning (Milton et al., 2007), as well as post-stroke motor (Ward et al., 2003) and language (Abel et al., 2015) rehabilitation.

In the present study, we examine the hypothesis that assigning community structure based on membership in well-established resting state networks will result in changes in modularity that are positively correlated with behavioral changes in narrative production, as found in dynamic functional connectivity. Such findings would support the notion that successful therapy leads to increased network segregation, and would provide insight into the mechanism underlying behavioral improvement following treatment.

7.2. Materials & Methods

7.2.1 Participants

Nineteen native English speakers with chronic aphasia secondary to ischemic stroke participated in a larger study of intensive, imitation-based aphasia therapy (Duncan et al., 2016; Lee et al., 2010), as described in Section 3.1. Fourteen subjects were selected from that superset based on participation in three baseline rsfMRI scans prior to the initiation of therapy (see Section 6.2.1 for excluded subjects). Two additional subjects were excluded due to excessive motion during scanning (see Section 7.2.3.2 for details). The remaining group included in this analysis consisted of twelve individuals (3 female; 25%) ages 31 to 70 years (mean= 52.08; SD= 11.75) who had sustained a single stroke 7 to 124 months prior to enrollment (mean= 40.33; SD 42.55). Further information about the therapy is included in Sections 3.3 and 3.4.

7.2.2 Behavioral Measures

A full description of the Cinderella task and associated behavioral results are reported in Chapters 3, 5, and 6. Participants were recorded telling a narrative (Cinderella; Saffran et al., 1989) four times over an eighteen week span (Weeks -6,0,6,12) during which the middle six weeks (Weeks 0 to 6) consisted of the IMITATE therapy (see Sections 3.3 and 3.4). The recorded narratives were scored for number of correct information units (CIUs) produced on the basis of whether words were novel, intelligible, and appropriate to the context (see Sections 3.4.3 for methods and 5.3.2 for results). Figure 3.2 depicts the study design. One subject (10) missed the fourth behavioral evaluation.

7.2.3 Neuroimaging Measures

7.2.3.1 Acquisition

Magnetic resonance imaging (MRI) was acquired at three baseline time points prior to six weeks of therapy (Weeks -6, -3, 0) and at up to three time points following the end of treatment (Weeks 6, 9, 12). Figure 3.2 depicts the study design, and study description can be found in Section 3.2. Acquisition details for structural and functional scans are included in Section 3.5.1. Five minutes (200 volumes) of functional resting state data were acquired.

7.2.3.2 Preprocessing of Resting State fMRI

RsfMRI preprocessing consisted of discarding the first four volumes, slice timing correction, despiking, and registration performed using AFNI (Cox, 1996) and FSL (Smith et al., 2004). The AFNI function 3dDeconvolve was used to regress out signals of no interest (from white matter, ventricles, lesion) as well as motion and polynomial (linear, quadratic) trends. If a volume had >3 mm displacement from the volume to which it was being registered, both that volume and the following one were censored and not included in the regression. A scan needed to have $\ge 55\%$ of volumes uncensored (108) to be included in the analysis. These cleaned time series were then band-pass filtered (0.01 to 0.1 Hz) to identify the low frequency fluctuations of interest. As some

participants missed or had excessive motion during one or two scanning sessions, a total of 76 scans were included in this analysis (rather than 12 subjects x 7 scans = 84).

7.2.3.3 Anatomical Preprocessing

Lesion masks drawn on the high-resolution structural scans were used to perform a Virtual Brain Transplant (Solodkin et al., 2010) to facilitate reconstruction of each participant's cortical surface with FreeSurfer (Fischl, 2012), brain parcellation into 463 regions (Hagmann et al., 2008), and the creation of a common template. Each participant's preprocessed and band-pass filtered rsfMRI was registered to the common template to permit group analysis.

7.2.3.4 Independent Component Analysis (ICA)

Spatial ICA was performed using the Group ICA of fMRI Toolbox (GIFT; (Calhoun, 2004)). In a preprocessing step prior to ICA, the time series underwent mean centering followed by whitening and dimension reduction using subject-specific principal component analysis (PCA) to extract the first twenty eigenvectors (low-order Gaussian features). Group ICA using the Infomax algorithm then identified twenty independent higher-order non-Gaussian features of the reduced data. ICA was repeated ten times with random initiation and bootstrapping to ensure the stability of the identified components (ICASSO; Himberg et al., 2004).

7.2.3.5 Graph Construction

Eight of the twenty identified components were selected as components of interest (Duncan & Small, under review) that did not overlap with regions of known vascular, motion, and

susceptibility artifacts and that were consistent with RSNs previously identified in the literature (Damoiseaux et al., 2006; Lee et al., 2012). The twelve components that were discarded can be seen in Appendix A (Figures A.1 through A.12).

We took the inverse of the transformation that registered individual scans to the group template and applied it to these eight RSNs in order to bring them from standard space into the native space in which the rsfMRI data were acquired. We then identified the anatomical regions included for all subjects as part of a given component (i.e., the intersection) through the use of the parcellation scheme applied during preprocessing of the anatomical volume (Hagmann et al., 2008). Each RSN included between 24 and 40 regions, for a total of 230 regions among the eight RSNs.

After identifying the peak voxel for each region included any one of the eight RSNs for each scan, we mean centered the time series and constructed a joint covariance matrix for all time series in all 230 regions. Sparsity was induced using the graphical lasso (Friedman et al., 2008) implemented in R (Friedman, Hastie, & Tibshirani, 2014), and the resulting inverse covariance matrix was used to construct a graph for each scan using NetworkX (Hagberg et al., 2004). In these graphs, each region is a node, and the edges are weighted by the strengths of the functional connectivity (covariance).

7.2.3.6 Modularity

Processing of the functional connectivity graphs included first removing all negative edges and then binarizing the remaining positive edges. We assigned each node of these resulting graphs to a module based on its membership in one of the original eight RSNs, and then computed a modularity value for each scan. A modularity measure (Q) was calculated through use of the NetworkX community.modularity function (Hagberg, Schult, & Swart, 2008) based on a well-established metric (Newman & Girvan, 2004).

To control for graphs simply changing in modularity without any consideration of RSNs, nodes were also assigned to modules based on a separate partition determined via application of a community detection algorithm using the Louvain method (Blondel, Guillaume, Lambiotte, & Lefebvre, 2008), with a separate modularity score calculated.

7.2.4 Correlation of Modularity and Behavior

Differences in modularity (Q) were correlated with differences in number of CIUs produced before and after therapy. Pre-therapy modularity was calculated as the average Q of the three baseline scans (Weeks -6, -3, 0). Post-therapy modularity was calculated as the average Q for the scans (from 1 to 3 total) acquired after therapy (Weeks 6, 9, 12). The pre-therapy CIU score was defined as the average number of CIUs for the two pre-therapy assessments. The post-therapy CIU score was averaged across the two post-therapy testing sessions (Week 6, 12). Four comparisons were made for the changes that occurred in behavior and functional RSN modularity: post-therapy vs. pre-therapy, Week 6 vs. pre-therapy, Week 12 vs. pre-therapy, and Week 12 vs. Week 6. Due to our strong *a priori* hypothesis that increased RSN modularity would subserve improvement, based on our previous analysis (see Chapter 6), one-tailed t-tests were used. These correlations were corrected for the three independent comparisons being made (α =0.05/3).

Significant comparisons were repeated twice: (i) using partial correlations controlling for pre-therapy CIU production, and (ii) using a separate community assignment partition (see Section 7.2.3.6) to control for changes in modularity unrelated to RSNs.

A repeated measures ANOVA was used to compare baseline Q values to ensure that RSN modularity did not significantly differ among pre-therapy scans.

7.3 Results

There was a positive correlation between change in CIUs and change in RSN modularity comparing pre- and post-therapy measures (r= 0.687; p= 0.007) and comparing Week 6 measures to baseline (r= 0.760; p= 0.006). Figure 7.2 shows these relationships. Neither the correlation of Week 12 with baseline (r = 0.549; p = 0.05) nor Week 12 with Week 6 (r = +0.237; p = 0.286) was significant following correction for multiple comparisons. Reduced power due to missed scans (leaving 8 or 10 subjects) may have played a role in the failure of these comparisons to reach significance.



Figure 7.2 Correlations Between Changes in CIU production and RSN Modularity. Significant positive correlations for post- compared to pre-therapy measures (left) and for immediately post-therapy (Week 6) compared to pre-therapy. CIU= correct information unit; RSN= resting state network.

Controlling for pre-therapy CIU production, the pre- vs. post-therapy partial correlation (r= 0.758; p= 0.004) and the Week 6 partial correlation (r= +0.676; p= 0.023) remained significant at α =0.05. There were no significant results for the control comparisons, in which nodes were assigned to communities based on an independent algorithm, rather than RSN membership (|r|< 0.484; p> 0.110).

An ANOVA comparing baseline modularity indicated no significant differences among the three pre-therapy sessions (p = 0.273).

7.4 Discussion

The present analysis provides confirmatory evidence for the hypothesis that individuals demonstrating behavioral improvement in narrative production following imitation-based aphasia therapy demonstrate increased segregation among functional networks. We interpret this to mean that increased functional segregation supports better ability to communicate a narrative. The present finding, that of improved behavioral performance and increased network modularity when a node's community is assigned based on RSN association, also supports that notion of an adaptive role for increased segregation.

Decreased modularity is associated with functional deficits in a variety of disorders, including Alzheimer's disease and carotid stenosis (Brier et al., 2014; Chang et al., 2016). However, it should be noted that higher modularity is also associated with decreased performance in some studies. Increased modularity is found in patients with multiple sclerosis (Muthuraman et al., 2016), for whom it is negatively correlated with working memory (Gamboa et al., 2014). In Parkinson's disease, results have been mixed (Baggio et al., 2014; Ma et al., 2016). These findings may indicate that, as with essentially all biological properties, there is an inverse U-shaped curve associated with the modularity of functional brain connectivity. If the modular organization of the brain is either too weak or too strong, the behavior of the organism is maladaptive.

However, for the present findings, it is believed that there is explanatory power associated with the observed changes in brain connectivity. Studies of healthy controls indicate that modularity decreases when task-based fMRI is compared to rsfMRI (Di, Gohel, Kim, & Biswal, 2013), as well as when task demands increase (Vatansever, Menon, Manktelow, Sahakian, & Stamatakis, 2015). Given that individuals with aphasia demonstrate deficits in cognitive domains outside language (Murray, 2012), the observed recovery in narrative production may be associated not only with language-specific improvement but also with cognitive benefit, resulting in better behavioral performance as well as less required effort. Additionally, connectivity changes in the language system secondary to stroke can ripple throughout other brain subnetworks not directly related to language (Warren, Crinion, Lambon Ralph, & Wise, 2009), causing connectivity changes both within and between modules.

While graph theoretical analyses are generally underutilized in the study of aphasia and other sequelae of stroke, a few prior studies use such methods to illuminate behavioral changes occurring following injury and therapy. In aphasia, local increases in network connectivity are found in bilateral angular gyrus and left pars triangularis, part of Broca's area, in individuals demonstrating benefit from a word finding treatment (Sandberg et al., 2015). Aphasia severity is associated with disruption of regions serving as connector hubs in the language network and a global reduction in the rich club coefficient (Gleichgerrcht et al., 2015), a measure of the tendency for highly connected nodes to be highly interconnected with each other. Patients with the semantic variant of primary progressive aphasia demonstrate lower global efficiency, or more remote functional connections, compared to controls (Agosta et al., 2014), as do those with post-

stroke motor deficits (Falcon et al., 2015). These findings suggest, similar to those of the present investigation, that the ability to effectively manage and transmit information throughout the brain is compromised following stroke, that these changes underlie behavioral impairment, and that the restoration of these properties may facilitate functional gains.

How to restore normal network topology in the brain is a question distinct from these collected findings of disturbance, however. Targeted stimulation or inhibition of nodes and modules demonstrating deviant patterns of connectivity, with the aim of reinstating network balance, can currently be explored through transcranial magnetic stimulation or transcranial direct current stimulation. Future insight into the neural mechanisms through which these changes occur may come from "build to understand" approaches, such as modeling with The Virtual Brain (Jirsa, Sporns, Breakspear, Deco, & McIntosh, 2010). Better understanding the biological underpinnings of network disruption and reorganization will stimulate more informed and effective interventions that, in turn, will promote greater recovery.

Chapter 8: Conclusion

In this dissertation, we examined behavioral and neuroimaging measures associated with changes in performance following a six-week intensive course of imitation-based aphasia therapy. The therapy was motivated by specific neurobiological findings about motor function from primate physiology and human neuroimaging, in which certain neurons, called mirror neurons, and their network connections are active during both the observation and execution of motor acts, such as speech. The heterogeneous group of nineteen individuals with post-stroke aphasia who completed the therapy program underwent behavioral and neuroimaging assessment at multiple time points before and after therapy. This research design allowed us to establish a definitive baseline and to examine the often elusive maintenance of therapeutic effects following the termination of treatment.

In Chapter 4, we find that an individual's baseline variability in performance is a behavioral predictor of improvement on a practiced task, in particular, that intra-individual variability decreases in tandem with improvement on an audiovisual speech imitation task. Speech-language pathologists have long used this concept informally through the assessment of stimulability during initial diagnostic sessions, and it is intuitive, in hindsight, that one's performance on a task should become more consistent with practice. However, the work described in Chapter 4 provides formal evidence for this practice in targeting a skill based on the variability with which it is performed at baseline. It also represents a clear departure from the vast majority of the literature on intra-individual variability, which suggests that increases mark vulnerability and precede decline. As most of this work has been done in aging, it is noteworthy that a different trajectory applies in stroke recovery.

One of the most significant goals of virtually all therapeutic interventions, and also one of the greatest challenges, is the generalization of treatment effects. Chapter 5 explores generalization by investigating the impact of the practiced imitation task on a task that is very different functionally, specifically the retelling of a well-known fairytale. Participants are found to increase in two measures of narrative production, both the number and the percent of correct information units (CIUs) produced before compared to after therapy. The study described in Chapter 5 also explores maintenance of therapeutic benefit through comparisons between performance immediately after therapy and assessment six weeks later. Maintenance is a second major goal of, and challenge in, aphasia therapy, as treatment effects tend to be extinguished following the termination of treatment, yet our participants demonstrated no significant differences between performance immediately after and six weeks following therapy. The results of these behavioral findings are particularly striking, in which there is a strong generalization from the therapeutic task (audiovisual imitation) to a quite different one (story retell).

The studies described in Chapters 6 and 7 are highly inter-related, as it was the data-driven analysis in the former that gave rise to the hypothesis-driven analysis in the latter. Both studies examine changes in resting state functional connectivity to determine signatures facilitative of behavioral gains on the narrative production (story retell) task, and both use the same intrinsic resting state networks (RSNs) identified via independent component analysis (ICA). In Chapter 6, a sliding window approach is used to examine the amount of time participants spent in various states, characterized by correlations among the identified RSNs. The study of Chapter 6 showed improvement to be positively associated with an increase in time spent in a state in which RSNs were minimally correlated with each other. Thus, in Chapter 7, we conducted a graph theoretic

analysis to determine whether the network of RSNs was, in fact, becoming increasingly segregated. We assigned communities within the larger network based on membership in the subcomponent RSNs, thus assessing the overall modularity of the larger network. Improvement on the narrative task was positively correlated with RSNs evolving to be more differentiable modules, supportive of the hypothesis of increased segregation advanced in Chapter 6. Taken together, these two investigations offer powerful support for a new finding about brain changes enabling successful post-stroke language rehabilitation.

In its entirety, the work performed in this dissertation offers both practical insight into aphasia therapy (e.g., targeting goals based on variability) in addition to novel avenues for future research. The clearest starting point going forward is continued investigation of the changing network properties supporting language recovery through further graph theoretic analyses. Understanding which RSNs, and which particular nodes within them, are the most influential in producing the behavioral changes measured here could inform future anatomical targets for highprecision brain stimulation, such as transcranial magnetic stimulation (TMS). Lower resolution brain modulation with transcranial direct current stimulation (tDCS) could also be used with one of the more anatomically distinct networks, such as the visual network, to determine whether altering the connectivity in one RSN induces comprehensive changes throughout the larger network.

Finally, despite the biological motivation for this imitation-based therapy, it was not found to be significantly more (or less) effective than other existing aphasia therapies – whether currently accepted or experimental – with an effect size of 0.38 on the narrative task. Future development of a new aphasia intervention using imitation will further consider personal relevance, ecological validity, and the context-dependent nature of communication. Just as mirror neurons tuned to

grasping in macaque premotor cortices require a graspable object to fire, human language is a fundamentally goal-driven behavior, and it is believed that incorporating purpose into speech imitation will optimize the neural activation, network engagement, and behavioral efficacy achieved with therapy. Further study will also recruit a new and larger pool of participants in order to better validate extrapolation of the current findings and to more precisely identify biomarkers indicative of patients with predictable benefit from imitation-based intervention.

References

- Abel, S., Weiller, C., Huber, W., Willmes, K., & Specht, K. (2015). Therapy-induced brain reorganization patterns in aphasia. *Brain*, 138(Pt 4), 1097-1112. doi:10.1093/brain/awv022
- Agosta, F., Galantucci, S., Valsasina, P., Canu, E., Meani, A., Marcone, A., . . . Filippi, M. (2014). Disrupted brain connectome in semantic variant of primary progressive aphasia. *Neurobiol Aging*, 35(11), 2646-2655. doi:10.1016/j.neurobiolaging.2014.05.017
- Allaire, J. C., & Marsiske, M. (2005). Intraindividual variability may not always indicate vulnerability in elders' cognitive performance. *Psychol Aging*, 20(3), 390.
- Altschuler, E. L., Vankov, A., Hubbard, E. M., Roberts, E., Ramachandran, V. S., & Pineda, J. A. (2000). Mu wave blocking by observation of movement and its possible use as a tool to study theory of other minds. Paper presented at the Society for Neuroscience, New Orleans, LA, USA.
- Anderson, J. M., Gilmore, R., Roper, S., Crosson, B., Bauer, R. M., Nadeau, S., ... Heilman, K. M. (1999). Conduction aphasia and the arcuate fasciculus: A reexamination of the Wernicke-Geschwind model. *Brain Lang*, 70(1), 1-12. doi:10.1006/brln.1999.2135
- Avants, B. B., Epstein, C. L., Grossman, M., & Gee, J. C. (2008). Symmetric diffeomorphic image registration with cross-correlation: evaluating automated labeling of elderly and neurodegenerative brain. *Med Image Anal*, 12(1), 26-41. doi:10.1016/j.media.2007.06.004
- Aziz-Zadeh, L., Koski, L., Zaidel, E., Mazziotta, J., & Iacoboni, M. (2006). Lateralization of the human mirror neuron system. J Neurosci, 26(11), 2964-2970. doi:10.1523/jneurosci.2921-05.2006
- Aziz-Zadeh, L., Wilson, S. M., Rizzolatti, G., & Iacoboni, M. (2006). Congruent embodied representations for visually presented actions and linguistic phrases describing actions. *Curr Biol*, 16(18), 1818-1823. doi:10.1016/j.cub.2006.07.060
- Baggio, H. C., Sala-Llonch, R., Segura, B., Marti, M. J., Valldeoriola, F., Compta, Y., ... Junqué, C. (2014). Functional brain networks and cognitive deficits in Parkinson's disease. *Hum Brain Mapp*, 35(9), 4620-4634.
- Baldo, J. V., Paulraj, S. R., Curran, B. C., & Dronkers, N. F. (2015). Impaired reasoning and problem-solving in individuals with language impairment due to aphasia or language delay. *Front Psychol*, 6, 1523. doi:10.3389/fpsyg.2015.01523
- Barsalou, L. W. (2008). Grounded cognition. *Annu Rev Psychol*, 59, 617-645. doi:10.1146/annurev.psych.59.103006.093639
- Basso, A. (1992). Prognostic factors in aphasia. Aphasiology, 6(4), 337-348. doi:10.1080/02687039208248605
- Basso, A. (2003). Aphasia and its therapy. New York, NY: Oxford University Press.
- Bement, L., Wallber, J., DeFilippo, C., Bochner, J., & Garrison, W. (1988). A new protocol for assessing viseme perception in sentence context: the lipreading discrimination test. *Ear Hear*, *9*(1), 33-40.
- Benton, A. L. (1964). Contributions to aphasia before Broca. Cortex, 1(3), 314-327.
- Benton, A. L., & Joynt, R. J. (1963). Three pioneers in the study of aphasia. J Hist Med Allied Sci, 18, 381-384.
- Bernard, D. (1889). De l'aphasie et de ses diverses formes: Progrès médical.
- Bhogal, S. K., Teasell, R., & Speechley, M. (2003). Intensity of aphasia therapy, impact on recovery. *Stroke*, *34*(4), 987-993.
- Binder, J. R., Rao, S. M., Hammeke, T. A., Yetkin, F. Z., Jesmanowicz, A., Bandettini, P. A., . . . et al. (1994). Functional magnetic resonance imaging of human auditory cortex. *Ann Neurol*, 35(6), 662-672. doi:10.1002/ana.410350606
- Binkofski, F., Buccino, G., Posse, S., Seitz, R. J., Rizzolatti, G., & Freund, H. (1999). A fronto-parietal circuit for object manipulation in man: evidence from an fMRI-study. *Eur J Neurosci, 11*(9), 3276-3286.
- Blondel, V. D., Guillaume, J.-L., Lambiotte, R., & Lefebvre, E. (2008). Fast unfolding of communities in large networks. *J Stat Mech Theor Exp*, 2008(10), P10008.
- Blumstein, S. E. (1997). A perspective on the neurobiology of language. *Brain Lang*, 60(3), 335-346. doi:10.1006/brln.1997.1796
- Boyle, M., & Coelho, C. A. (1995). Application of Semantic Feature Analysis as a Treatment for Aphasic Dysnomia. *Am J Speech Lang Pathol*, 4(4), 94-98. doi:10.1044/1058-0360.0404.94
- Brady, M. C., Kelly, H., Godwin, J., & Enderby, P. (2012). Speech and language therapy for aphasia following stroke. *Cochrane Database Syst Rev, 5*, CD000425. doi:10.1002/14651858.CD000425.pub3 [doi]
- Brass, M., Bekkering, H., & Prinz, W. (2001). Movement observation affects movement execution in a simple response task. *Acta Psychol (Amst), 106*(1-2), 3-22.

- Brass, M., Bekkering, H., Wohlschlager, A., & Prinz, W. (2000). Compatibility between observed and executed finger movements: comparing symbolic, spatial, and imitative cues. *Brain Cogn*, 44(2), 124-143. doi:10.1006/brcg.2000.1225
- Brier, M. R., Thomas, J. B., Fagan, A. M., Hassenstab, J., Holtzman, D. M., Benzinger, T. L., ... Ances, B. M. (2014). Functional connectivity and graph theory in preclinical Alzheimer's disease. *Neurobiol Aging*, 35(4), 757-768. doi:10.1016/j.neurobiolaging.2013.10.081
- Broca, P. P. (1861). Nouvelle observation d'aphémie produite par une lésion de la moitié postérieure des deuxième et troisième circonvolutions frontales. *Bulletin de la Société Anatomique de Paris, 36*, 398-407.
- Buccino, G., Binkofski, F., Fink, G. R., Fadiga, L., Fogassi, L., Gallese, V., . . . Freund, H. J. (2001). Action observation activates premotor and parietal areas in a somatotopic manner: an fMRI study. *Eur J Neurosci,* 13(2), 400-404.
- Buccino, G., Lui, F., Canessa, N., Patteri, I., Lagravinese, G., Benuzzi, F., . . . Rizzolatti, G. (2004). Neural circuits involved in the recognition of actions performed by nonconspecifics: an FMRI study. *J Cogn Neurosci*, *16*(1), 114-126. doi:10.1162/089892904322755601
- Buchsbaum, B. R., Baldo, J., Okada, K., Berman, K. F., Dronkers, N., D'Esposito, M., & Hickok, G. (2011). Conduction aphasia, sensory-motor integration, and phonological short-term memory - an aggregate analysis of lesion and fMRI data. *Brain Lang*, 119(3), 119-128. doi:10.1016/j.bandl.2010.12.001
- Bullmore, E., & Sporns, O. (2009). Complex brain networks: graph theoretical analysis of structural and functional systems. *Nat Rev Neurosci, 10*(3), 186-198. doi:10.1038/nrn2575
- Calhoun, V. D. (2004). Group ICA of fMRI toolbox (GIFT). Online at http://icatb.sourceforge.net.
- Calhoun, V. D., Miller, R., Pearlson, G., & Adali, T. (2014). The chronnectome: time-varying connectivity networks as the next frontier in fMRI data discovery. *Neuron*, 84(2), 262-274. doi:10.1016/j.neuron.2014.10.015
- Carpenter, J., & Cherney, L. R. (2016). Increasing aphasia treatment intensity in an acute inpatient rehabilitation program: A feasibility study. *Aphasiology*, *30*(5), 542-565. doi:10.1080/02687038.2015.1023695
- Chan, M. Y., Park, D. C., Savalia, N. K., Petersen, S. E., & Wig, G. S. (2014). Decreased segregation of brain systems across the healthy adult lifespan. *Proc Natl Acad Sci U S A*, 111(46), E4997-5006. doi:10.1073/pnas.1415122111
- Chang, T. Y., Huang, K. L., Ho, M. Y., Ho, P. S., Chang, C. H., Liu, C. H., . . . Liu, H. L. (2016). Graph theoretical analysis of functional networks and its relationship to cognitive decline in patients with carotid stenosis. J Cereb Blood Flow Metab, 36(4), 808-818. doi:10.1177/0271678x15608390
- Cherney, L. R. (2004). Aphasia, alexia, and oral reading. *Top Stroke Rehabil, 11*(1), 22-36. doi:10.1310/vupx-wdx7-j1eu-00tb
- Cherney, L. R., Kaye, R. C., Lee, J. B., & van Vuuren, S. (2015). Impact of Personal Relevance on Acquisition and Generalization of Script Training for Aphasia: A Preliminary Analysis. Am J Speech Lang Pathol, 24(4), S913-922. doi:10.1044/2015_ajslp-14-0162
- Cochin, S., Barthelemy, C., Roux, S., & Martineau, J. (1999). Observation and execution of movement: similarities demonstrated by quantified electroencephalography. *Eur J Neurosci, 11*(5), 1839-1842.
- Code, C. (2010). Aphasia. In J. S. Damico, N. Müller, & M. J. Ball (Eds.), *The Handbook of Language and Speech Disorders* (pp. 317-336). Oxford, UK: Wiley-Blackwell.
- Cohen-Séat, G., Gastaut, H., Faure, J., & Heuyer, G. (1954). Etudes expérimentales de l'activité nerveuse pendant la projection cinématographique. *Rev. Int. Filmologie, 5*, 7-64.
- Coltheart, M. (1981). The MRC psycholinguistic database. Q J Exp Psychol, 33(4), 497-505.
- Courage, M. L., Edison, S. C., & Howe, M. L. (2004). Variability in the early development of visual selfrecognition. *Infant Behav Dev*, 27(4), 509-532.
- Cox, R. W. (1996). AFNI: software for analysis and visualization of functional magnetic resonance neuroimages. *Comput Biomed Res*, 29(3), 162-173.
- Craighero, L., Bello, A., Fadiga, L., & Rizzolatti, G. (2002). Hand action preparation influences the responses to hand pictures. *Neuropsychologia*, 40(5), 492-502.
- Cubelli, R., & Montagna, C. (1994). A reappraisal of the controversy of Dax and Broca. *J Hist Neurosci*, 3(4), 215-226.
- Damasio, A. R., & Geschwind, N. (1984). The neural basis of language. *Annu Rev Neurosci*, 7, 127-147. doi:10.1146/annurev.ne.07.030184.001015
- Damoiseaux, J. S., Rombouts, S. A., Barkhof, F., Scheltens, P., Stam, C. J., Smith, S. M., & Beckmann, C. F. (2006). Consistent resting-state networks across healthy subjects. *Proc Natl Acad Sci U S A*, 103(37), 13848-13853. doi:10.1073/pnas.0601417103
- de Aguiar, V., Bastiaanse, R., Capasso, R., Gandolfi, M., Smania, N., Rossi, G., & Miceli, G. (2015). Can tDCS

enhance item-specific effects and generalization after linguistically motivated aphasia therapy for verbs? *Front Behav Neurosci*, *9*, 190. doi:10.3389/fnbeh.2015.00190

- Decety, J., & Grezes, J. (1999). Neural mechanisms subserving the perception of human actions. *Trends Cogn Sci*, 3(5), 172-178.
- Decety, J., Grezes, J., Costes, N., Perani, D., Jeannerod, M., Procyk, E., ... Fazio, F. (1997). Brain activity during observation of actions. Influence of action content and subject's strategy. *Brain*, 120 (Pt 10), 1763-1777.
- Dechene, L., Tousignant, M., Boissy, P., Macoir, J., Heroux, S., Hamel, M., ... Page, C. (2011). Simulated in-home teletreatment for anomia. *Int J Telerehabil*, 3(2), 3-10. doi:10.5195/ijt.2011.6075
- di Pellegrino, G., Fadiga, L., Fogassi, L., Gallese, V., & Rizzolatti, G. (1992). Understanding motor events: a neurophysiological study. *Exp Brain Res*, 91(1), 176-180.
- Di, X., Gohel, S., Kim, E. H., & Biswal, B. B. (2013). Task vs. rest-different network configurations between the coactivation and the resting-state brain networks. *Front Hum Neurosci*, 7, 493. doi:10.3389/fnhum.2013.00493
- Dronkers, N. F., Plaisant, O., Iba-Zizen, M. T., & Cabanis, E. A. (2007). Paul Broca's historic cases: high resolution MR imaging of the brains of Leborgne and Lelong. *Brain*, 130(5), 1432-1441.
- Duchek, J. M., Balota, D. A., Tse, C.-S., Holtzman, D. M., Fagan, A. M., & Goate, A. M. (2009). The utility of intraindividual variability in selective attention tasks as an early marker for Alzheimer's disease. *Neuropsychology*, 23(6), 746.
- Duffy, J. R. (1995). *Motor speech disorders: Substrates, differential diagnosis, and management* (1st ed.). Philadelphia, PA: Mosby.
- Duncan, E. S., Schmah, T., & Small, S. L. (2016). Performance Variability as a Predictor of Response to Aphasia Treatment. *Neurorehabil Neural Repair*. doi:10.1177/1545968316642522
- Duncan, E. S., & Small, S. L. (2015). Imitation-based aphasia therapy. In G. S. Hickok & S. L. Small (Eds.), The Neurobiology of Language (pp. 1055-1065): Elsevier Health Sciences.
- Duncan, E. S., & Small, S. L. (under review). Changes in dynamic resting state network connectivity following aphasia therapy.
- Déjerine, J. J., & Déjerine-Klumpke, A. (1895). Anatomie des centres nerveux. Paris: Rueff.
- Eling, P., & Whitaker, H. (2009). History of aphasia: From brain to language. *Handbook of clinical neurology*, 95, 571-582.
- Ellis, C., Dismuke, C., & Edwards, K. K. (2010). Longitudinal trends in aphasia in the United States. *NeuroRehabilitation*, 27(4), 327-333. doi:10.3233/nre-2010-0616
- Elman, J. L. (1993). Learning and development in neural networks: The importance of starting small. *Cognition*, 48(1), 71-99.
- Engelter, S. T., Gostynski, M., Papa, S., Frei, M., Born, C., Ajdacic-Gross, V., . . . Lyrer, P. A. (2006). Epidemiology of aphasia attributable to first ischemic stroke: incidence, severity, fluency, etiology, and thrombolysis. *Stroke*, 37(6), 1379-1384. doi:10.1161/01.STR.0000221815.64093.8c
- Epstein, C. M. (1998). Transcranial magnetic stimulation: language function. J Clin Neurophysiol, 15(4), 325-332.
- Erickson, R. J., Goldinger, S. D., & LaPointe, L. L. (1996). Auditory vigilance in aphasic individuals: Detecting nonlinguistic stimuli with full or divided attention. *Brain Cogn*, *30*(2), 244-253.
- Ertelt, D., Small, S., Solodkin, A., Dettmers, C., McNamara, A., Binkofski, F., & Buccino, G. (2007). Action observation has a positive impact on rehabilitation of motor deficits after stroke. *Neuroimage*, 36 Suppl 2, T164-173. doi:10.1016/j.neuroimage.2007.03.043
- Fadiga, L., Craighero, L., Buccino, G., & Rizzolatti, G. (2002). Speech listening specifically modulates the excitability of tongue muscles: a TMS study. *Eur J Neurosci, 15*(2), 399-402.
- Fadiga, L., Fogassi, L., Pavesi, G., & Rizzolatti, G. (1995). Motor facilitation during action observation: a magnetic stimulation study. J Neurophysiol, 73(6), 2608-2611.
- Fagg, A. H., & Arbib, M. A. (1998). Modeling parietal-premotor interactions in primate control of grasping. Neural Netw, 11(7-8), 1277-1303.
- Fair, D. A., Dosenbach, N. U., Church, J. A., Cohen, A. L., Brahmbhatt, S., Miezin, F. M., . . . Schlaggar, B. L. (2007). Development of distinct control networks through segregation and integration. *Proc Natl Acad Sci* US A, 104(33), 13507-13512.
- Falcon, M. I., Riley, J. D., Jirsa, V., McIntosh, A. R., Shereen, A., Chen, E. E., & Solodkin, A. (2015). The Virtual Brain: Modeling Biological Correlates of Recovery after Chronic Stroke. *Front Neurol*, 6. doi:10.3389/fneur.2015.00228
- Ferguson, A. (1999). Clinical Forum Learning in aphasia therapy: It's not so much what you do, but how you do it! *Aphasiology*, *13*(2), 125-150.

- Ferrari, P. F., Gallese, V., Rizzolatti, G., & Fogassi, L. (2003). Mirror neurons responding to the observation of ingestive and communicative mouth actions in the monkey ventral premotor cortex. *Eur J Neurosci*, 17(8), 1703-1714.
- Fillingham, J., Sage, K., & Lambon Ralph, M. (2005). Further explorations and an overview of errorless and errorful therapy for aphasic word-finding difficulties: The number of naming attempts during therapy affects outcome. *Aphasiology*, 19(7), 597–614. doi:http://dx.doi.org/10.1080/02687030544000272
- Finger, S. (2000). Minds behind the brain: A history of the pioneers and their discoveries: Oxford University Press.
- Fischl, B. (2012), FreeSurfer, Neuroimage, 62(2), 774-781, doi:10.1016/j.neuroimage.2012.01.021
- Fogassi, L., Gallese, V., Fadiga, L., & Rizzolatti, G. (1998). *Neurons responding to the sight of goal-directed hand/arm actions in the parietal area PF (7b) of the macaque monkey.* Paper presented at the Society of Neuroscience Abstracts.
- Fowler, O. S., & Fowler, L. N. (1859). New Illustrated Self-instructor in Phrenology and Physiology: With Over One Hundred Engravings: Together with the Chart and Character of... as Marked by. New York: Fowler & Wells.
- Freud, S. (1891). Zur Auffassung der Aphasie. Eine kritische Studie. Leipzig/Vienna: Deuticke.
- Fridriksson, J., Baker, J. M., Whiteside, J., Eoute, D., Jr., Moser, D., Vesselinov, R., & Rorden, C. (2009). Treating visual speech perception to improve speech production in nonfluent aphasia. *Stroke*, 40(3), 853-858. doi:10.1161/strokeaha.108.532499
- Fridriksson, J., Hubbard, H. I., Hudspeth, S. G., Holland, A. L., Bonilha, L., Fromm, D., & Rorden, C. (2012). Speech entrainment enables patients with Broca's aphasia to produce fluent speech. *Brain*, 135(Pt 12), 3815-3829. doi:10.1093/brain/aws301
- Fridriksson, J., Kjartansson, O., Morgan, P. S., Hjaltason, H., Magnusdottir, S., Bonilha, L., & Rorden, C. (2010). Impaired speech repetition and left parietal lobe damage. *J Neurosci*, 30(33), 11057-11061. doi:10.1523/jneurosci.1120-10.2010
- Friederici, A. D. (2009). Pathways to language: fiber tracts in the human brain. *Trends Cogn Sci, 13*(4), 175-181. doi:10.1016/j.tics.2009.01.001
- Friedman, J., Hastie, T., & Tibshirani, R. (2008). Sparse inverse covariance estimation with the graphical lasso. *Biostatistics*, 9(3), 432-441. doi:10.1093/biostatistics/kxm045
- Friedman, J., Hastie, T., & Tibshirani, R. (2014). glasso: Graphical lasso-estimation of Gaussian graphical models. *R* package version, 1.
- Gall, F. J. (1825). Sur les fonctions du cerveau et sur celles de chacune de ses parties (Vol. 6). Paris: J.B. Bailliére.
- Gallese, V., Fadiga, L., Fogassi, L., & Rizzolatti, G. (1996). Action recognition in the premotor cortex. *Brain, 119 (Pt 2)*, 593-609.
- Gamaldo, A. A., An, Y., Allaire, J. C., Kitner–Triolo, M. H., & Zonderman, A. B. (2012). Variability in performance: identifying early signs of future cognitive impairment. *Neuropsychology*, *26*(4), 534.
- Gamboa, O. L., Tagliazucchi, E., von Wegner, F., Jurcoane, A., Wahl, M., Laufs, H., & Ziemann, U. (2014). Working memory performance of early MS patients correlates inversely with modularity increases in resting state functional connectivity networks. *Neuroimage*, 94, 385-395. doi:10.1016/j.neuroimage.2013.12.008
- Garrett, D. D., Macdonald, S. W., & Craik, F. I. (2012). Intraindividual reaction time variability is malleable: feedback- and education-related reductions in variability with age. *Front Hum Neurosci*, 6, 101. doi:10.3389/fnhum.2012.00101
- Geschwind, N. (1965). Disconnexion syndromes in animals and man. Brain, 88(3), 585-585.
- Geschwind, N. (1970). The Organization of Language and the Brain. Science, 170, 940-944.
- Girvan, M., & Newman, M. E. (2002). Community structure in social and biological networks. *Proc Natl Acad Sci* US A, 99(12), 7821-7826.
- Gleichgerrcht, E., Kocher, M., Nesland, T., Rorden, C., Fridriksson, J., & Bonilha, L. (2015). Preservation of structural brain network hubs is associated with less severe post-stroke aphasia. *Restor Neurol Neurosci*(Preprint), 1-11.
- Glenberg, A. M., & Kaschak, M. P. (2002). Grounding language in action. Psychon Bull Rev, 9(3), 558-565.
- Goldin-Meadow, S. (1999). The role of gesture in communication and thinking. Trends Cogn Sci, 3(11), 419-429.
- Goodale, M. A. (1993). Visual pathways supporting perception and action in the primate cerebral cortex. *Curr Opin Neurobiol*, *3*(4), 578-585.
- Goodglass, H., & Kaplan, E. (1983). Boston diagnostic aphasia examination booklet: Lea & Febiger.
- Gorno-Tempini, M. L., Hillis, A. E., Weintraub, S., Kertesz, A., Mendez, M., Cappa, S. F., ... Grossman, M. (2011). Classification of primary progressive aphasia and its variants. *Neurology*, *76*(11), 1006-1014.

doi:10.1212/WNL.0b013e31821103e6

- Grafton, S. T., Arbib, M. A., Fadiga, L., & Rizzolatti, G. (1996). Localization of grasp representations in humans by positron emission tomography. 2. Observation compared with imagination. *Exp Brain Res*, 112(1), 103-111.
- Gratton, C., Nomura, E. M., Pérez, F., & D'Esposito, M. (2012). Focal brain lesions to critical locations cause widespread disruption of the modular organization of the brain. *J Cogn Neurosci*, 24(6), 1275-1285.
- Hagberg, A., Schult, D., Swart, P., Conway, D., Séguin-Charbonneau, L., Ellison, C., . . . Torrents, J. (2004). Networkx. High productivity software for complex networks. *Webová strá nka* https://networkx. *lanl.* gov/wiki.
- Hagberg, A., Schult, D. A., & Swart, P. (2008). *Exploring network structure, dynamics, and function using NetworkX.* Paper presented at the Proceedings of the 7th Python in Science Conferences (SciPy 2008).
- Hagmann, P., Cammoun, L., Gigandet, X., Meuli, R., Honey, C. J., Wedeen, V. J., & Sporns, O. (2008). Mapping the structural core of human cerebral cortex. *PLoS Biol*, 6(7), e159.
- Hari, R., Forss, N., Avikainen, S., Kirveskari, E., Salenius, S., & Rizzolatti, G. (1998). Activation of human primary motor cortex during action observation: a neuromagnetic study. *Proc Natl Acad Sci U S A*, 95(25), 15061-15065.
- Hauk, O., Johnsrude, I., & Pulvermuller, F. (2004). Somatotopic representation of action words in human motor and premotor cortex. *Neuron*, 41(2), 301-307.
- Helm-Estabrooks, N. (1981). Helm elicited language program for syntax stimulation. Austin, TX: Pro-Ed.
- Helm-Estabrooks, N., Fitzpatrick, P. M., & Barresi, B. (1982). Visual action therapy for global aphasia. J Speech Hear Disord, 47(4), 385-389.
- Helm-Estabrooks, N., Morgan, A. R., & Nicholas, M. (1989). *Melodic intonation therapy*: Riverside Publishing Company.
- Hickok, G., & Poeppel, D. (2004). Dorsal and ventral streams: a framework for understanding aspects of the functional anatomy of language. *Cognition*, 92(1-2), 67-99.
- Hickok, G., & Poeppel, D. (2007). The cortical organization of speech processing. *Nat Rev Neurosci, 8*(5), 393-402. doi:10.1038/nrn2113
- Hillis, A. E. (2007). Aphasia: progress in the last quarter of a century. *Neurology*, 69(2), 200-213. doi:10.1212/01.wnl.0000265600.69385.6f
- Himberg, J., Hyvarinen, A., & Esposito, F. (2004). Validating the independent components of neuroimaging time series via clustering and visualization. *Neuroimage*, 22(3), 1214-1222. doi:10.1016/j.neuroimage.2004.03.027
- Huber, R., Ghilardi, M. F., Massimini, M., & Tononi, G. (2004). Local sleep and learning. *Nature*, 430(6995), 78-81. doi:10.1038/nature02663
- Huber, W., Poeck, K., & Willmes, K. (1984). The Aachen Aphasia Test. Adv Neurol, 42, 291-303.
- Hultsch, D. F., MacDonald, S. W., Hunter, M. A., Levy-Bencheton, J., & Strauss, E. (2000). Intraindividual variability in cognitive performance in older adults: comparison of adults with mild dementia, adults with arthritis, and healthy adults. *Neuropsychology*, 14(4), 588.
- Hutchison, R. M., Womelsdorf, T., Allen, E. A., Bandettini, P. A., Calhoun, V. D., Corbetta, M., . . . Chang, C. (2013). Dynamic functional connectivity: promise, issues, and interpretations. *Neuroimage*, 80, 360-378. doi:10.1016/j.neuroimage.2013.05.079
- Iacoboni, M., Molnar-Szakacs, I., Gallese, V., Buccino, G., Mazziotta, J. C., & Rizzolatti, G. (2005). Grasping the intentions of others with one's own mirror neuron system. *PLoS Biol*, 3(3), e79. doi:10.1371/journal.pbio.0030079
- Iacoboni, M., Woods, R. P., Brass, M., Bekkering, H., Mazziotta, J. C., & Rizzolatti, G. (1999). Cortical mechanisms of human imitation. *Science*, 286(5449), 2526-2528.
- Ingham, R. J. (1980). Modification of maintenance and generalization during stuttering treatment. *J Speech Hear Res*, 23(4), 732-745.
- Jeannerod, M. (1994). The representing brain: Neural correlates of motor intention and imagery. *Behav Brain Sci, 17*(02), 187-202.
- Jirsa, V., Sporns, O., Breakspear, M., Deco, G., & McIntosh, A. R. (2010). Towards the virtual brain: network modeling of the intact and the damaged brain. *Arch Ital Biol*, *148*(3), 189-205.
- Kamen, G. (2004). Reliability of motor-evoked potentials during resting and active contraction conditions. *Med Sci* Sports Exerc, 36(9), 1574-1579.
- Kasselimis, D. S., Simos, P. G., Economou, A., Peppas, C., Evdokimidis, I., & Potagas, C. (2013). Are memory deficits dependent on the presence of aphasia in left brain damaged patients? *Neuropsychologia*, 51(9),

1773-1776. doi:10.1016/j.neuropsychologia.2013.06.003

- Kertesz, A. (1982). The Western Aphasia Battery. New York: Grune and Stratton.
- Kertesz, A. (2006). Western Aphasia Battery (Revised) PsychCorp. San Antonio, Tx.
- Keysers, C., & Perrett, D. I. (2004). Demystifying social cognition: a Hebbian perspective. *Trends Cogn Sci*, 8(11), 501-507. doi:10.1016/j.tics.2004.09.005
- Kilner, J. M., Marchant, J. L., & Frith, C. D. (2006). Modulation of the mirror system by social relevance. Soc Cogn Affect Neurosci, 1(2), 143-148. doi:10.1093/scan/nsl017
- Kiran, S., Des Roches, C., Villard, S., & Tripodis, Y. (2015). The effect of a sentence comprehension treatment on discourse comprehension in aphasia. *Aphasiology*, 29(11), 1289-1311. doi:10.1080/02687038.2014.997182
- Kiran, S., & Thompson, C. K. (2003). The Role of Semantic Complexity in Treatment of Naming DeficitsTraining Semantic Categories in Fluent Aphasia by Controlling Exemplar Typicality. J Speech Language Hear Res, 46(3), 608-622.
- Klatzky, R. L., Pellegrino, J. W., McCloskey, B. P., & Doherty, S. (1989). Can you squeeze a tomato? The role of motor representations in semantic sensibility judgments. *J Mem Lang*, 28(1), 56-77.
- Kohler, E., Keysers, C., Umilta, M. A., Fogassi, L., Gallese, V., & Rizzolatti, G. (2002). Hearing sounds, understanding actions: action representation in mirror neurons. *Science*, 297(5582), 846-848. doi:10.1126/science.1070311
- Kucera, H., & Francis, W. N. (1967). Computational Analysis of Present-Day American English. Providence, RI: Brown University Press.
- Kuhl, P., & Rivera-Gaxiola, M. (2008). Neural substrates of language acquisition. *Annu Rev Neurosci, 31*, 511-534. doi:10.1146/annurev.neuro.30.051606.094321
- Laska, A. C., Hellblom, A., Murray, V., Kahan, T., & Von Arbin, M. (2001). Aphasia in acute stroke and relation to outcome. *J Intern Med*, 249(5), 413-422.
- Lazar, R. M., Minzer, B., Antoniello, D., Festa, J. R., Krakauer, J. W., & Marshall, R. S. (2010). Improvement in aphasia scores after stroke is well predicted by initial severity. *Stroke*, *41*(7), 1485-1488.
- Lazar, R. M., Speizer, A. E., Festa, J. R., Krakauer, J. W., & Marshall, R. S. (2008). Variability in language recovery after first-time stroke. *J Neurol Neurosurg Psychiatry*, 79(5), 530-534. doi:10.1136/jnnp.2007.122457
- Lee, B., Moon, H. I., Lim, S. H., Cho, H., Choi, H., & Pyun, S. B. (2016). Recovery of language function in Korean-Japanese crossed bilingual aphasia following right basal ganglia hemorrhage. *Neurocase*, 1-6. doi:10.1080/13554794.2016.1141966
- Lee, J., Fowler, R., Rodney, D., Cherney, L., & Small, S. L. (2010). IMITATE: An intensive computer-based treatment for aphasia based on action observation and imitation. *Aphasiology*, 24(4), 449-465.
- Lee, M. H., Hacker, C. D., Snyder, A. Z., Corbetta, M., Zhang, D., Leuthardt, E. C., & Shimony, J. S. (2012). Clustering of resting state networks. *PLoS One*, 7(7), e40370. doi:10.1371/journal.pone.0040370
- Lepage, J. F., & Theoret, H. (2007). The mirror neuron system: grasping others' actions from birth? *Dev Sci*, 10(5), 513-523. doi:10.1111/j.1467-7687.2007.00631.x
- Li, S.-C., Lindenberger, U., Hommel, B., Aschersleben, G., Prinz, W., & Baltes, P. B. (2004). Transformations in the couplings among intellectual abilities and constituent cognitive processes across the life span. *Psychol Sci*, 15(3), 155-163.
- Li, S.-C., Lindenberger, U., & Sikström, S. (2001). Aging cognition: from neuromodulation to representation. *Trends Cogn Sci*, 5(11), 479-486.
- Lichtheim, L. (1885). On Aphasia. Brain, 7, 433-484.
- Linebaugh, C. W., Shisler, R. J., & Lehner , L. (2005). Cueing hierarchy and word retrieval: a therapy program. *Aphasiology, 19*(1), 77-92. doi:http://dx.doi.org/10.1080/02687030444000363
- Love, R. J., & Webb, W. G. (1977). The efficacy of cueing techniques in Broca's aphasia. J Speech Hear Disord, 42(2), 170-178.
- Luzzatti, C. (2002). Johann August Philipp Gesner (1738–1801). A review of his essay "The language amnesia" in the bicentennial anniversary of his death. *J Hist Neurosci*, 11(1), 29-34.
- Luzzatti, C., & Whitaker, H. (1996). Johannes Schenck and Johannes Jakob Wepfer: clinical and anatomical observations in the prehistory of neurolinguistics and neuropsychology. *J Neuroling*, 9(3), 157-164.
- Lövdén, M., Li, S.-C., Shing, Y. L., & Lindenberger, U. (2007). Within-person trial-to-trial variability precedes and predicts cognitive decline in old and very old age: Longitudinal data from the Berlin Aging Study. *Neuropsychologia*, 45(12), 2827-2838.
- Ma, Q., Huang, B., Wang, J., Seger, C., Yang, W., Li, C., . . . Huang, R. (2016). Altered modular organization of intrinsic brain functional networks in patients with Parkinson's disease. *Brain Imaging Behav*.

doi:10.1007/s11682-016-9524-7

- MacDonald, S. W., Hultsch, D. F., & Dixon, R. A. (2003). Performance variability is related to change in cognition: evidence from the Victoria Longitudinal Study. *Psychol Aging*, 18(3), 510.
- MacDonald, S. W., Hultsch, D. F., & Dixon, R. A. (2008). Predicting impending death: inconsistency in speed is a selective and early marker. *Psychol Aging*, 23(3), 595.
- Marangolo, P., Bonifazi, S., Tomaiuolo, F., Craighero, L., Coccia, M., Altoe, G., . . . Cantagallo, A. (2010). Improving language without words: first evidence from aphasia. *Neuropsychologia*, 48(13), 3824-3833. doi:10.1016/j.neuropsychologia.2010.09.025
- Marangolo, P., Cipollari, S., Fiori, V., Razzano, C., & Caltagirone, C. (2012). Walking but not barking improves verb recovery: implications for action observation treatment in aphasia rehabilitation. *PLoS One*, 7(6), e38610. doi:10.1371/journal.pone.0038610
- Marcotte, K., Adrover-Roig, D., Damien, B., de Preaumont, M., Genereux, S., Hubert, M., & Ansaldo, A. I. (2012). Therapy-induced neuroplasticity in chronic aphasia. *Neuropsychologia*, 50(8), 1776-1786. doi:10.1016/j.neuropsychologia.2012.04.001
- Marcotte, K., Perlbarg, V., Marrelec, G., Benali, H., & Ansaldo, A. I. (2013). Default-mode network functional connectivity in aphasia: therapy-induced neuroplasticity. *Brain Lang*, *124*(1), 45-55.
- Marshall, P. J., Young, T., & Meltzoff, A. N. (2011). Neural correlates of action observation and execution in 14month-old infants: an event-related EEG desynchronization study. *Dev Sci*, 14(3), 474-480. doi:10.1111/j.1467-7687.2010.00991.x
- Mashal, N., Solodkin, A., Chen, E. E., Dick, A. S., & Small, S. L. (2012). A network model of observation and imitation of speech. *Front Psychol*, *3*(84).
- McIntosh, A. R., & Gonzalez-Lima, F. (1994). Structural equation modeling and its application to network analysis in functional brain imaging. *Hum Brain Mapp*, 2(1-2), 2-22.
- Meinzer, M., Darkow, R., Lindenberg, R., & Floel, A. (2016). Electrical stimulation of the motor cortex enhances treatment outcome in post-stroke aphasia. *Brain*. doi:10.1093/brain/aww002
- Milton, J., Solodkin, A., Hlustik, P., & Small, S. L. (2007). The mind of expert motor performance is cool and focused. *Neuroimage*, *35*(2), 804-813. doi:10.1016/j.neuroimage.2007.01.003
- Minagar, A., Ragheb, J., & Kelley, R. E. (2003). The Edwin Smith surgical papyrus: description and analysis of the earliest case of aphasia. *J Med Biogr*, *11*(2), 114-117.
- Mishkin, M., Ungerleider, L. G., & Macko, K. A. (1983). Object vision and spatial vision: two cortical pathways. *Trends Neurosci*, 6, 414-417.
- Mottonen, R., & Watkins, K. E. (2012). Using TMS to study the role of the articulatory motor system in speech perception. *Aphasiology*, 26(9), 1103-1118. doi:10.1080/02687038.2011.619515
- Mozaffarian, D., Benjamin, E. J., Go, A. S., Arnett, D. K., Blaha, M. J., Cushman, M., . . . Turner, M. B. (2016). Heart Disease and Stroke Statistics-2016 Update: A Report From the American Heart Association. *Circulation*, 133(4), e38-e360. doi:10.1161/cir.00000000000350
- Mukamel, R., Ekstrom, A. D., Kaplan, J., Iacoboni, M., & Fried, I. (2010). Single-neuron responses in humans during execution and observation of actions. *Curr Biol*, 20(8), 750-756. doi:10.1016/j.cub.2010.02.045
- Mummery, C. J., Patterson, K., Price, C. J., Ashburner, J., Frackowiak, R. S., & Hodges, J. R. (2000). A voxel-based morphometry study of semantic dementia: relationship between temporal lobe atrophy and semantic memory. *Ann Neurol*, 47(1), 36-45.
- Murphy, T. H., & Corbett, D. (2009). Plasticity during stroke recovery: from synapse to behaviour. *Nature Reviews Neuroscience*, *10*(12), 861-872.
- Murray, L. L. (2012). Attention and other cognitive deficits in aphasia: presence and relation to language and communication measures. *Am J Speech Lang Pathol*, 21(2), S51-64. doi:10.1044/1058-0360(2012/11-0067)
- Muthukumaraswamy, S. D., & Johnson, B. W. (2004). Changes in rolandic mu rhythm during observation of a precision grip. *Psychophysiology*, *41*(1), 152-156. doi:10.1046/j.1469-8986.2003.00129.x
- Muthuraman, M., Fleischer, V., Kolber, P., Luessi, F., Zipp, F., & Groppa, S. (2016). Structural Brain Network Characteristics Can Differentiate CIS from Early RRMS. *Front Neurosci, 10*, 14. doi:10.3389/fnins.2016.00014
- Myung, J. Y., Blumstein, S. E., & Sedivy, J. C. (2006). Playing on the typewriter, typing on the piano: manipulation knowledge of objects. *Cognition*, *98*(3), 223-243. doi:10.1016/j.cognition.2004.11.010
- Nesselroade, J. R. (2002). Elaborating the differential in differential psychology. *Multivariate Behav Res, 37*(4), 543-561.
- Nesselroade, J. R., & Salthouse, T. A. (2004). Methodological and theoretical implications of intraindividual

variability in perceptual-motor performance. J Gerontol B Psychol Sci Soc Sci, 59(2), P49-P55.

- Newman, M. E., & Girvan, M. (2004). Finding and evaluating community structure in networks. *Phys Rev E Stat Nonlin Soft Matter Phys*, 69(2), 026113.
- Nicholas, L. E., & Brookshire, R. H. (1993). A system for quantifying the informativeness and efficiency of the connected speech of adults with aphasia. *J Speech Lang Hear Res, 36*(2), 338-350.
- Nickels, L. (2002). Therapy for naming disorders: Revisiting, revising, and reviewing. *Aphasiology*, 16(10-11), 935-979.
- NIDCD. (2010). Statistics and Voice, Speech, and Language [NIDCD Health Information]. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/
- Nishitani, N., & Hari, R. (2002). Viewing lip forms: cortical dynamics. Neuron, 36(6), 1211-1220.
- Nusbaum, H. C., Pisoni, D. B., & Davis, C. K. (1984). Sizing up the Hoosier mental lexicon: Measuring the familiarity of 20,000 words. *Research on speech perception progress report*, 10(10), 357-376.
- Ojemann, G., Ojemann, J., Lettich, E., & Berger, M. (1989). Cortical language localization in left, dominant hemisphere: an electrical stimulation mapping investigation in 117 patients. *J Neurosurg*, 71(3), 316-326.
- Okada, K., & Hickok, G. (2006). Left posterior auditory-related cortices participate both in speech perception and speech production: Neural overlap revealed by fMRI. *Brain Lang*, *98*(1), 112-117. doi:10.1016/j.bandl.2006.04.006
- Onoda, K., & Yamaguchi, S. (2013). Small-worldness and modularity of the resting-state functional brain network decrease with aging. *Neurosci Lett, 556*, 104-108. doi:10.1016/j.neulet.2013.10.023
- Orjada, S., & Beeson, P. (2005). Concurrent treatment for reading and spelling in aphasia. *Aphasiology*, 19(3-5), 341-351.
- Owens, E., & Blazek, B. (1985). Visemes observed by hearing-impaired and normal-hearing adult viewers. J Speech Lang Hear Res, 28(3), 381-393.
- Palyo, W. J., Cooke, T. P., Schuler, A. L., & Apolloni, T. (1979). Modifying echolalic speech in preschool children: training and generalization. *Am J Ment Defic*, 83(5), 480-489.
- Papathanassiou, D., Etard, O., Mellet, E., Zago, L., Mazoyer, B., & Tzourio-Mazoyer, N. (2000). A common language network for comprehension and production: a contribution to the definition of language epicenters with PET. *Neuroimage*, 11(4), 347-357. doi:10.1006/nimg.2000.0546
- Parker, G. J., Luzzi, S., Alexander, D. C., Wheeler-Kingshott, C. A., Ciccarelli, O., & Lambon Ralph, M. A. (2005). Lateralization of ventral and dorsal auditory-language pathways in the human brain. *Neuroimage*, 24(3), 656-666. doi:10.1016/j.neuroimage.2004.08.047
- Pedersen, P. M., Jorgensen, H. S., Nakayama, H., Raaschou, H. O., & Olsen, T. S. (1995). Aphasia in acute stroke: incidence, determinants, and recovery. *Ann Neurol*, 38(4), 659-666.
- Petrides, M., & Pandya, D. N. (2009). Distinct parietal and temporal pathways to the homologues of Broca's area in the monkey. *PLoS Biol*, 7(8), e1000170. doi:10.1371/journal.pbio.1000170
- Pickersgill, M. J., & Lincoln, N. B. (1983). Prognostic indicators and the pattern of recovery of communication in aphasic stroke patients. *J Neurol Neurosurg Psychiatry*, 46(2), 130-139.
- Plowman, E., Hentz, B., & Ellis, C. (2012). Post-stroke aphasia prognosis: A review of patient-related and strokerelated factors. *J Eval Clin Pract*, 18(3), 689-694.
- Pring, T., Hamilton, A., Harwood, A., & Macbride, L. (1993). Generalization of naming after picture/word matching tasks: Only items appearing in therapy benefit. *Aphasiology*, 7(4), 383–394. doi:http://dx.doi.org/10.1080/02687039308249517
- Prins, R., & Bastiaanse, R. (2006). The early history of aphasiology: from the Egyptian surgeons (c. 1700 BC) to Broca (1861). *Aphasiology*, 20(8), 762-791.
- Pulvermüller, F., Neininger, B., Elbert, T., Mohr, B., Rockstroh, B., Koebbel, P., & Taub, E. (2001). Constraintinduced therapy of chronic aphasia after stroke. *Stroke*, *32*(7), 1621-1626.
- Rauschecker, J. P., & Tian, B. (2000). Mechanisms and streams for processing of "what" and "where" in auditory cortex. *Proc Natl Acad Sci U S A*, 97(22), 11800-11806. doi:10.1073/pnas.97.22.11800
- Rizzolatti, G., & Arbib, M. A. (1998). Language within our grasp. Trends Neurosci, 21(5), 188-194.
- Rizzolatti, G., Fadiga, L., Gallese, V., & Fogassi, L. (1996). Premotor cortex and the recognition of motor actions. *Brain Res Cogn Brain Res*, 3(2), 131-141.
- Rizzolatti, G., Fadiga, L., Matelli, M., Bettinardi, V., Paulesu, E., Perani, D., & Fazio, F. (1996). Localization of grasp representations in humans by PET: 1. Observation versus execution. *Exp Brain Res*, 111(2), 246-252.
- Rizzolatti, G., Fogassi, L., & Gallese, V. (2001). Neurophysiological mechanisms underlying the understanding and imitation of action. *Nat Rev Neurosci*, 2(9), 661-670. doi:10.1038/35090060
- Rombouts, S. A., Barkhof, F., Goekoop, R., Stam, C. J., & Scheltens, P. (2005). Altered resting state networks in

mild cognitive impairment and mild Alzheimer's disease: an fMRI study. *Hum Brain Mapp, 26*(4), 231-239. doi:10.1002/hbm.20160

- Rozzi, S., Calzavara, R., Belmalih, A., Borra, E., Gregoriou, G. G., Matelli, M., & Luppino, G. (2006). Cortical connections of the inferior parietal cortical convexity of the macaque monkey. *Cereb Cortex*, 16(10), 1389-1417. doi:10.1093/cercor/bhj076
- Rozzi, S., Ferrari, P. F., Bonini, L., Rizzolatti, G., & Fogassi, L. (2008). Functional organization of inferior parietal lobule convexity in the macaque monkey: electrophysiological characterization of motor, sensory and mirror responses and their correlation with cytoarchitectonic areas. *Eur J Neurosci, 28*(8), 1569-1588. doi:10.1111/j.1460-9568.2008.06395.x
- Rubinov, M., & Sporns, O. (2010). Complex network measures of brain connectivity: uses and interpretations. *Neuroimage*, 52(3), 1059-1069. doi:10.1016/j.neuroimage.2009.10.003
- Rumiati, R. I., Weiss, P. H., Tessari, A., Assmus, A., Zilles, K., Herzog, H., & Fink, G. R. (2005). Common and differential neural mechanisms supporting imitation of meaningful and meaningless actions. *J Cogn Neurosci, 17*(9), 1420-1431. doi:10.1162/0898929054985374
- Saffran, E. M., Berndt, R. S., & Schwartz, M. F. (1989). The quantitative analysis of agrammatic production: procedure and data. *Brain Lang*, *37*(3), 440-479.
- Sandberg, C. W., Bohland, J. W., & Kiran, S. (2015). Changes in functional connectivity related to direct training and generalization effects of a word finding treatment in chronic aphasia. *Brain Lang*, 150, 103-116. doi:S0093-934X(15)00187-X [pii] 10.1016/j.bandl.2015.09.002
- Sarasso, S., Määttä, S., Ferrarelli, F., Poryazova, R., Tononi, G., & Small, S. L. (2014). Plastic Changes Following Imitation-Based Speech and Language Therapy for Aphasia. *Neurorehabil Neural Repair*, 28(2), 129-138. doi:10.1177/1545968313498651
- Saur, D., Kreher, B. W., Schnell, S., Kummerer, D., Kellmeyer, P., Vry, M. S., . . . Weiller, C. (2008). Ventral and dorsal pathways for language. *Proc Natl Acad Sci U S A*, *105*(46), 18035-18040.
- Schmahmann, J. D., & Pandya, D. (2009). Fiber pathways of the brain: OUP USA.
- Seltzer, B., & Pandya, D. N. (1978). Afferent cortical connections and architectonics of the superior temporal sulcus and surrounding cortex in the rhesus monkey. *Brain Res, 149*(1), 1-24.
- Siegler, R. S. (2007). Cognitive variability. Dev Sci, 10(1), 104-109.
- Siegler, R. S., & Lemaire, P. (1997). Older and younger adults' strategy choices in multiplication: testing predictions of ASCM using the choice/no-choice method. *J Exp Psychol Gen*, *126*(1), 71.
- Sigurðardóttir, Z. G., & Sighvatsson, M. B. (2006). Operant conditioning and errorless learning procedures in the treatment of chronic aphasia. *Int J Psychol*, *41*(6), 527-540.
- Skipper, J. I., Nusbaum, H. C., & Small, S. L. (2006). Lending a helping hand to hearing: another motor theory of speech perception. In M. A. Arbib (Ed.), *Action to language via the mirror neuron system* (pp. 250-285). Cambridge: Cambridge University Press.
- Skipper, J. I., van Wassenhove, V., Nusbaum, H. C., & Small, S. L. (2007). Hearing lips and seeing voices: how cortical areas supporting speech production mediate audiovisual speech perception. *Cereb Cortex*, 17(10), 2387-2399. doi:10.1093/cercor/bhl147
- Small, S. L. (2000). The future of aphasia treatment. Brain Lang, 71(1), 227-232. doi:10.1006/brln.1999.2256
- Small, S. L. (2004). A biological model of aphasia rehabilitation: Pharmacological perspectives. *Aphasiology*, *18*(5-7), 473-492.
- Small, S. L. (2009). A biological basis for aphasia treatment: mirror neurons and observation-execution matching. Poz Stud Contemp Lin, 45(2), 313-326.
- Small, S. L., Buccino, G., & Solodkin, A. (2012). The mirror neuron system and treatment of stroke. *Dev Psychobiol*, *54*(3), 293-310.
- Small, S. L., Holland, A. L., Hart Jr, J., Forbes, M. M., & Gordon, B. (1995). Response Variability in Naming: A Computational Study. In Clinical Aphasiology Conference: Clinical Aphasiology Conference (1993 : 23rd : Sedona, AZ) / : Pro-Ed(1995), pages 25-38.
- Small, S. L., & Llano, D. A. (2009). Biological approaches to aphasia treatment. *Curr Neurol Neurosci Rep*, 9(6), 443-450.
- Small, S. L., Noll, D. C., Perfetti, C. A., Hlustik, P., Wellington, R., & Schneider, W. (1996). Localizing the lexicon for reading aloud:replication of a PET study using fMRI. *Neuroreport*, 7(4), 961-965.
- Smith, S. M., Jenkinson, M., Woolrich, M. W., Beckmann, C. F., Behrens, T. E., Johansen-Berg, H., . . . Matthews, P. M. (2004). Advances in functional and structural MR image analysis and implementation as FSL. *Neuroimage, 23 Suppl 1*, S208-219. doi:10.1016/j.neuroimage.2004.07.051
- Solodkin, A., Hasson, U., Siugzdaite, R., Schiel, M., Chen, E. E., Kotter, R., & Small, S. L. (2010). Virtual brain

transplantation (VBT): a method for accurate image registration and parcellation in large cortical stroke. *Arch Ital Biol, 148*(3), 219-241.

- Soman, S., Prasad, G., Hitchner, E., Massaband, P., Moseley, M. E., Zhou, W., & Rosen, A. C. (2016). Brain structural connectivity distinguishes patients at risk for cognitive decline after carotid interventions. *Hum Brain Mapp*, 37(6), 2185-2194.
- Song, J., Birn, R. M., Boly, M., Meier, T. B., Nair, V. A., Meyerand, M. E., & Prabhakaran, V. (2014). Age-related reorganizational changes in modularity and functional connectivity of human brain networks. *Brain Connect*, 4(9), 662-676. doi:10.1089/brain.2014.0286
- Sporns, O. (2013). Network attributes for segregation and integration in the human brain. *Curr Opin Neurobiol*, 23(2), 162-171. doi:10.1016/j.conb.2012.11.015
- Sporns, O., Honey, C. J., & Kotter, R. (2007). Identification and classification of hubs in brain networks. *PLoS One*, 2(10), e1049. doi:10.1371/journal.pone.0001049
- Stevens, A. A., Tappon, S. C., Garg, A., & Fair, D. A. (2012). Functional brain network modularity captures interand intra-individual variation in working memory capacity. *PloS one*, 7(1), e30468.
- Strafella, A. P., & Paus, T. (2000). Modulation of cortical excitability during action observation: a transcranial magnetic stimulation study. *Neuroreport*, 11(10), 2289-2292.
- Szaflarski, J. P., Ball, A. L., Grether, S., Al-fwaress, F., Griffith, N. M., Neils-Strunjas, J., . . . Reichhardt, R. (2008). Constraint-induced aphasia therapy stimulates language recovery in patients with chronic aphasia after ischemic stroke. *Med Sci Monit*, 14(5), CR243.
- Tanaka, H., Toyonaga, T., & Hashimoto, H. (2014). Functional and occupational characteristics predictive of a return to work within 18 months after stroke in Japan: implications for rehabilitation. Int Arch Occup Environ Health, 87(4), 445-453. doi:10.1007/s00420-013-0883-8
- Tesak, J., & Code, C. (2008). *Milestones in the history of aphasia: theories and protagonists*: Psychology Press.
- Tettamanti, M., Buccino, G., Saccuman, M. C., Gallese, V., Danna, M., Scifo, P., ... Perani, D. (2005). Listening to action-related sentences activates fronto-parietal motor circuits. *J Cogn Neurosci*, 17(2), 273-281. doi:10.1162/0898929053124965
- Tinaz, S., Lauro, P., Hallett, M., & Horovitz, S. G. (2016). Deficits in task-set maintenance and execution networks in Parkinson's disease. *Brain Struct Funct, 221*(3), 1413-1425. doi:10.1007/s00429-014-0981-8
- Tremblay, C., Robert, M., Pascual-Leone, A., Lepore, F., Nguyen, D. K., Carmant, L., . . . Theoret, H. (2004). Action observation and execution: intracranial recordings in a human subject. *Neurology*, *63*(5), 937-938.
- Tseng, C.-H., McNeil, M., & Milenkovic, P. (1993). An investigation of attention allocation deficits in aphasia. Brain Lang, 45(2), 276-296.
- Umilta, M. A., Kohler, E., Gallese, V., Fogassi, L., Fadiga, L., Keysers, C., & Rizzolatti, G. (2001). I know what you are doing: a neurophysiological study. *Neuron*, 31(1), 155-165.
- Van Geert, P., & Van Dijk, M. (2002). Focus on variability: New tools to study intra-individual variability in developmental data. *Infant Behav Dev*, 25(4), 340-374.
- van Hees, S., McMahon, K., Angwin, A., de Zubicaray, G., Read, S., & Copland, D. A. (2014). A functional MRI study of the relationship between naming treatment outcomes and resting state functional connectivity in post-stroke aphasia. *Hum Brain Mapp*, 35(8), 3919-3931. doi:10.1002/hbm.22448
- Vatansever, D., Menon, D. K., Manktelow, A. E., Sahakian, B. J., & Stamatakis, E. A. (2015). Default Mode Dynamics for Global Functional Integration. *J Neurosci*, 35(46), 15254-15262. doi:10.1523/jneurosci.2135-15.2015
- Wambaugh, J. L., Kalinyak-Fliszar, M. M., West, J. E., & Doyle, P. J. (1998). Effects of treatment for sound errors in apraxia of speech and aphasia. *J Speech Lang Hear Res*, 41(4), 725-743.
- Ward, N. S., Brown, M. M., Thompson, A. J., & Frackowiak, R. S. (2003). Neural correlates of motor recovery after stroke: a longitudinal fMRI study. *Brain*, 126(Pt 11), 2476-2496. doi:10.1093/brain/awg245
- Warren, J. E., Crinion, J. T., Lambon Ralph, M. A., & Wise, R. J. (2009). Anterior temporal lobe connectivity correlates with functional outcome after aphasic stroke. *Brain*, *132*(Pt 12), 3428-3442.
- Watkins, K. E., Strafella, A. P., & Paus, T. (2003). Seeing and hearing speech excites the motor system involved in speech production. *Neuropsychologia*, 41(8), 989-994.
- Wernicke, C. (1874). Der Aphasische Symptomencomplex. Breslau: Max Cohn & Weigert.
- Williams, B. R., Hultsch, D. F., Strauss, E. H., Hunter, M. A., & Tannock, R. (2005). Inconsistency in reaction time across the life span. *Neuropsychology*, 19(1), 88.
- Williams, J. H., Whiten, A., Suddendorf, T., & Perrett, D. I. (2001). Imitation, mirror neurons and autism. *Neurosci Biobehav Rev*, 25(4), 287-295.
- Willis, S. L., Tennstedt, S. L., Marsiske, M., Ball, K., Elias, J., Koepke, K. M., . . . Stoddard, A. M. (2006). Long-

term effects of cognitive training on everyday functional outcomes in older adults. *JAMA*, 296(23), 2805-2814.

- Willmes, K., & Poeck, K. (1993). To what extent can aphasic syndromes be localized? *Brain, 116 (Pt 6)*, 1527-1540.
- Wittenauer, R., & Smith, L. (2012). Update on 2004 background paper, BP 6.6 Stroke.
- Woodward, N. D., Rogers, B., & Heckers, S. (2011). Functional resting-state networks are differentially affected in schizophrenia. *Schizophr Res*, 130(1-3), 86-93. doi:10.1016/j.schres.2011.03.010
- Youmans, G., Holland, A., Muñoz, M., & Bourgeois, M. (2005). Script training and automaticity in two individuals with aphasia. *Aphasiology*, 19(3-5), 435-450.
- Zachary, W. W. (1977). An information flow model for conflict and fission in small groups. J Anth Res, 452-473.
- Zhu, D., Chang, J., Freeman, S., Tan, Z., Xiao, J., Gao, Y., & Kong, J. (2014). Changes of functional connectivity in the left frontoparietal network following aphasic stroke. *Front Behav Neurosci*, *8*, 167.

Appendix A: Artifactual ICA Components

The following components were not selected as components of interest for the analyses described in Chapters 6 and 7 due to their overlap with regions of known artifact (e.g., brainstem, ventricles, venous sinuses) and their lack of consistency with RSNs identified in the literature. Each of these twelve components is displayed on the group template (highest 5% of Z scores for mean components for the group) and at the coordinates (upper right corner) of its peak voxel (highest Z score).



Figure A.1 Component 1. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.2 Component 7. Left= coronal; middle= sagittal (left hemisphere); right= axial.



Figure A.3 Component 9. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.4 Component 10. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.5 Component 11. Left= coronal; middle= sagittal (left hemisphere); right= axial.



Figure A.6 Component 12. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.7 Component 15. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.8 Component 16. Left= coronal; middle= sagittal (right hemisphere); right= axial.


Figure A.9 Component 17. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.10 Component 18. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure A.11 Component 19. Left= coronal; middle= sagittal (left hemisphere); right= axial.



Figure A.12 Component 20. Left= coronal; middle= sagittal (left hemisphere); right= axial.

Appendix B: Overlap Of Resting State Networks with Healthy Controls

The following figures (B.1 - B.8) show the overlap between the resting state networks (RSNs) used in the analyses with participants with post-stroke aphasia, as described in Chapters 6 and 7, and RSNs from a group of healthy controls (n=27; 16 female; mean age= 38; SD=20.38; range=18-70). Healthy control data specs: Preprocessing steps and independent component analysis (ICA) were identical to those described in Sections 7.2.3.2 and 7.2.3.4, with the exception of the use of a lesion mask in the regression used to clean the time series of nuisance signals. Figures below show the eight components for each group (highest 5% of Z scores for mean components for the group) that were identified as RSNs (vs. artifact; see Appendix A). RSNs are displayed on the group template (participants with aphasia; see Section 3.5.2) and at the coordinates (upper right corner) of the peak voxel for the healthy controls group (highest Z score). RSNs for healthy control group are shown in blue. RSNs for participants with aphasia are shown in red. Overlap for both groups is shown in purple.



Figure B.1 Overlap for dorsal attention network (DAN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.2 Overlap for default mode network (DMN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.3 Overlap for frontoparietal control network (FPC). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.4 Overlap for language network (LAN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (left hemisphere); right= axial.



Figure B.5 Overlap for left ventral attention network (LVAN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.6 Overlap for right ventral attention network (RVAN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.7 Overlap for sensorimotor network (SMN). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Figure B.8 Overlap for visual network (VIS). Healthy controls (blue) and participants with aphasia (red); overlap shown in purple. Left= coronal; middle= sagittal (right hemisphere); right= axial.



Appendix C: Non-Significant Dynamic Functional Connectivity States

Figure C.1 Correlation matrices for the nine dynamic states not significantly correlated with changes in behavior on the narrative production task, as reported in Chapter 6. From left to right on the x axis and top to bottom on the y axis, resting state networks (RSNs) are: dorsal attention (DAN), default mode (DMN), frontoparietal control (FPC), language (LAN), left ventral attention (LVAN), right ventral attention (RVAN), sensorimotor (SMN), and visual (VIS). Color bars show binned correlation values.